THE MAKING OF A MENTAL PATIENT

THE MAKING OF A MENTAL PATIENT:

AN ETHNOGRAPHIC STUDY OF THE PROCESSES AND CONSEQUENCES
OF INSTITUTIONALIZATION ON PATIENT SELF-IMAGES AND IDENTITIES

BY

NANCY J. HERMAN, B.A.

A Thesis

Submitted to the School of Graduate Studies
in Partial Fulfilment of the Requirements
for the Degree
Master of Arts

McMaster University
September 1981

MASTER OF ARTS (1981) (Sociology)

McMASTER UNIVERSITY Hamilton, Ontario

TITLE:

The Making of a Mental Patient: An Ethnographic Study of the Processes and Consequences of Institutionalization on Patient Self-Images and Identities

AUTHOR:

Nancy J. Herman, B.A.

(McMaster University)

SUPERVISOR: Professor J. Haas

NUMBER OF PAGES: vidi, 207

ABSTRACT

This thesis presents an updated study of the social world of the mental patient as he/she subjectively experiences it within a Canadian psychiatric setting. Adopting a symbolic interactionist perspective, specifically a labelling approach to the study of mental illness, this thesis examines the 'moral career' of the mental patient. Objectively, this study focuses on some of the processes by which an individual is segregated, labelled, and subsequently treated as mentally ill. Subjectively, this research examines the manner in which an individual's self-identity is transformed upon hospitalization through various stripping procedures, and how he/she, through institutional processing, is forced to adopt a new definition of self as 'mentally ill'--a social identity that may subsequently lead to the person's stigmatization by society. The findings, in support of previous research, indicated that during the pre-patient phase of their moral careers individuals often undergo experiences which they conceive as alienating and treacherous in mature. Moreover, this study also found that although in-patients are subjected to more 'humane' treatment than in the past, they are still subjected to various institutional procedures which serve to strip the person of his/her former identity and force: the person to adopt this new identity of mental patient. Although this study is restricted to the general and exploratory level, it provides a contribution to our understanding of the processes and consequences of institutionalization on the self-images and identities of mental patients and toward the formulation of a comprehensive sociological theory of mental illness in general.

ACKNOWLEDGEMENTS

Theses are not written without the help and support of others.

In my case, I wish to express my appreciation to the members of my committee: Drs. Jack Haas, William Shaffir, Barry Edginton and Richard Brymer. Each of these people provided constructive criticism and invaluable guidance throughout this research.

I wish to thank the Social Sciences and Humanities Research Council for the financial support of this project.

I also wish to thank my family for their continual emotional support and for putting up with my 'craziness' during the writing of this thesis.

Finally, I wish to thank all the patients and ex-patients I interviewed for allowing me to enter and share in their world—an experience which proved highly worthwhile.

TABLE OF CONTENTS

CHAPTER	ONE	INTRODUCTION	1
CHAPTER	TWO	THEORETICAL ORIENTATION	7
		The Underlying Assumptions of Symbolic Interactionism	7
		The Main Tenets of the Labelling Theory of Deviant Behaviour	18
		Career, Status, and Identity Trans- formation	23
•			
CHAPTER	THREE	LITERATURE REVIEW	29
		The Medical Model	30
		The Structural-Functionist Model	38
		The Existential-Phenomenological Model	44
		The Social-Role Model	L.S

CHAPTER	FOUR	THE METHODOLOGICAL ORIENTATION	59
		Participant Observation	60
		The Research Settings and Sample	62
		Gaining Access into the Research Settings	64
	c	Learning the RopesResearcher Data Collection Experiences with the Sub- jects	88
		Explanations of My Study and Role as a Researcher to the Subjects	91
CHAPTER	FIVE	THE PRE-PATIENT PHASE IN THE CARRER	
		OF THE MENTAL PATIENT	99
		Pathways Leading to the Hospital	100
		The Types of Rule Violations to which Audiences React	101
		Agents and Agencies that Participate in the Definitional Process	104
CHAPTER	SIX	THE IN-PATIENT PHASE IN THE CAREER	
		OF THE MENTAL PATIENT.	123
		The Processes of Admission: The Context Within Which Self is Mortified	123

	The Privilege System: The Context Within Which Self is Reconstituted	134
CHAPTER SEVEN	THE POST-PATIENT PHASE IN THE CAREER OF THE MENTAL PATIENT	150
-	The Perception of Mental Illness as Stigma	152
	Strategies of Information Manage- ment Employed by Post-Patients	155
CHAPTER EIGHT	SUMMARY AND CONCLUSIONS	168

BIBLICGRAPHY

181

CHAPTER ONE

INTRODUCTION

Who can tell when health ends and disease begins? When disease is found to have shed its blighting influence over the system, is it possible, after establishing the fact, to decide what amount or kind is necessary to occasion aberration of mind, and when this amount or quality is developed? When developed, does it at once manifest its baleful influence upon the brain, by producing insanity; or does it not rather brood over the delicate organ of the mind, and gradually fulfill its dread commission? When again the mind begins to totter, and reason to sit insecurely upon her throne, do the friends and acquaintances of the unhappy sufferer recognize these first monitions? Or do they not rather beholdif indeed they observe any thing--a simple change of habit, slightly perverted moral feelings, or trifling eccentricities of character? (Esquirol, 1845, in Goshen, 1967:316).

The study of insanity has been approached from a variety of perspectives. Traditionally, psychiatrists and psychologists, advocating a medical model stance have conceptualized mental disorders as disease entities located within the individual. Adopting an objective theoretical and nosological classification scheme, such persons have focused on the examination, classification, etiology and treatment of observable pathological behaviour of the patient. In short, in this perspective mental disorders are comprehended from an objective point of view and treated in a manner similar to other illnesses. Man is conceived as a 'diseased' psychological or physic-

logical entity—no attempt is made to understand the world of the mental patient from his/her perspective (e.g. Siegler and Osmond, 1974; Lewis, 1967; and Whybrow, 1972).

In reaction against the premise that mental disorders are intrinsic defects located within the individual, sociologists and social psychologists developed an alternative conceptual framework that places emphasis on the social aspects of mental illness. Conceiving of mental illness as subjectively problematic, proponents of this perspective examine the social processes by which an individual comes to be labelled and treated as mentally ill. and how such a definition impinges upon the actor (e.g. Goffman, 1961; Kutner, 1962; Mechanic, 1962; Sampson, et al., 1955; and Scheff, 1964, 1966, 1967, 1975). The majority of these sociological studies on mental patients were conducted in the 1950's and 1960's. I would argue that institutional life and ideology are not static, but rather dynamic in nature; therefore, it is useful to re-examine the social world of the mental patient as he/she subjectively experiences it in the 1980's. Moreover, these previous studies were conducted primarily in American psychiatric facilities. Up to this point, little research has been conducted on mental patients in Canadian psychiatric institutions.

The purpose of this thesis, then, is to present an updated

The theoretical focus of this research is threefold. First, it examines the pre-patient phase of the mental patient's career—the social conditions upon which an individual is adjudged to be 'mentally ill' and is segregated from the community, and the corresponding reaction of the person so adjudged.

For the purposes of this study, the term 'mental patient' is defined strictly in the sociological sense. Specifically, this term refers to individuals who possess any of the conditions described in the Canadian Psychiatric Association's Diagnostic and Statistical Manual. In this perspective, the psychiatric conception of mental illness is important only to the extent that this view transforms a person's social fate. Thus, for the purposes of this thesis, anyone who has been admitted into a mental hospital shall be included in the category, 'mental patient.' While one cannot discount that mental patients differ in terms of the type and severity of illness, during the processes of admission, institutionalization, and discharge these individuals undergo similar social experiences. It is these social experiences that are the central focus of the study.

The second theoretical concern of this study is to focus on the in-patient phase in the career of the mental patient. Specifically, this research explores the various admission procedures to which the patient is subjected upon institutionalization, and how such processes serve to strip the individual of his/her present identity. Moreover, this study focuses on the institution's 'privilege system'—the chief context within which the patient is forced to adopt a new definition of self as 'mentally ill.'

Thirdly, this study focuses on the ex-patient phase in the mental patient's career. Specifically, this research examines the problems ex-patients face in attempting to rebuild a more positive identity due to the widespread negative societal attitudes toward individuals of their kind, and the various mechanisms such post-patients develop in order to mitigate the stigma potential of mental illness on their daily round.

As with most research, the present study does suffer from certain limitations. In order to gather data on patient social life, strict statistical measurements or controls were not employed. Rather, descriptive data were gathered on the nature of patient social life through the employment of participant observation and informal interviewing techniques.

Although this thesis is restricted to the general and exploratory level, it provides a contribution to our knowledge concerning the processes and consequences of institutionalization on the self-images and identities of mental patients.

Working from the fundamental premise that self is constituted in the context of social interaction, Chapter Two of this thesis outlines

Se section of

the theoretical framework underlying the present research. The basic tenets of the symbolic interactionist perspective and the labelling theory of deviant behaviour are presented as a basis for interpreting the data. Chapter Three presents a review of the existing literature on mental patients, mental institutions and mental illness in general, with specific focus on the four major models of madness and the model adopted for the present research study. Chapter Four discusses the methodological orientation of the study, the research settings and sample employed, followed by a description of the researcher's experiences in gaining accessibility into the settings, and the subsequent data collection experiences with the patients. Chapters Five, Six and Seven present a discussion of the empirical data concerning the three stages in the career of the mental patient. Chapter Eight provides a summary of the findings and discusses the overall implications of the thesis and suggestions for future research.

FOOTNOTES FOR CHAPTER ONE

- 1 Hathaway's (1980) research is one recent notable exception.
- Following Goffman (1961:128), the term 'moral career' shall refer to: "the regular sequence of changes that career entails in the person's self and in his framework of imagery for judging himself and others."

CHAPTER TWO

THEORETICAL ORIENTATION

This chapter serves to introduce the reader to the theoretical framework underlying the present investigation. Specifically, this chapter discusses the following:

- (1) the underlying assumptions of symbolic interactionism;
- (2) the basic concepts of this theory such as, meaning, symbol, mind, role-taking, self, generalized other and society—concepts which are utilized in the present study;
- (3) the main tenets of labelling theory—a theory of deviant behaviour that is rooted in the Meadian interactionist tradition and has subsequent importance for this research.

The theoretical perspective employed in this research stems primarily from the works of George Herbert Mead. Examination of Meadian theory reveals that it is based on six basic fundamental assumptions:

1. Man lives in a symbolic world of learned meanings. The objective world with its independent laws has no meaning for man; rather, subjective reality has prime importance for him. For Mead, reality is mediated through symbols.

2. Meanings are derived from symbols which arise in the social process and are universal.

THE STREET

1 |

- 3. Symbols have motivational significance. The definition of the situation aids in determining how an individual will behave. In this sense then, symbols are themselves motives.
- 4. Mind is a functional, volitional, purposive sensory process which sustains life by serving the organism in response to the environment. Mind emerges out of the social process and is wholly social in character.
- 5. If the world of objects is mediated through symbols, then other objects can only be experienced in this same way. Self is constructed from meanings that arise in symbolic interaction. That is, self is a linguistic construct that is derived from the social process.
- 6. Society is a linguistic or symbolic construct that arises out of the social process (Rose, 1962:3-20).

A fundamental assumption of symbolic interactionism is that humans live in a world of meanings. They respond to objects and events on the basis of the meanings they have attributed to them. For Mead (1934:85), meanings of events are not static or inflexible in nature for if this were the case, social change would be impossible. Neither are the meanings of objects and events merely bestowed upon the individual and learned by habituation. Rather, meanings of events and objects can be changed through the creative acts of individuals and the individual may influence the numerous meanings that form his culture as well as being influenced by these meanings himself. Meanings then, are social products that are formed through the defining activities of people as they interact with one another:

The universe of discourse is constituted by a group of individuals carrying on and participating in a common social process of experience and behavior, within which these gestures or symbols have the same or common meanings for all members of that group, whether they make them or address them to other individuals, or whether they overtly respond to them as made or addressed to them by other individuals. A universe of discourse is simply a system of common or social meanings (Mead, 1934:89-90).

These culturally-shared meanings which arise from an interpretive process sheds light on the collective character of the process. Although one can advance the argument that an individual has the ability to engage in this interpretive process alone, through thinking, however, individual thought is <u>not</u> the origin of socially-shared definitions. Mead contends that thinking is made possible through interaction with others. In this sense, thinking is really a <u>social</u> activity. Therefore, meanings of objects and events are shared in a double sense: not only do people hold common meanings but they also participate collectively in creating them.

Further, these shared meanings which have arisen through social interaction aid in determining the behaviour of individuals. In this sense, symbols may be said to have motivational significance. For example, if we define an object as a chair, by virtue of this definition determines how we will act toward it. That is, we know that a chair is used to sit upon and not to place one's dishes upon. For Mead then, symbols are motives.³

Explicit in Mead's theory is the notion that individuals are selfconscious beings; they possess selves and minds which although absent at birth, emerge as a result of the individual's participation in the social world.

In terms of the nature of mind, Mead states that the mind is a unique human characteristic which arises in the evolutionary process when the impulsive behaviour of the individual is hindered as he attempts to adapt to his environment. Mind, for Mead develops out of the non-mental behaviour of pre-human biological forms.

Mead contends that the lower animals respond to the environment but are neither able to understand factors that affect their behaviour, nor are they able to conceive of alternative behaviours. In contrast however, humans are able to select out and indicate to themselves and others the meanings of certain environmental features to which they are responding. In this sense, individuals are able to achieve control over the stimuli to which they wish to respond:

Mentality simply comes in when the organism is able to point out meanings to others and itself...Mentality resides in the ability of the organism to indicate that in the environment which answers to his responses, so that he can control those responses in various ways... Mentality in our approach simply comes in when the organism is able to point out meanings to others and to himself (Mead. 1934:132).

For Mead, mind is the mechanism of control over meaning. It is a process that functions to serve the needs of individuals in their environmental adjustments.

Mead conceives of mind not as a substance but as a social product that emerges out of the social process and is made possible by meanings and symbols which are social in character. This is rendered possible through language. Mead asserts that only when an individual has acquired a self and is able to use significant symbols is he able to exercise mind in relation to certain objects. Thus, in this sense social consciousness is historically prior to phsycial consciousness. Mind and social consciousness emerge when gestures stimulate implicitly in the actor the same response as in others explicitly.

Just as mind is a social product that arises through interaction with others, Mead conceives of self in a similar manner. Specifically, he asserts that self is not initially present at birth; it develops in the social process and undergoes continual development through the life of the individual. Mead contends that self is not wholly impregnated in society for once it has arisen, it is able to provide social experience for itself. Thus, while the self can be regarded as a social structure, it is also a process within the larger, ongoing social structure:

The self has a character which is different from that of the physiological organism proper. The self is something which has a development; it is not initially there at birth but arises in the process of social experience and activity, that is, develops in the given individual as a result of his relations to that process as a whole and to other individuals within that process...The self...is essentially a social structure, and it arises in social experience. After a self has arisen, it in a certain sense provides for itself its social experiences, and so we can conceive of an absolutely solitary self (Mead, 1964:199; 204).

For Mead, the distinctive nature of the self is found in the ability of the individual to be an object unto itself. Mead (1934:150)

contends that an individual has the ability to be both subject and object by 'taking the role of the other':

The individual experiences himself as such, not directly, but only indirectly from the particular standpoints of other individual members of the same social group, or from the generalized standpoint of the social group as a whole to which he belongs. For he enters his own experience as a self or individual, not by becoming a subject to himself but only in so far as he first becomes an object to himself just as other individuals are objects to him or his experience; and he becomes an object toward himself only by taking the attitude of other individuals toward himself within a social environment or context of experience and behavior in which both he and they are involved (Mead, 1934:138).

For Mead then, consciousness of self occurs when the individual is able to take the organized attitudes of others toward himself. Specifically, consciousness of self occurs through a type of communication utilized by what Mead (1934:67) refers to as 'significant symbols'—gestures that stimulate implicitly in oneself the same response that they stimulate explicitly in others. In short, self is conceived of as a social product that is made possible by language. One develops a self and becomes conscious of his self by applying words and meanings to his being that he obtains from society. Thus, self is conceived as a symbolic or linguistic construct that arises through social interaction:

It is in the social process itself that is responsible for the appearance of the self; it is not there as a self apart from this type of experience (Mead, 1934:40).

Having outlined the general nature of the social self, this

discussion will now briefly examine the conditions within which self originates and develops. As was previously stated, individuals are born into the world without a self. Self is a product of social interaction, which during the early stages of one's life is unconscious. In terms of a newborn baby, his needs such as food and shelter are met within a social context. Mother and baby interact with one another but only at a biosocial level. In this sense, the child is merely engaging in unconscious social behaviour or what Mead (1934:167) refers to as a *conversation of gestures. * Self evolves as a result of this unconscious behaviour. As a child gets older, he is able to take the attitudes of others toward himslef. Significant symbols and language emerge when the child is able to complete an inhibited action on the level of his imagination. Whereas consciousness and meaning arise out of unconscious communication within interaction, it is in this same process that self arises. Through this process, the individual is able to take the role of the other. In short then, explicit in Meadian theory is the idea that consciousness, meaning, self and facility with significant symbols are merely different phases of the same process.

In his discussion of the genesis of the self, Mead distinguishes between two stages: the play stage and the game stage. Genesis of the self involves the individual gaining the capacity to order experience in terms of social roles. That is, in the early stage of development (the play stage), the child 'plays at' being various people. For example, he plays at being father, policeman, teacher, etc. At this point, the child takes the role of particular others. It is through this play stage that

A

THERMAN

the child gains facility in organizing the attitudes of others in terms of social roles.

THE STATE OF THE

Not only must the individual learn to take the particular attitudes of individuals separately, but he must also learn to take the attitudes of the organized society collectively, or what Mead (1964:219) refers to as the 'generalized other.' This phase, which is termed by Mead as the 'game stage' may be likened to the game of baseball:

The fundamental difference between the game and play is that in the latter the child must have the attitude of the other players in that game. The attitudes of the other players which the participant assumes organize into a sort of unit, and it is that organization which controls the response of the individual (Mead, 1964:218).

In the game stage, the individual is placed in situations where he is forced to take on a number of social roles at the same time—he is forced to respond to the expectations of many individuals at once. In the game of baseball for example, the person has to visualize the attitudes and expectations of the entire team as over against the particular social roles. In this context, he must learn to abstract a general role out of particular roles—a generalized other that represents the organization of the entire social group. Only when the individual is able to take the attitudes of the generalized other toward himself is he able to possess a fully—developed self:

In the play stage, the individual's self is constituted simply by an organization of the particular

attitudes of other individuals toward himself and toward one another in the specific social acts in which he participates with them. But at the second stage in the full development of the individual's self, that self is constituted not only by an organization of these particular individual attitudes, but also by an organization of the social attitudes of the generalized other or social group as a whole to which he belongs. These social or group attitudes are brought within the individual's field of direct experience and are included as elements in the structure or constitution of his self. in the same way that the attitudes of particular other individuals are... So the self reaches its fullest development by organizing these individual attitudes of others into the organized social or group attitudes, and by thus becoming an individual reflection of the general systematic pattern of social or group behavior in which it and its others all are involved....(Mead, 1964:222).

At this point, it is worthy to make brief mention of the phases of self as distinguished by Mead. Explicit in Mead's theory is the notion that self is not merely an importation of the organized set of attitudes of others. In the self, there is involved a response of the person whose self it is. Mead distinguished between the 'I' and the 'me', the former representing the response of the individual to the organized attitudes of others, and the latter representing the individual's importation of the generalized attitudes of the community. These two phases are parts of the self that evolve in symbolic communication:

The 'I' is the response of the organism to the attitudes of the others; the 'me' is the organized set of
attitudes of others which one himself assumes. The
attitudes of the others constitute the organized 'me,'
and then one reacts toward that as an 'I' (Mead, 1964:230).

For Mead then, self is conceived of as a process that moves in and out of these two phases. The 'me' represents a reflection of the existing social process; it allows the individual to orient his behaviour toward the organized attitudes of the community. The 'I' represents the subjective aspect of social life; it symbolizes the unpredictable, illusive aspects of the individual, what Mead (1934:117) terms as the 'fictitious I' and provides the basis for novelty.

Although the 'I' and the 'me' appear in experience as distinct and separate characteristics of the individual, it is apparent that they are really parts of a single entity. With reference to these phases, Mead points out that self is incapable of appearing in consciousness as an 'I' but can only do so as a 'me.' To quote Mead:

Such an 'I' is a presupposition, but never a presentation of conscious experience, for the moment it is presented it has passed into the objective case, presuming, if you like, an 'I' that observes—but an 'I' that can disclose himself only by cassing to be the subject for whom the object 'me' exists (1913:374).

Just as mind and self were conceived by Mead as being symbolic or linguistic constructs that arise in social interaction, it logically follows that he viewed society in the same manner.

For Mead, society is a process in which mind and self emerge and change. Humans participate in two qualitatively different social processes: the biosocial and the symbolic, the symbolic emerging out of the biosocial:

The behavior of all living organisms has a basically social aspect: the fundamental biological or physiological impulses

and needs which lie at the basis of all such behavior—especially those of hunger and sex, those connected with nutrition and reproduction—are impulses and needs which, in the broadest sense, are social in character or have social implications since they involve or require social situations and relations for their satisfaction by any given individual organism and they thus constitute the foundation of all types or forms of social behavior, however simple or complex, crude or highly organized, rudimentary or well developed (1934:227-228).

THE STREET

Mead contends that human symbolic interaction arises out of this simple social interaction involving organisms. It is out of this symbolic interaction that mind, self and society evolve.

In each social process, be it either biosocial or symbolic, the fundamental unit is the act, and society may be conceived of as a coordinated series of actions. For Mead, human societies are structured with regard to the generalized attitudes of others—attitudes that dictate similar responses. An institution, in Mead's view, is comprised of a grouping of such common responses. It follows then that society is comprised of these institutions. In sum, society is viewed as an arrangement of the generalized attitudes of others.

While Mead conceives of society as a process that evolves through time, he also possesses a similar conception of social structure (Hormosty, 1980:see 3.4). In contrast to Durkheim who argues that society is sui generis in nature, Mead in his theory attempts to explain individual actions in their relation to society. Mead (1934:7) asserts:

For social psychology, the whole (society) is prior to the part (the individual), not the part to the whole; and the part is explained in terms of the whole, not the whole of the part or parts.

For Mead then, society is viewed as an 'evolving, functionally differentiated constellation of social roles' (1934:7).

Having examined the fundamental assumptions of Meadian symbolic interactionism and its basic concepts, this discussion will now turn to an outline of the labelling theory of deviant behaviour—a theory clearly rooted in the interactionist tradition.

The Labelling Theory of Deviant Behaviour

The emergence of labelling theory in 1960 developed out of ideas first expounded by Tannenbaum in 1938 and later more systematically developed by Lemert in 1951. In reaction against the traditional theories of deviant behaviour which centred on an examination of the causes and consequences of various forms of deviance, Tannenbaum and Lemert placed primary emphasis on the societal reaction to deviant behaviour.

Tannenbaum, in his discussion of the role that societal stigmatization plays in creating a criminal career states:

The process of making the criminal, therefore, is a process of tagging, defining, identifying, segregating, describing, emphasizing, making conscious and self-conscious; it becomes a way of stimulating, suggesting, emphasizing, and evoking the very traits that are complained of... The person becomes the thing he is described as being. Nor does it seem to matter whether the valuation is made by those who would punish or by those who would reform... The harder they work to reform the evil, the greater the evil grows under their hands. The persistent suggestion, with whatever good intentions, works mischief, because it leads to bringing out the bad behaviour that it would suppress. The way out is then a refusal to dramatize the evil (1938:19-20).

In a more systematic attempt at theoretical explication, Lemert provides much of the foundation upon which current labelling theory is built:

...we start with the idea that persons and groups are differentiated in various ways, some of which result in social penalties, reaction and segregation. These penalties and segregative reactions of society or the community are dynamic factors which increase, decrease and condition the form which the initial differentiation or deviation takes...The deviant person is one whose role, status, function and self-definition are importantly shaped by how much deviation he engages in, by the degree of its social visibility, by the particular exposure he has to the societal reaction, and by the nature and strength of the societal reaction (1951:22, 23).

Examination of the major tenets of labelling theory reveals that 7 it is clearly oriented within an interactionist framework. A first, fundamental assumption of labelling theory, corresponding with the Chicago school, posits that one is not able to fully comprehend deviant behaviour in terms of the actions themselves but only if it is realized that deviant behaviour, like 'normal' behaviour involves social interaction with others. For labelling theorists, deviance is conceived of as a product of social interaction between the actor committing a deviant act and the audience who responds to his behaviour. In short, a central tenet of labelling theory is not concerned with action per se but with societal reaction. As Erikson aptly expresses it, labelling deviance is contingent upon the reaction of an audience:

From the sociological standpoint, deviance can be defined as a conduct which is generally thought to require the attention of social control agencies—that is, conduct about which 'something should be done.' Deviance is not a property inherent in certain forms of behavior; it is a property conferred upon these forms by the audiences which directly or indirectly witness them. Sociologically then, the critical variable in the study of deviance is the social audience rather than the individual person, since it is the audience which eventually decides whether or not any given action or actions will become a visible case of deviation (1962:308).

Similarly, Becker states:

From this point of view, deviance is not a quality of an act a person commits but rather a consequence of the application by others of rules and sanctions to an 'offender.' The deviant is one to whom the label has been successfully applied (1963:9).

Kitsuse supports the emphasis on societal reaction when he says:

Forms of behavior per se do not differentiate deviants from non-deviants; it is the response of conventional and conforming members of society who identify and interpret behavior as deviant which sociologically transforms people into deviants (1962:253).

Conceptualization of deviant behaviour as a 'reaction process' of society leads to a second related assumption of labelling theory which asserts that the demarcation between conventional and unconventional behaviour is disputable and ambiguous. According to the labelling perspective, what are considered acceptable and unacceptable behaviours is entirely relative. As Scheff (1975:10) argues, labelling is not an auto-

matic process: whether one is labelled a 'deviant' is contingent upon several factors such as:

- (1) the visibility of the rule-violation;
- (2) the power of the rule-violator in relation to the individuals responding to this violation;
- (3) the severity of the violation;
- (4) the tolerance level of the community;
- (5) the availability of the society to channel the reaction to something other than labelling.

In short, proponents of labelling theory posit a reciprocal process operating between the actor and his audience. Rather than conceiving of deviance as 'objectively given' and attempting to gather descriptive data on rule-violators from the official records, labelling theorists, in contrast, conceive of deviance as subjectively problematic. For advocates of the labelling perspective: "processes of social interaction must be inspected to ascertain the conditions under which deviance is defined and what consequences flow from that definition" (Rubington and Weinberg, 1973:1-2).

Iabelling theorists, in their conceptualization of deviance as subjectively problematic, focus their analyses upon issues generated in the interactions between the actor and others in society. That is, this perspective focuses on the following: (1) those who define an individual as a deviant; and (2) the individual who has been negatively labelled and stigmatized by others as being 'deviant.' In terms of the former, pro-

ponents of this perspective focus upon: (a) the conditions under which an individual is segregated and labelled a deviant; (b) how the individual is cast into the deviant role; (c) the behaviour of others toward this redefined person; and (d) the positive or negative value that others place upon the facts of deviance (Rubington and Weinberg, 1973:2).

In terms of the latter, labelling theorists focus upon the actor himself who has been negatively labelled and stigmatized by others as 'deviant.' Specifically, this perspective centres upon: (a) the reaction of the actor to the label bestowed upon him; (b) the manner in which he/she adopts the deviant role; and (c) the extent to which the actor adopts this new conception of self (Rubington and Weinberg, 1973:3).

Career, Status and Identity Transformation

Proponents of the labelling perspective often employ the concept of 'career' in their analyses of deviant socialization processes.

In traditional form, the concept of 'career' was employed in relation to a profession or occupation where it referred to the course of an individual's expected or actual occupational activities from one stage to another. That is, this concept specifically referred to the advancements or promotions within a certain occupation or profession (Goffman, 1961:127).

More broadly however, sociologists have expanded the meaning of this concept to refer to the temporal sequencing of actions in <u>any</u> sphere of life and not solely in reference to occupations.

According to Everett C. Hughes (1958:67), the concept of career

may be viewed as:

...the moving perspective in which people orient themselves with reference to the social order...Institutions are but the forms in which the collective behavior and collective action of people go on. In the course of a career, the person finds his place within these forms, carries on his active life with reference to other people, and interprets the meaning of the one life he has to live.

And Tamotsu Shibutani (1962:137) writes:

Career lines are organized, and there is usually an orderly sequence of steps through which one moves from apprenticeship to mastery. It is in terms of such ladders that aspirations are shaped, and a person who is on his way measures his progress by comparing himself with his predecessors, not with outsiders...Common memories are built up and reinforced within the limited communication network.

Further, Erving Goffman (1961:168) states:

The moral career of a person of a given social category involves a standard sequences of changes in his way of conceiving of selves, including, importantly, his own. These half-buried lines of development can be followed by studying his moral experiences—that is, happenings which mark a turning point in the way in which the person views the world--although the particularities of this view may be difficult to establish... By taking note of the moral experiences and overt personal stands, one can obtain a relatively objective tracing of relatively subjective matters. Each moral career, and behind this, each self, occurs within the confines of an institutional system... The self, then, can be seen as something that resides in the arrangements prevailing in a social system for its members...the self dwells in the pattern of social control that is exerted in connection with the person by himself and those around him. This special kind of institutional arrangement does not so much support the self as constitute it.

In short, the concept of career may be viewed as a movement of individuals through a structure of society. The different stages in a career symbolize the transformation of the person's status and identity, and allows him to view himself as moving along a continuum. Hughes (1958:72) contends that the concept of career is two-sided in nature. That is, objectively this concept can be conceived of as a passage through various roles and statuses. Subjectively, the concept of career may be conceived of as being comprised of individuals' self-images and identities as they move through different institutions and organizations.

For labelling theorists then, careers occur within institutional frameworks where neophytes are processed and socialized into various roles and statuses, which in turn, affect their conceptions of self.

In short, many deviant identities and careers are molded, developed and sustained in what have been termed as 'people-processing institutions: 'll

This term...refers to a type of social institution in which human beings constitute both the raw materials and the products of organizational work. Although all social institutions are involved in some degree in people-processing activities, this term is properly restricted to those whose primary goal is the shaping, reshaping, removing, overhauling, retooling, reassembling, and recording the physical, psychological, social, legal or moral aspects of human objects (Kitsuse, 1970:163).

As a consequence of this institutional processing, the individual is confronted with a new social identity—an identity which is incompatible with his prior conception of self. The individual may initially
reject this redefinition of self as 'deviant' but through reinforcement

from institutional staff (and sometimes even from his peers), the individual is forced to accept this new definition of self—a social definition that may subsequently lead to his stigmatization by society. In essence, the institution's perception of the individual produces a self-fulfilling prophecy whereby the individual comes to be molded in the image that the institutional staff have of him.

In conclusion, this chapter has attempted to introduce the reader to the theoretical stance adopted in this thesis. Specifically, the fundamental assumptions and concepts of symbolic interactionism were described. Secondly, a brief sketch of the labelling theory of deviant behaviour and its related concepts was presented, a perspective that will be employed in the subsequent analysis.

The next section will present a review of the existing literature on mental patients, psychiatric institutions, and the nature of mental illness in general.

FOOTNOTES FOR CHAPTER TWO

- 1 See, George Herbert Mead (1932), (1934), (1936), and (1938).
- For a detailed discussion of the tenets of symbolic interactionism, see the following: Blumer (1966), (1969); Rose (1962); Manis and Meltzer (eds.) (1967). In recent years, various criticisms have been levelled against certain tenets of symbolic interactionist theory by both in-house critics (interactionists) and outsiders (non-interactionists). See for example: Kolb (1944:291-296); Meltzer (1959), (1972:4-22); Huber (1973:278-284); Lichtman, (1970:75-94); Gouldner (1970): Shaskolsky (1970:7-30); Ropers (1973:15-28); Kanter (1972:77-92); Smith (1973:62-75); Denzin (1979:922-934); Kuhn (1964: 61-84); Manis, Petras and Reynolds (1975:83-115).
- Not only do social groups provide the individual with meanings and definitions about appropriate forms of action, but they also function to provide rationales for behaviour. As C.W. Mills (1940:907) notes, "motives are accepted justifications for present, future, or past progress of acts." In this sense then, motives are conceptualized as socially learned expressions that serve to provide the individual and others with rationalizations for behaviours. See also Schutz (1967) for a discussion of motives as being both rationalizations for behaviour as well as plans of action.
- This idea is fully expounded in Mead (1910:178).
- For evidence in support of the labelling perspective, see: Scott (1970); Jewell (1952:32-36); Mercer (1966:21-34); Freidson (1966a:82-93); Lemert (1962:2-25).
- A detailed discussion of the traditional approaches to the study of social problems may be found in the following: Bernard (1957); Davis (1975); Herman (1980); Matza (1969); and Rubington and Weinberg (eds.) (1977).

- For a discussion of the various criticisms of labelling theory see: Fine (1977:166-190); Gove (1970:873-884); Davis (1972:447-474), (1975:197-234); Gibbs (1966:9-14). It is beyond the scope of this discussion to deal with the various criticisms levelled against labelling theory. This chapter will deal solely with the underlying assumptions of this perspective in relation to their subsequent importance for this research.
- Rubington and Weinberg (1973:1-4) provide an elaboration of the two approaches to the study of deviant behaviour.
- 9 See: Sykes (1958); Bryan (1964); Weinberg (1966); Crespo(1973: 129-145); and Hall (1948:327-336).
- for a discussion of a sociological theory of identity transformation, careers, and changes in status, see Glaser and Strauss (1971). See Roth (1963) and Davis (1963) for an examination of the careers of the tuberculosis patient and the polio patient respectively. Anthropologists have also been interested in examining transformations in identity and alterations in status. See for example: Van Gennep (1960); Turner (1969) and Gluckman (1962).
- This term, originating with Goffman (1961) has been employed by Freidson (1966b); Cicourel and Kitsuse (1963); and Bittner (1967).

CHAPTER THREE

LITERATURE REVIEW

The history of man's attempt to understand his world in order to explain, control and predict events inevitably involves the utilization of conceptual models or classification schemes. It is through these various theories and conceptual frameworks that man selects, orders and explains his observations—through these perspectives, courses of action are determined.

One such problem man has sought to understand concerns the nature of 'insanity'--a concern which has resulted in the construction of numerous 'models of madness.'

This chapter serves to present to the reader a review of the existing literature on mental patients, psychiatric institutions and the nature of mental illness in general. Specifically, this chapter focuses upon four major models of insanity:

- (1) the medical model
- (2) the sociological models: (a) the structural-functionalist model
 - (b) the existential-phenomenological model
 - (c) the social-role model

This discussion centres upon the fundamental tenets of each conceptual model, the primary figures who have adopted these models in their research endeavours concerning insanity, followed by an examination of the strengths and weaknesses of each model. The chapter concludes with a discussion of the model adopted for the present research study.

The Medical Model of Mental Illness

Examination of the medical model of mental illness reveals that it is based upon two fundamental tenets:

- (1) mental disorders are conceptualized as 'diseases;'
- (2) the notion of clinical universalism.

In terms of the former, advocates of the medical model contend that mental disorders are diseases located within the individual and should be treated as such—they have biological or genetic causes and thus, should be treated in a manner similar to diseases of the body. Diseases, according to this model are conceived of as disturbing conditions that can be cured or alleviated through the proper care of trained physicians. Within this model, the physician is placed in an authoritarian role for he is granted the right to explore both the physical and psychological dimensions of the individual—a right that entails an obligation to help the patient. The patient is conceptualized as a 'sick person' in need of psychiatric care—a role characterized by

dependency and lack of responsibility for self.

Aubrey Lewis (1967:179-194), in his discussion on the nature of health and illness asserts that the medical model of mental illness contains three types of data: (1) the objective—a functional or behavioural disorder that is visible to the physician; (2) the subjective—the complaints of the patient; and (3) classification—the placement of signs and symptoms within a clinical typological classificatory system. Adopting an objective theoretical stance and scientific classification scheme, proponents of this model seek to examine, define and treat the observable behaviour of the patient. Specifically, advocates of the medical model examine the symptomatology of the patient upon which a diagnosis is based, and subsequently prescribe a form of treatment. Siegler and Osmond (1974: 24), advocating a medical model stance aptly sum up this position:

First, a patient voluntarily comes to a building called a 'hospital' where he agrees to be called a 'patient' and where he is referred to as a 'case.' If he is unable to come on his own two feet, he may be brought by others who stand for him and for whom he would do the same. He then agrees to be handled with extraordinary physical intimacy by strangers called 'doctors' in a way allowed to no one else, not even a sexual partner. The function of this examination is to find out what the patient 'has'--no what he 'is' or 'does.' This means: into what category of previously described illnesses does the patient best fit?...On the basis of diagnosis, the doctor may decide upon a treatment, which is what the doctor wants to do for the patient... This brings us to prognosis, another essential medical function. The doctor must maintain the patient's hope and his will to live, while giving him a realistic idea of the likely course and outline of the illness.

In short then, in this model mental illness is comprehended from an objective point of view and treated in a manner similar to other illnesses.

Man is conceptualized as a 'diseased' psychological or physiological entity—no attempt is made to understand mental illness from the point of view of the patient himself.

Further, explicit in this model is the notion that social criteria are irrelevant in the definition of mental illness. In Lewis' (1967:194) view, mental illness cannot be conceived in terms of negative societal reaction because negative reaction is a function of the norms and values of a society and the individuals who make those judgments. Moreover, according to Lewis non-conformity cannot be criterion of mental illness because it is usually expressed in terms of social role and is influenced by such factors as conflict, culture change etc. Therefore, Lewis dismisses social criteria for defining mental illness because they reflect value judgments and are culturally-relative in nature. As Lewis (1967:194) states:

...the criteria of health are not primarily social: it is misconceived to equate ill-health with social deviation or maladjustment.

For Lewis then, (as with other advocates of the medical model) mental illness is conceptualized as an abnormal state of the organism.

Whybrow (1972), in his discussion of the medical model, conceives of mental health and illness as relative conditions. Specifically, Whybrow (1972:334) asserts that:

... (mental health is the) ability in the human animal to adapt, organize, and respond to a constantly changing social, psychological and biological environment... Disease, therefore, on a continuum with health, may be seen to correspond with a compromising or a failure of these functions.

In this perspective, mental health is equated with successful biological and social adaptation while mental illness, conceived of as a disease, results from failure at such adaptations. Moreover, the medical model emphasizes the processural nature of mental health and illness—few, if any, individuals are totally ill. Conversely, few, if any, socialled 'normal' individuals are completely free from emotional problems (Schwab and Schwab, 1978:122).

TO STREET

A second fundamental tenet of the medical model approach to the study of mental disorders involves the notion of clinical universalism.

Essentially, proponents of this model contend that except for cursory variation in content, mental illnesses are virtually the same throughout the world. E.B. Forster (1962:35) aptly expresses this view:

Psychiatric syndromes or reactions, by and large, are similar in all races throughout the world. The mental reactions seen in our African patients can be diagnosed according to the Western textbook standards. The basic illness and reaction types are the same. Environmental, constitutional, and tribal background merely modify the symptom constellation. Basically, the disorders of thinking, feeling, willing and knowing are the same.

E.L. Margetts (1965:24), a cross-cultural psychiatrist also advocates this universalist stance:

The more I listen to the discussion of transcultural psychiatry, the more I am coming to believe that perhaps there is no such thing...(Psychiatrists) have not learned a great deal about it since the time of Kraepelin. As far as I am concerned, psychiatry is the same all around the world: the signs and symptoms of mental diseases are the same, the diagnoses are the same, and there is probably just as much possession syndrome in England as there is in equatorial Africa.

The cross-cultural investigations suggest that relativism has been exaggerated by labeling theorists and that in widely different cultural and environmental situations sanity appears to be distinguishable from insanity by cues that are very similar to those in the Western world.

In summary, the medical model dismisses social/cultural factors in the definition of mental illness. Mental disorders are conceptualized as diseases located within the individual—they have biological or genetic causes and hence, should be treated similar to other illnesses. In this view, symptomatology of the disease determines the societal reaction rather than the converse.

Further, insofar as mental disorders are conceived as diseases, it logically follows for the clinical universalists that the symptoms should exhibit minimal cross-cultural variation. Clinicians assert that the symptomatology of mental illnesses are universal and societal reactions are similar cross-culturally. Murphy (1976:1025) aptly sums up this universalist stance:

If one defines intolerance of mental illness as the use of confinement, restraint or exlusion from the community (or allowing people to confine or exclude themselves), there does not appear to be a great deal of difference between Western and non-Western groups in intolerance of the mentally ill. Furthermore, there seems to be little that is distinctively cultural in the attitudes and actions directed toward the mentally ill, except in such matters as that an abandoned anthill could not be used as an asylum in the arctic or a barred igloo in the tropics. There is apparently a common range of possible responses to the mentally ill person, and the portion of the range brought to bear regarding a particular person is determined more by the nature of his behavior than the pre-existing cultural set to respond in a uniform way to whatever is labeled mental illness.

while many clinicians and social scientists advocate the medical approach to the study of mental disorders, others have found it unsatisfactory. One of the major criticisms of the medical model is its failure to include social processes in the examination of the nature of mental illness. That is, this conceptual framework is mainly concerned with the individual in contrast to the social system. Proponents of the medical model of mental illness seek genetic, biochemical and biological causal agents—dynamic systems that are located within the individual. In short, this model isolates the symptom from the social context within which it occurs. It merely postulates 'unconscious causes' of behaviour without examining the social context in which the behaviour occurs. In this sense, I would argue that the medical model is narrow in scope.

A second, related criticism centres on the model's contradictory position concerning the issue of cultural relativism. As Schwab and Schwab (1978:122) note, there exists a logical inconsistency in the

medical model of mental illness: on the one hand, the model presupposes societal norms and values—it posits a clinical universalist stance asserting that mental disorders exhibit similar symptomatology through—out the world and that different societies react to mental illness in a similar manner. On this basis, advocates of the medical model dismiss social/cultural factors in the definitional process of what constitutes mental illness. On the other hand however, the medical model implicitly states that within this perspective the patient supplies individual norms which are subsequently evaluated by psychiatrists in terms of the general norms of the psychiatric profession. Thus, I would conclude that the medical model is inconsistent in terms of its logic regarding the issue of cultural relativism.

A third limitation of the medical model centres on its derivation from the Darwinian notion of adaptation. Within this perspective, health is conceptualized as successful biological and social adaptation while illness (whether physical or mental) is conceived as a failure at such adaptations. As Schwab and Schwab (1978:122) note, this viewpoint is tautological in nature because it suggests that mental health and successful adaptation are synonymous. Aubrey Lewis (1967:185-186) aptly sums up this point:

...mental health cannot be equated with good social adaptation, as many have proposed, without risk of tautology: the valued and desired state which adaptation is to attain or maintain may itself turn out to be health.

Further, anthropologist Clyde Kluckholn (1962:260-264) contends

that it is narrow to think in terms of adaptation:

We require a way of thinking that takes account of the pull of expectancies as well as the push of tension, that recognizes that growth and creativity come as much or more from instability as from stability, and that emphasizes <u>culturally created values</u> as well as the immediately observable external environment. (emphasis mine)

THE STREET

A fourth criticism levelled against the medical model of mental illness centres on its lack of utility and applicability. Within this model, mental disorders are conceptualized as diseases that are caused by certain biological, bio-chemical or genetic factors. However, in strict scientific terms the etiology of the majority of the mental illnesses remains unknown. While many clinicians speculate about the etiological factors concerning the nature of mental disorders, the empirical evidence in inconclusive at this point. Therefore, on this basis the medical model has difficulty in categorizing many of the interpersonal difficulties, problems in living and conflicts faced by individuals—it merely has applicability for the diagnosis and classification of some of the more clearly-delineated types of disorders such as organic brain syndromes.

A final criticism levelled against the medical model is its failure to focus on the patient from a subjective point of view-failure to comprehend his experiences, his definition of reality. Adopting an objective theoretical and scientific classificatory scheme, proponents of this model merely examine the observable behaviour of the patient—they look at the patient's actions as 'signs' of a 'disease' upon which a diag-

nosis is conferred and treatment is prescribed. However, this model does not allow one to <u>understand</u> mental illness from the point of view of the patient himself. In this model, man is merely viewed as a 'diseased' psychological or physiological entity—no attempt is made to understand the overall situation of the individual in society. In summary, this narrow clinical focus with its objective classification scheme and technical vocabulary prevent the psychiatrist from attaining a holistic understanding of the world of the mental patient.

The Structural-Functionalist Model of Mental Illness

In contrast to the medical model which views mental disorders as diseases located within the individual, the structural-functionist model conceives of mental illness as an inherently social phenomenon. Talcott Parsons (1951b:453) in his discussion on the nature of mental illness asserts:

Seen in this perspective illness is to be treated as a special type of what sociologists 'deviant behavior.' By this is meant behavior which is defined in sociological terms as failing in some way to fulfill institutionally defined expectations of one or more of the roles in which the individual is implicated in the society. Whatever the complexities of the motivational factors which may be involved, the dimension of conformity with, versus deviance or alienation from, the fulfillment of role expectations is always one crucial dimension of the process. The sick person, is by definition, in some respect disabled from fulfilling normal social obligations, and the motivation of the sick person in being or staying sick has some reference to this fact. Conversely, since being a normally satisfactory member of social groups is always one aspect of health, mental or physical, the therapeutic process must always have as one dimension the restoration of capacity to play social roles in a normal way.

For Parsons, the primary definitional criteria of mental illness are based upon the social role performance of the individual. Parsons (1957:109) acknowledgeds that the etiological factors of mental illness may be biological, biochemical or genetic in anture—in this sense, mental illness is a state of the individual. However, "this state of the individual manifests itself and creates problems for the sick person as well as others in society with whom the individual interacts" (Parsons, 1957:109)—hence, mental illness may also be conceived of as a social phenomenon. In his discussion on the distinction between somatic and mental illness, Parsons (1957:109) states:

In the case of somatic illness the focus of disturbance seems to center in the relations between the body as a system and the personality. Somatic illness is in the first instance an 'intra-individual' phenomenon, and only secondarily a social phenomenon in that the functioning of the individual in his social relationships may be impaired through the ramifications of the central disturbance. Thus the primary focus of an acute infection is impairment in organic functioning... In the case of mental illness...the focus of distrubance is the relations between the personality of the individual and the social system or systems in which he participates. The etiological or diagnostic factors may of course be mainly somatic. The crucial issue, however, is the problem occasioned by the presence in the community of a person in this condition. A mentally ill person is then ... a person who by definition cannot get along adequately with his fellows, who presents a problem to them directly on the behavioral level.

In short, within this model mental illness is conceptualized not merely as a 'condition' but also as a social role—a role characterized by an incapacity of the individual to perform normally—expected roles and ob-

ligations.

Parsons (1958:108) in his discussion on the social genesis of mental illness asserts that mental disorders develop as a result of the inability of the individual to cope with 'strains' imposed upon him in the context of the social process:

There is a set of mechanisms in the operation of which social system and personality aspects are interwoven, which make possible the many complex adjustments to changing situations which always occur continually in the course of social processes. It is when the mechanisms involved in these adjustive processes break down ('adjustive' as between personalities involved in social interaction with each other) that mental illness becomes a possibility, that is, it constitutes one way in which the individual can react to the 'strains' imposed upon him in the course of social process.

For Parsons (1951b:454), the pathogenic strains centre around two elements: (1) the support system of the individual; and (2) the value patterns of the social group. In terms of the former, support may be defined as unconditional acceptance of the individual as a member of the social group—his membership is not provisional upon his behaviour. In terms of the latter, this aspect centres upon the maintenance of the value patterns by the group members. For Parsons (1951b:454) the major sources of strain in social relationships result when prominent group members reject the group norms and values, when members evade their responsibilities for the enforcement of norms, and when the norms must be 'legally' enforced.

Parsons (1951b:455) contends that mental illness may be one manner in which the individual reacts to the various strains placed upon him by society. In this model, insofar as mental illness is conceptualized as a form of deviant behaviour, it is not a unique phenomenon; rather, it represents one type of aberrant behaviour within a larger category of deviance. For Parsons, mental illness merely constitutes one set of alternatives which are open to the individual in response to these strains—other alternatives include the formation of delinquent gangs, the development of religious sects, etc.

In his conception of mental illness as not merely a 'condition' but also a social role, Parsons (1958:117) details four features of this 'sick role:' (1) the individual cannot be held responsible for his incapacity; (2) his incapacity serves as a legitmate basis for exemption from normal role obligations; (3) being defined as 'mentally ill' allows the individual to be able to deviate 'legitimately.' However, this legitimation is provisional upon the person's recognition that to be ill is undesirable—thus, he has an obligation to attempt to regain his health; (4) insofar as the sick person is unable to 'cure himself,' he must therefore seek the help of competent personnel in order to achieve this goal.

Within this framework, the sick role functions as a mechanism of social control. That is, being categorized as mentally ill places the individual in the position of being defined as a patient in need of help. Thus, he is expected to cooperate with the institution and its officials who bestow this definition upon him. In this sense, the sick role serves as a mechanism of social control for it allows the patient to deviate legitimately within institutionally-defined limits, thereby maintaining the equilibrium of the social system.

Moreover, the sick role also functions to isolate the deviant from interaction with other deviants. It prevents him from joining deviant sub-

cultures which present a threat to the stability of the society. The sick role serves to place the individual in contact with re-equilibrating influences:

The role of illness, that is to say, channels those categorized as belonging in it into contact with therapeutic agencies. It is therefore involved in both negative and positive mechanisms of social control, negative in that the spread of certain types of deviance is inhibited, positive in that remedial processes are facilitated (Parsons, 1958:118).

Further, in Parsons' perspective the sick role functions to separate the deviant from the non-deviant. Specifically, this role serves to insulate the sick from the healthy and reinforces the latter not to fall ill. Similar to Durkheim's (1956:103) conception of crime as serving positive functions for society in terms of strengthening normative consensus, so too Parsons (1958:118) views illness in a similar light. In Parsons's view, the sick role serves to reinforce healthy individuals' motivations not to fall ill. For Parsons (1951a:477):

The sick role is...a mechanism which...channels deviance so that the two most dangerous potentialities, namely group formation and successful establishment of the claim to legitimacy are avoided. The sick are tied up, not with other deviants to form a 'subculture' of the sick, but each with a group of non-sick, his personal circle and, above all, physicians. The sick thus become a statistical status class and are deprived of the possibility of forming a solidary collectivity. Furthermore, to be sick is by definition to be in an undesirable state, so that it simply does not 'make sense' to assert a claim that the way to deal with the frustrating aspects of the social system is 'for everybody to get sick.'

While we can credit Parsons for developing a social definition of 10 mental illness, nevertheless, there are many weaknesses in this model.

Specifically, one of the major criticisms of Parsons' model of mental illness centres on its lack of consideration of the patient from a subjective point of view. In this model, Parsons focuses upon the interrelations between actors and the staic patterns formed out of institutionalized expectations that function to constrain the actions of the individuals. However, no attempt is made to study the actions of the actors per se. Moreover, examination of the underlying tenets of his conception of the 'sick role' imply that the patient readily accepts the institution's definition of himself as being mentally ill; further, it implies that the patient wants to cooperate with trained officials in order to regain his health—assumptions made by Parsons without taking into account the feelings of the patient himself. Within this perspective then, mental illness is analyzed solely from the institution's point of view—no consideration is given to the understanding of mental illness from the patient's point of view.

A second weakness of the structural-functionalist model is its failure to examine societal reaction factors in the analysis of mental illness. Specifically, this model centres on the causal factors of mental illness in terms of role strain, to the sole exclusion of the social processes by which individuals come to be defined and treated as mentally ill. The underlying assumption of this model is that certain types of behaviour are per se deviant and are defined as such by the rest of society—in this perspective, societal reactions are conceptualized as being constant rather than problematic in nature. In sum, I would argue that the neglect of

societal reaction factors in the study of mental illness represents a major shortcoming of the structural-functionalist model.

The Existential-Phenomenological Model of Mental Illness

In reaction against the medical model of mental illness, the existential-phenomenological perspective asserts that objective theory and scientific classifications must be disbanded and primacy be given to the individual and his unique experiences in the world. For advocates of this framework, the medical model of mental illness is seen to pose two central problems: (1) it employs a reductive terminology that serves to depersonalize the patient into components; and (2) it utilizes a *vocabulary of denigration* that attempts to provide a pretense of objective neutrality while actually functioning to distance the patient from the patient*s being-in-the-world (Gordon, 1971:53-54). The existential psychiatrist, R.D. Laing (1965:18) poses this central question:

How can I go straight to the patients if the psychiatric words at my disposal keep the patient at a distance from me? How can one demonstrate the general human relevance and significance of the patient's condition if the words one has to use are specifically designed to isolate and circumscribe the meaning of the patient's life to a particular clinical entity?

Rollo May (1967:4) also expresses this concern:

Can we be sure...that we are seeing the patient as he really is, knowing him in his own reality; or are we seeing merely a

projection of our own theories about him?...the crucial question is always the bridge between the system and the patient—how can we be certain that our system, admirable and beautifully wrought as it is may be in principle, has anything whatsoever to do with this specific Mr. Jones, a living, immediate reality sitting opposite us in the consulting room. May not just this particular person require another system, another quite different frame of reference? And does not this patient, or any person for that matter, evade our investigations, through our scientific fingers like seafoam precisely to the extent that we rely on the logical consistency of our own system?

Rejecting the objective theoretical stance and nosological classifactory scheme of the medical model, propoents of the existential-phenomenological model offer in its place, an approach that attempts to grasp sets of elements as totalities—elements which must be understood from the subjective point of view of the patient. As Michel Foucault (1976:45) states:

It is no longer enough to say that the child's fear is the cause of the adolescent's phobias; we must rediscover beneath that original fear and beneath its morbid symptoms the same style of anxiety that gives them their significative unity. Discursive logic is out of place here: it becomes tangled in the threads of delusion and exhausted in an attempt to follow the reasonings of the paranoiac. Intuition goes further and more quickly when it succeeds in restoring the fundamental experience that dominates all pathological processes...At the same time as it reveals in a single gaze essential totalities, intuition reduces, to the point of extenuating it, the distance that constitutes all objective knowledge: the naturalist analysis envisages the patient with a distance of a natural object... Intuition, leaping into the interior of morbid consciousness, tries to see the pathological world with the eyes of the patient himself: the truth it seeks is of the order not of objectivity, but of intersubjectivity.

Advocates of the phenomenological-existentialist model contend that

one cannot merely observe the symptomatology of the patient's behaviour and subsequently bestow a diagnostic label upon him but rather, one must relate the patient's actions to his manner of experiencing the situation. By focusing solely upon the patient's behaviour as 'signs' of some pathological entity, one is already imposing his own being-in-the world onto that of the patient. It is the aim of the existential-phenomenological perspective to achieve a subjective understanding of the world of the mental patient by entering his experience:

...the therapist must have the plasticity to transpose himself into another strange and even alien view of the world... Only thus can he arrive at an understanding of the patient's existential position...What is necessary is to know how the patient is experiencing himself and the world, including oneself (Laing, 1965:34).

In short, for the existential-phenomenologist, "the world of the particular patient must be grasped from the inside, be known and seen as far as possible from the one who exists in it" (May, 1967:56).

When one attempts to examine mental illness from an objective point of view, the world of the mental patient may seem strange and bizarre in nature—a world that is incomprehensible to the outsider. However, when one subjectively enters the patient's experience, he often discovers a meaningful and intelligible world. It is the aim of the existential—phenomenological perspective to achieve an understanding of the patient's experience of his world and himself.

While rejecting the objective theoretical stance of the medical model, proponents of the existential-phenomenological perspective also dismiss its nosological classification scheme. Specifically, although ad-

vocates of this model assert that the etiological factors of certain mental illnesses may be biological or psychological in nature—however, what is important for the existential—phenomenologist is <u>not</u> merely to conceive of mental illness as an organic or psychological pathological entity, but rather to view it in terms of the entire experience of the patient:

...whether its first designations are organic or psychological, the illness concerns the overall situation of the individual in the world; instead of being a physiological or psychological essence, the illness is a general reaction of the individual taken in his psychological and physiological totality (Foucault, 1976:9).

In terms of the strengths of this model, we can credit the existential-phenomenological approach for attempting to achieve an empathetic understanding of the mental patient's experience of his world and himself.

In contrast to the models previously described which seek to understand mental illness from an objective standpoint, the existential-phenomenological model seeks to grasp the world of the mental patient from the inside.

One of the major shortcomings of this approach centres on its nonprocessual nature. That is, proponents of the existential-phenomenological
model focus on the situation of the mental patient—his experiences in
the world, to the sole exclusion of the social processes by which individuals come to be defined and treated as mentally ill, and the consequences
of this social definition for self-attitudes and further participation in
society.

The Social-Role Model of Mental Illness

Advocates of the labelling theory of mental illness, in reaction against the medical model, offer in its place a model which has been termed variously as the 'cultural deterministic approach,' the 'sociological model,' or the 'social-role model.'

In contrast to the medical model which conceives of mental discreters as disease entities located within the individual, the social-role model places primary emphasis on the social processes by which an individual comes to be defined as mentally ill. Subsumed under the heading of deviant behaviour, the social-role model conceptualizes mental illness as an arbitrary label ascribed to certain types of behaviour. In this perspective, mental illness is not the result of any inherent feature which characterizes the mentally ill; rather, it is a social definition bestowed upon individuals as a consequence of certain types of behaviour. As Spitzer and Denzin (1968:2) suggest:

...mental illness is not a function of the content of an individual's acts (his symptomatology), but is, instead, defined by the reactions to his acts and the categorizations of them by those with whom he is in association. Certain behaviors viewed as violating the rules of conduct imposed by various audience members will earn him the label 'mentally ill' and may result in the instigation of steps for the treatment of the 'illness.'

Examination of the social-role model of mental illness reveals that it is comprised of three fundamental components:

- societal reaction to residual rule-violations is the major element in the development of the mentally ill role;
- (2) mental illness is comprised of culturally-conditioned deviant social roles--roles conditioned by the same order of forces as 'normal' roles;
- (3) labelling an individual as 'mentally ill' has important consequences for his attitudes toward self and for his further participation in society.

According to this model, the majority of psychiatric symptoms are instances of what Scheff (1966:33) terms as 'residual deviance.'

In our society, we possess a number of terms for the categorization of many rule violations such as perversion, crime, drunkenness, etc.—each of these terms being derived from the type of norm broken and also from the type of action involved. However, as Scheff (1966:34) suggests, there exists a number of violations for which society provides no explicit classification or label. Mental illness, in this model is conceived as a residual category of deviant behaviour having no clearly defined label.

Advocates of the social-role model contend that residual rule-breaking evolves from diverse sources, such as external stress, psychological sources, organic causes and from willful acts of innovation or defiance (Scheff, 1966:39-47).

Although residual deviance is highly prevalent in our society, a large proportion of it goes undetected and untreated: 16

There is evidence that gross violations of rules are often not noticed, or, if noticed, rationalized as eccentricity. Apparently, many persons who are extremely withdrawn, or who 'fly off the handle' for extended periods of time, who imagine fantastic events, or who hear voices or see visions, are not labelled as insane either by themselves or others. Their rule-breaking, rather, is unrecognized, ignored, or rationalized. This pattern of inattention and rationalization (is) called 'denial' (Scheff. 1966:48).

Yarrow et al. (1955:12-24) in their research concerning the pre-patient phase of the career of the mental patient, document the great capacity of family members, prior to hospitalization, to overlook, minimize and rationalize evidence of mental illness in one of their intimates. Similarly, Sampson et al. (1962:88-96) found that family members frequently develop elaborate mechanisms of accommodation in order to keep the individual within the family. It is only when a crisis occurs, when the behaviour becomes intolerable that this accommodative pattern is broken and the residual rule—violation no longer remains undetected:

The career of the mental patient and his family ordinarily comes to the attention of treatment personnel during the course of an 'unmanageable' emergency...Prior to this public phase of the crisis, the disturbance of the patient is contained within an interpersonal community setting. It is the collapse of accommodative patterns between the future patient and his interpersonal community which renders the situation unmanageable and ushers in the public phase of the pre-hospital (or rehospitalization) crisis (Sampson et al., 1962:95).

While the majority of rule-violations are ignored or rationalized, and thus, are of transitory significance, however, the deviant behaviour becomes visible when the audience reacts toward it. Specifically, when an individual's actions become intolerable, when he can no longer respond accurately to the demands of others in the situation, when a crisis occurs,

family members or official third parties, such as the family physician, police, court officials, or social agencies react to this behaviour and define the individual as 'mentally ill.'

Through the persuasive pressures of these primary and secondary groups, the individual is forced to leave his present setting and move into a psychiatric institution designed to deal with his 'illness.' Sometimes, a person may himself recognize the deviant aspects of his behaviour and voluntarily seek psychiatric help, but more frequently, the individual is hospitalized by others against his will.

When the individual becomes hospitalized, he/she initially undergoes a 'psychiatric examination,' or through what is referred to as 'commitment proceedings.' As Scheff (1964:401-413) suggests, these hearings are quite brief in nature. The decision to commit a person to the mental hospital is largely based upon the presumption that he is mentally ill. That is, by virtue of the fact that laypersons on the outside have diagnosed the individual as mentally ill and have taken him to the psychiatric hospital, psychiatric staff presume that he is suffering from some type of mental disorder. Mechanic (1962:66-75), in his discussion of the commitment proceedings at two large mental hospitals writes:

In the two mental hospitals studied over a period of three months, the investigator never observed a case where the psychiatrist advised the patient that he did not need treatment. Rather, all persons who appeared at the hospital were absorbed into the patient population regardless of their ability to function adequately outside the hospital.

Similarly, Kutner (1962:383-399) also reports a strong presumption of

mental illness by staff members:

Certificates are signed as a matter of course by staff physicians after little or no examination... The so-called examinations are made on an assemby-line basis, often being completed in two or three minutes, and never taking more than ten minutes. Although psychiatrists agree that it is practically impossible to determine a person's sanity on the basis of such a short and hurried interview, the doctors recommend confinement in 77% of the cases. It appears in practice that the alleged-mentally-ill is presumed to be insane and bears the burden of proving his sanity in the few minutes allotted to him...

Scheff (1964:410), in his analysis of the motivation behind this presumption of illness states that it is due to the interpretation of current psychiatric dogma by the examiners—specifically, it is thought that: (1) the condition of mentally ill persons will worsen without psychiatric assistance; (2) psychiatric treatment is generally beneficial for the patient; (3) there are no risks involved in involuntary psychiatric treatment: 'it either helps or is neutral, it can't hurt;' (4) patients are potentially dangerous to themselves as well as others; thus, it is better to risk unnecessary hospitalization than to risk harm to self or others.

According to proponents of the social-role model, the individual enters the 'in-patient' phase of his career upon hospitalization. During the first stages of hospitalization, the patient generally feels betrayal by his significant others, agitated boredom and lonliness. Upon admission, the patient is stripped of many of the rights and accustomed satisfactions he had on the outside—the patient is forced to undergo various processes which serve to mortify his old identity. The institution offers in its

place a redefinition of self as 'mentally ill.' Although the patient may initially reject this new identity, through the constant reaffirmation of the institutional staff, and sometimes even fellow patients, the patient is forced to adopt this definition of self. As Scheff (1966:84) notes, patients are rewarded by the staff for adopting the role of mental patient, and punished for failure to do so:

Ordinarily patients who display 'insight' are rewarded by psychiatrists and other personnel. That is, patients who manage to find evidence of 'their illness' in their past and present behavior, confirming the medical and societal diagnosis, receive benefits.

In short, while the patient may not initially accept this redefinition of self as 'deviant,' due to the rewards and punishments accorded him by the institution, coupled with the shared definitions significant others have of him, the patient is forced to accept the definition *mental patient'-- a generally permanent label and stigma that affects the individual's further participation in society.

Upon release from the mental institution, the patient enters into what advocates of the social-role model refer to as the 'post-patient' phase of his career—a phase generally characterized by stigmatization. Because of the unfavourable definition given to mental illness, discharged patients are stigmatized by society. As Goffman (1963:3) notes:

By definition, of course, we believe the person with a stigma is not quite human. On this assumption we exercise varieties of discrimination, through which we effectively, if oten unthinkingly, reduce his life chances. Scheff (1966:87) and Lemert (1951:434-440) contend that ex-patients are blocked from entry into 'normal roles' once this deviant label has been applied to them—they are faced with discrimination in social, marital and occupational spheres. Although society urges the discharged patient to rehabilitate himself and 'make it out in the community,' it actually functions to block his re-entry into old roles, as well as blocking his entry into new 'normal roles.' In short, society forces the individual to retain permanently the label 'mental patient.'

THEFT

In terms of the strengths of this model, we can credit the socialrole approach for focussing on the social processes by which an individual
comes to be defined and treated as mentally ill. In contrast to previous
models, the social-role model of mental illness centres on the societal
reaction aspect to residual rule violations as the major feature in the
development of the mentally ill role.

In addition to focussing on those who define an individual as mentally ill, we can also credit the social-role model for centring on the world of the mental patient from his point of view. That is, adopting the perspective of the person adjudged to be mentally ill, proponents of this approach examine how the individual reacts to this designation, and how his self-concept is altered after the deviant label has been affixed to him.

The present research adopts the social-role model in its examination of mental patients and mental institutions. Specifically, this study seeks to examine the three phases in the career of the mental patient:

- (1) the pre-patient phase;
- (2) the in-patient phase;
- (3) the post-patient phase.

Conceiving of mental illness as subjectively problematic, this study will focus on the social definition of mental disorders. Adopting the perspective of those who define an individual as mentally ill, this research will examine the circumstances under which an individual is set apart and considered to be disturbed, how the person is cast into the mentally ill role, and how others act toward the individual after he has been redefined.

Moreover, adopting the perspective of the individual designated as mentally ill, this research will also focus on the person's reactions to this label, how he adopts this new role, and to what extent his self-conception is realigned in accordance with the new role bestowed upon him.

In conclusion, this chapter has presented to the reader a discussion of four major models of insanity, their basic assumptions, proponents of the models, followed by an examination of the strengths and weaknesses of each model. The chapter concluded with a discussion of the model adopted for the present study and an outline of the general areas of exploration in the thesis.

The next section will present a formal analysis of the data.

FOOTNOTES FOR CHAPTER THREE

- 1 See Rosen (1968).
- The reader should note that the framework adopted for this discussion does not represent an absolute, all inclusive classificatory scheme of the models of insanity. Others have classified the various perspectives on mental disorders into different conceptual models. See, for example: Siegler and Osmond's (1974) eight model classification scheme; Schwab and Schwab's (1978) trichotomous cheme; and Townsend's (1978) dichotomous scheme. It is beyond the scope of this discussion to deal with all the models of insanity that have been constructed throughout history. This chapter will focus solely on the major models that have specific relevance to the conceptual model adopted by the present study.
- The processual nature of mental health and illness is elaborated by Offer and Sabshin (1966).
- For a detailed discussion of the clinical universalist stance, see: Kiev (1972); Chauncey (1975); Zung (1969); Ciba Foundation Symposium (1965).
- Social scientists have rejected the medical model of mental illness. See for example: Laing (1965), (1969); Sarbin (1969); Scheff (1966), (1967); (1974); and (1975); Braginsky et al. (1969); Foucault (1976); Erikson (1957). Similarly, clinicians also find this model unsatisfactory. See, for example: Szasz (1957a:599-607), (1957b: 405-413), (1960), (1963), and (1970); Adams (1964:191-197); Balance et al. (1970:133-137); Leifer (1970-71:13-21); Sarason and Ganser (1968: 507-510); Engel (1977:129-136); Carpenter et al. (1977:14-20); Strauss et al. (1977); Begelman (1971:38-58).
- Ethnopsychiatrists have criticized the medical model for its failure to focus on social/cultural factors in the examination of mental illness. See for example: Eaton and Weil (1955); Haywood (ed.) (1970); Favazza and Oman (1977); Foulks et al. (eds.) (1977); Ken-

nedy (1974).

- 7 See, for example, Rosenthal (1961); Kringlen (1969).
- 8 See Laing (1965) for a detailed discussion of this area.
- For a discussion of the social definition of illness in general, see: Parsons (1951a: 428-479).
- It is beyond the scope of this chapter to deal with <u>all</u> the criticisms levelled against Parsons' model of illness. The interested reader may consult the following: Gordon (1966); Freidson (1970); Waitzkin and Waterman (1974).
- For a detailed critique of the medical model from an existentialist perspective, see: Cooper (1967:1-14).
- 12 This term is used by Edgerton (1966).
- 13 This term is used by Murphy (1976).
- This term, employed by Townsend (1978), has been adopted in the present study.
- With reference to the application of labelling theory to the study of mental illness, Scheff (1966), (1967), and (1975) has developed the most systematic statement of this orientation. Partial aspects of this perspective can be found in Szasz (1961), (1970); Lemert (1951), (1967); Goffman (1961); Erikson (1957); (1964) and (1966).

- Lemert (1951) refers to unrecognized or residual rule-breaking as primary deviance. Balint (1957:18) terms this behaviour as the unrecognized phase of illness.
- See also, Hollinghead and Redlich (1958:172-176); Cumming and Cumming (1957:92-103).
- 18 See also, Schwartz (1957:271-291).
- For a discussion of the pathways leading to hospitalization see: Smith and Hall (1963:228-233); Mechanic (1962:66-74); Denzin and Spitzer (1966:265-271); Miller and Schwartz (1966:26-35); Lemert (1962:2-20); Goffman (1959:123-142); Hammer (1963:243-251); Wood et al. (1960:632-641).
- 20 See Goldman et al. (1970:427-434); Goffman (1961).
- 21 See Caudill et al. (1952:314-334); Goffman (1961).
- See for example, Philips (1963:963-973); Whatley (1959:313-320); Cumming and Cumming (1965:135-143); Miller and Dawson (1965: 281-287).
- For this tudy, 'data analysis' refers to: "the process which entails an effort to formally identify themes and to construct hypotheses (ideas) as they are suggested, and an attempt to demonstrate support for those themes and hypotheses." (Bogdan and Taylor, 1975:79); See also Glaser and Strauss (1967).

CHAPTER FOUR

METHODOLOGICAL ORIENTATION

The preceding chapters outline the general areas of concern of this thesis, the theoretical orientation upon which the study is based and the concepts employed, and provide a discussion of the relevant literature on the nature of mental illness. The present chapter focuses on the methodological orientation of this research which seeks to understand the world of the mental patient as he/she experiences it. Specifically, it begins with a discussion of the utilization of participant observation as the major form of data collection, followed by a description of the research settings and sample studied. It discusses how access was obtained into the two research settings and the problems faced when attempting to 'make a bargain.' The chapter concludes with a discussion of the researcher's data collection experiences with the patients.

As outlined earlier, adopting a symbolic interactionist perspective, this research centres on learning about the world of the mental patient from his/her point of view. It will be recalled that the symbolic interactionist perspective argues that in order to understand human behaviour, one must take the role of the other. That is, one must attempt to understand behaviour from the actor's point of view. Understanding, in this perspective, requires being socialized into the groups one wishes to comprehend. Only in

this manner is one able to learn the symbols and referents of the group and come to share in its meanings. The data described in this thesis was gathered through participation with mental patients while attempting to understand their world by sharing in their 'definitions of the situation' (Thomas, 1931:41) and 'constructions of reality' (Berger and Luckmann, 1966).

In order to learn the meanings that the patients define as being important an open theoretical scheme was adopted. As Becker et al. (1961:18) state, it is necessary to 'use methods that would allow us to discover phenomena whose existence we were unaware of at the beginning of the research.'

In this present study, the underlying social psychological orientation led to the adoption of a qualitative and inductive methodology, namely participant observation.

Schwartz and Schwartz (1955:91) define participant observation as a process in which:

the observer's presence in a social situation is maintained for the purpose of scientific investigation. The observer is in a face-to-face relationship with the observed, and, by participating with them in their natural life setting, he gathers data. Thus, the observer is part of the context being observed, and he both modifies and is influenced by this context. The role of the participant observer may be either formal or informal, concealed or revealed; the observer may spend a great deal or very little time in the research situation; the participant-observer role may be an integral part of the social structure or largely peripheral to it.

Similarly, Florence Kluckholn, in her classic description of participant observation states that this methodology involves a:

...conscious and systematic sharing, insofar as circumstances permit, in the life activities, and on occasion, in the interests and affects of a group of persons (1940: 331).

In contrast to traditional empiricists who adopt a positivist stance and seek facts or the causes of social phenomena with little interest in the subjective states of the individual actors, the participant observer, adopting a phenomenological perspective, is concerned with 'verstehen' or 'subjective understanding.' For the participant observer, primary significance is placed upon the interpretation of the actors. Reality is not an entity external to individual knowledge as the empiricists believe; for the phenomenologist, reality is what the actors define to be real. As W. I. Thomas (1928) writes: "If men define situations as real, they are real in their consequences." By taking the role of the other, the researcher shares some of the experiences of the participants under investigation. Through symbolic interpretation, the participant observer comes to understand the symbols and meanings that the actors use to organize their world. Much like cultural anthropologists, the aim of the participant observer is to 'penetrate beneath the veil.' Herbert Blumer (1962:188) aptly sums up the role of the participant observer:

To catch the process, the student must take the role of the acting unit whose behavior he is studying. Since interpretation is being made by the acting unit in terms of objects designated and appraised, meanings acquired, and decisions made, the process has to be seen from the standpoint of the acting unit...To try to catch the interpretative process by remaining aloof as a so-called 'objective' observer and refusing to take the role of the acting unit is to risk the worst kind of subjectivism—the objective observer is likely to fill in the process of interpretation with

his own surmises in place of catching the process as it occurs in the experience of the acting unit which uses it.

In sum, the present study is concerned with meaning, with those elements through which mental patients understand, communicate and act within their world. The aim of this research is to grasp the reality of the mental patients as they experience, define and construct it.

As with any methodological approach, there are certain limitations with the methodological approach adopted by this research. In order to obtain ethnographic data about aspects of the patient's social life, rigorous statistical measures were not employed; rather, than beginning with a priori hypotheses and proceeding to test the relationships between variables in these hypotheses, this study focuses upon the collection of descriptive data on the nature of patient social life.

Description of the Research Settings and the Sample Employed in this Study

The present research is based on approximately one hundred and sixty hours of participant observation conducted on two patient populations in different research settings: (1) from September, 1979 to April, 1980 fieldwork was conducted on institutionalized mental patients at a government psychiatric hospital in Southern Ontario, designated here as 'Springville Psychiatric Hospital;' 5 and (2) from January, 1981 to June, 1981 participant observation was conducted on discharged mental patients during weekly therapy sessions in the psychiatry department of a general hospital.

In terms of the former, Springville Asylum was constructed over

a century ago on five hundred acres of farm land situated on the periphery of an urban community. In the early days, the asylum functioned as a self-sufficient community—livestock, poultry, fruits and vegetables were raised on the land with the patients actively taking part in this enterprise. The building complex itself was a four-storey brick structure with heavy mesh screening covering the windows. Patients, segregated on the basis of sex and acuteness of illness were placed in locked wards. The main corridors in the wards were dimly lit; the walls were painted dreary institutional colours. Physical comforts such as draperies, carpets and comfortable furniture were minimal, the emphasis placed on utility rather than beauty. There were only a few heavy oak chairs and benches for the patients to sit upon. Moreover, patient privacy was minimal. Bedrooms designed for one individual, were shared by four patients.

In the early 1960's with the shift in ideology from custodial to therapeutic and humanistic care, the old building was torn down and replaced by a modern facility—an institution with an entirely different character. In contrast to the wards in the old building, the new wards have cheerfully painted walls, draperies, comfortable furniture, numerous lounges, activity areas with televisions, stereos and pool tables. Patient rooms are larger, brighter and have closets to store personal belongings. Generally, while there are four beds in each room, some private rooms are available.

In order to gather data on the world of the institutionalized mental patient, participant observation was conducted on an 'admission ward' in this new psychiatric facility. The ward had a fluctuating patient

population of between twenty to forty patients. In contrast to the old ward system which segregated patients by sex, the admission ward had both males and females, ranging in age from seventeen to mid-seventies. The length of stay on this ward varied from a few weeks to several months, after which they were either discharged or transferred to another ward for further treatment. In terms of the admission status of the patients, the ward contained a mixture of 'first admissions' and 'readmissions,' the majority of which were admitted 'involuntarily.'

Further, in order to gather data on the world of the discharged mental patient, participant observation was also conducted on ex-mental patients during weekly group therapy sessions, arts and crafts sessions and social activities in the psychiatry department of a general hospital. Persons were also interviewed on an individual basis. Specifically, a group of between twenty to forty-five males and females, both young and old were interviewed—individuals who had been institutionalized for varying periods of time and are now living in the community.

Gaining Access Into the Research Settings

This section of the chapter attempts to document the initial stages of the fieldwork experience. Specifically, it focuses on my attempts to make bargains with the gatekeepers of two powerful institutions, namely the professional staff of a large psychiatric hospital, and the staff of a general hospital respectively.

When field researchers conceive of the 'bargaining stage' of research, we often think of it as a rather static phase. The underlying tenet is that the researcher is able to enter a setting, make a bargain, develop rapport with his respondents and no further negotiation is necessary. My fieldwork experience (similar to others) reveals that the bargaining stage is actually a continuous process—negotiation and renegotiation continues throughout the research endeavour. As Haas and Shaffir (1980:245) aptly point out:

What is referred to as the 'bargain stage' of research is more accurately conceptualized as a series of negotiations throughout the research endeavour wherein the researcher continually attempts to secure others' cooperation. Bargain negotiations typically require the development of relationships of equality, involving an idea of exchange, or 'give and take.' This is particularly true in cooperative relationships between professionals.

Examination of the series of negotiations and renegotiations between the gatekeepers and myself reveals that both parties were struggling for superiority. Each party sought to gain control of the situation.

As Becker (1964:272) notes, an irreconciable dichotomy exists between the interests of the researcher and the individuals representing the institution under investigation. Every institution, when under examination seeks to control the outsiders' study of their behaviour. The institution wants to ensure that the results of the study are consonant with its official ideology, hence allowing the organization to remain in a positive light.

I met several problems when attempting to make a bargain with a large, government psychiatric hospital. My interest in studying the institutionalization of mental patients from their point of view was responded to by the gatekeepers in an extremely negative and wary manner.

In order to receive permission to conduct this research, an initial series of meetings occurred between an administrative official in
charge of the educational services of the institution and myself. During
the first meeting, I introduced myself and outlined my research interests
in a general manner:

I'm a fourth year sociology student. As part of our qualitative methods course, each student is required to conduct a study of a social group...I am particularly interested in studying mental patients from a subjective point of view... (Interview #1, October 3, 1979:9).

I continued to explain that the study would be strictly confidential in terms of both the identity of the hospital and the patients involved.

Moreover, the research would be unobtrusive in nature. At no time would the fieldwork interfere with the formal treatment activities of the patients. It was also stressed that the study would be non-evaluative in nature. My aim as a sociologist was not to evaluate the treatment programmes or the general conditions of the hospital; rather, the aim of the proposed study would seek to understand the social world of the mental patient as he subjectively experiences it.

Subsequent to this initial meeting a detailed research proposal was requested and submitted to the gatekeeper providing a description of

the aims, the theoretical and methodological orientation and anticipated significance of the study. This official objected to the proposed project on several grounds:

- (1) the study was too 'broad' and 'vague.' The project lacked rigorous design—it did not possess specific hypotheses that could be empirically tested;
- (2) the proposed study was not valid. It was felt that the study possessed no validity because it was not examining the biological and genetic factors affecting mental illness:
- (3) the unobtrusiveness of the research was questioned—the gatekeeper felt that the patients rights would be violated by such an investigation. The study would undoubtedly be disruptive for the patients;
- (4) the safety of the researcher was a major issue—it was conceived that I might be seriously injured if allowed to conduct research on a psychiatric ward;
- (5) the gatekeeper suggested that the ward staff might object to my presence on the ward;
- (6) finally, the gatekeeper felt that the research lacked significance for the hospital—the hospital would not benefit in any form from a 'sociological' study (Interview #1, October 3, 1979: 11-16).

Although I had expected that my proposal would be critically examined. I did not anticipate the number of obstacles that were placed in front of me. I had no idea that an institution which is an accredited teaching hospital' would be so unsympathetic to alternative types of research involving mental patients. From the outset of the negotiations, very little attempt on the part of the gatekeeper was made to understand the sociological significance of the proposed project. The inductive nature of the research was viewed by the gatekeeper not as a methodological orientation within which one gathers descriptive data on human behaviour; rather, it was interpreted as a lack of knowledge about the nature of mental illness on the part of the researcher. In order to clarify my perspective. I repeatedly emphasized the theoretical and methodological orientation of the research and stressed why it is of utmost importance to begin the study with no preconceived hypotheses in mind. to professionals schooled in the natural sciences who approach the study of phenomena from a positivist stance, my proposal seemed 'too vague' and meaningless.

A second major objection to the research proposal centred on the validity of the study. Specifically, it was felt (by two ward psychiatrists and an administrative official) that my study was invalid because "mental illness or the process of mental illness (was) not being considered." The administrative gatekeeper emphasized that the proposed study lacked validity because it did not focus upon the biological or genetic bases of mental illness. Adopting the medical model of mental illness, the gatekeepers flatly rejected the social-role model of men-

tal illness—the model upon which the proposed study was based. In response to this objection, I argued that while the medical model may have some credibility in terms of explaining the nature of mental illness, it is of equal importance to examine the nature of mental illness from a sociological perspective.

Regarding the issue of patient rights, it was repeatedly stressed by the gatekeeper that my participant observation would undoubtedly upset the patients. Conceiving of sociologists in a negative manner, the gatekeeper told me at one of our initial meetings:

All they (sociologists) do is come in and disturb everyone... From my experiences with anthropologists and sociologists, all they do is get everyone upset. Even clergymen coming in, when they try to talk religion to the patients they just upset them because they (the patients) end up having religious delusions... If you were qualified and experienced with the patients, I might let you in but I'm afraid you would only upset them (Interview #1, October 3, 1979:12).

I assured the gatekeeper that my presence would not upset the patients.

I would only converse with the patients if they so desired. We would talk about subjects that they defined as being important. No attempt would be made to pry into areas they did not wish to discuss.

Another major objection to the proposed study concerned my own safety. Throughout our negotiations it was repeatedly expressed that this study would be potentially 'dangerous' to me. Every attempt was made to point out to me that 'there are three hundred <u>psychotic patients</u> in this hospital.' I felt that the gatekeeper was employing a scare tactic in the hope that I would become apprehensive about conducting the study.

However, I assured the gatekeeper that I was not afraid to conduct field-work in this setting. Moreover, if the hospital was worried about the legal obligations of my safety, I would agree to sign a statement releasing the institution of any liabilities.

Regarding the problem of the anticipated objection of the ward staff, I offered to speak with both the ward supervisor and ward psychiatrist outlining the aims of the project.

One final objection recurred throughout the negotiation process related to the anticipated significance of the proposed study. Specifically, the official gatekeepers repeatedly asked: "What will the hospital get out of this study:" "There doesn't seem to be anything in it for us." In response, I argued that the anticipated significance of the study was threefold: it has significance for sociologists, social psychologists, psychiatrists and all who are interested in obtaining a fuller understanding of social problems of which mental patients are a part. Secondly, this study has practical significance in that certain processes which may come to light as a result of this fieldwork may be taken to a higher level of abstraction and utilized to understand other social groups. Finally, through this study both the researcher and the hospital will gain an understanding of the processes and consequences of institutionalization upon the self-images and identities of mental patients.

The initial series of negotiations, prior to the commencement of field research, lasted approximately one month, with the researcher formulating and reformulating research proposals in an attempt to clarify and 'sell' the proposed study. During this time, a meeting was held with

the ward supervisor, ward psychiatrist and myself in which the theoretical and methodological orientation of the study was discussed. Similar to my earlier meetings with the administrative gatekeeper, these gatekeepers reacted to the study in much the same manner. Two things were suggested to me: (1) that I "go home and rethink the project and come back some other time." or (2) that maybe I should go and study some other group outside this institution such as an ex-mental patient group in the community. Despite the procrastinations, I did not discourage. After presenting the administrative gatekeeper with an extensive, revised research proposal providing solutions to each objection raised, a bargain was finally made. However, still in one final attempt for the institution to control the situation, the gatekeeper suggested that he be given access to my fieldnotes in order to 'give direction' to my research.

....I think that you will need some guidance. You know Nancy, you could easily get lost in this institution. I don't mean physically, but you could be out there (he points to the air) trying to collect data but in the process you get lost in the shuffle. I think that I should be your advisor. You should report to me once a month with your notes so that I can direct your research. If you are going down a wrong path, I can set you straight. (Interview #5, October 15, 1979:6).

In response to the gatekeeper's request for access to my fieldnotes I emphasized that as an ethical researcher I had an obligation to protect the patients under investigation. Under no circumstances would the gatekeeper be allowed to examine my fieldnotes—they were strictly confidential and would remain the sole property of the researcher. I expressed

the point of view that sharing the fieldnotes would be a violation of the researcher's code of ethics, specifically protecting patient confidences and anonymity. Through the 'give and take' process researchers and gate-keepers engage in when bargaining it was finally agreed that the field-notes would remain the property of the researcher, but I would meet with the gatekeeper on a regular basis to discuss my research in a general 13 manner.

After this arduous series of negotiations during which both parties continually struggled for superiority a bargain was achieved. I finally began to conduct fieldwork on the ward. Although I thought that negotiations had been completed at this point, this was far from being the case for two and a half weeks later I was notified by the administrative gatekeeper that a meeting was scheduled to 'discuss my proposal'—this time the meeting would involve the ward supervisor, ward psychiatrist, administrative official, my advisor from the university, and myself.

Further, I was also informed that I was suspended from conducting field—work until after this meeting. During this meeting, we once again discussed the theoretical underpinnings of the proposed study and the methodology. Addressing my advisor, one of the gatekeepers expressed his central concerns and reservations about the study:

In our previous discussion with Nancy, I expressed concern with her study on the grounds that it was abstract and possessed certain contradictions. I find it difficult to deal with this notion of 'understanding'....(Interview # 10, November 13, 1979:3).

The gatekeeper felt that one could not get at an understanding of what it is like to be a mental patient from the type of study proposed. He argued that the study would just be detailing what a 'researcher thinks it is like to be a patient.' Moreover, the gatekeeper stated that this study could not help but be evaluative in nature -- the researcher would not be objective. In response to these areas of concern, my advisor argued that our aim as social scientists is to understand. We were not interested in evaluating the conditions of the setting or the treatment approaches. He explained that we were not journalists interested in publishing our findings in popular magazines or newspapers. We instead were scientists, attempting to understand social phenomena. He further pointed out that we were interested in examining social processes: my study would be an attempt to examine the processes of institutionalization and its effects on patient self-images. The study would be an attempt to look at reality in terms of what the patient, individually and/or collectively, define as important and real.

However, the gatekeepers (particularly the ward supervisor and ward psychiatrist) still viewed my study in a negative manner. My advisor asked the gatekeepers if they had ever read any sociological studies which adopt an interactionist perspective. As was anticipated, they had not. As a result, I handed the gatekeepers a book of readings that adopted this theoretical stance along with a brief introduction to symbolic interactionism. In a rather reserved tone, one of the gatekeepers replied:

Well, I will have to examine these ideas further. Maybe through further reading, I will come to grips with this... (Interview #19, November 13, 1979:4).

A second gatekeeper concern centred on the learning objectives discussed in the proposal—the same proposal which was agreed to by the administrative gatekeeper two weeks prior to this meeting. Once again, the gatekeepers wanted to know how the 'hospital would benefit' from the study. I described the theoretical and practical significance of the project pointing out that both the hospital and myself would undoubtedly learn about the processes and consequences of institutionalization upon the patient's self-concept. Because of the institution's interest in 'getting something out of this research,' I offered to give a seminar to the ward staff upon completion of the study, outlining my findings in a general manner. This offer was responded to by the gatekeepers in a favourable way.

A third concern was expressed concerning my role on the ward. One of the gatekeepers stated that the ward staff were confused about my identity and the objectives of my study:

The staff are confused about your role, what it is that you're doing here. You have to realize that you're different, Nancy. You are not the conventional student coming onto the ward. We usually have nursing students, medical students who have specific objectives to meet...We have a very regimented structure here...Your proposal is very different (Interview #10, November 13, 1979:6).

In response to this area of concern, I stated that, in violation of our original agreement in which both the administrative gatekeeper and the ward supervisor agreed to inform the staff about my presence and the aims of the study, nevertheless, they were confused about my role; hence, I

was being treated very coldly by the staff. Essentially, the staff thought I was conducting an evaluative study of them, despite my efforts to explain what I was actually doing. And this placed me in a precarious position—I was afraid that my interacting with the staff would seriously harm my rapport with the patients. It was agreed by the gatekeepers that this situation would be cleared up immediately—one of the gatekeepers would personally ensure that every staff member was informed. (Unfortunately, this promise was not carried through for throughout the entire fieldwork experience, I was continually faced with individuals for which I had to explain my role and objectives). 15

One final concern was expressed by one gatekeeper, specifically related to the problem of 'transference.' The gatekeeper stated that he had reservations about the manner in which data would be collected—it was felt that, conducting a study from a subjective point of view, participating with the patients and coming to an empathetic understanding, would undoubtedly result in myself 'going native' or associating myself with the patients. I assured the gatekeeper that this would not happen; while the participant observer actively participates in the lives of his informants, at the same time he has the capacity to remain detached and objective.

After long deliberation, I was once again granted access into the hospital. Walking out the door, I remarked to my advisor that I never thought that the bargaining stage would continue throughout the research endeavour. One of the gatekeepers, overhearing this statement replied in a half-joking manner:

This is only the beginning Nancy. I told you this wasn't going to be easy (he chuckles). You ain't seen nothin' yet! (Interview #10, November 13, 1979:8).

Ignoring this warning, I assured myself that I was finally allowed access into the research setting—the bargain had been made for good. I was sure that no further negotiation would be necessary between the gatekeepers and myself, but I was wrong.

Trouble loomed once more about two months after this second bargain was made. I was informed by the administrative gatekeeper's assistant that she had received an alarming phonecall from an anonymous source regarding my research:

Nancy, I had this call yesterday from an hysterical woman who was saying some disturbing things...This woman said that your reports (fieldnotes) were being given out to the thirty people in your class and that they were also available to anyone else who wanted them. She told us that these reports were derogatory to the hospital—they stated that there were all kinds of suicides and escapes going on that no one knows about...This woman expressed that she didn't want to phone us but her conscience was bothering her and she waid that we have to stop you (Interview #14, November 27, 1979:1).

I was shocked when I learned of this news. In response to these allegations, I emphasized to the gatekeeper that the notes were <u>not</u> being distributed to 'any one who wanted them.' The only individuals allowed to see my fieldnotes were my professor, myself and my fellow classmates who had taken an oath of confidentiality. I pointed out that at present I had no reason to believe that anyone had violated this oath. I stressed that:

We can't deal with hearsay evidence here. My fieldnotes are not an attempt to damage the hospital in any way...I am not interested in conducting an expose of the hospital. I think that this phonecall represents an effort to damage myself, rather than to damage the hospital (Interview #14, November 27, 1979:1).

The gatekeeper agreed with this point but emphasized that because the anonymous phonecaller specifically identified my name and the research class I was in, this phonecall could not be disregarded. The gatekeeper's assistant wondered if the phonecaller was one of the patients that I was interviewing who "became upset with my questioning." But because the caller gave my full name, knew details about the qualitative methods course in which I was enrolled, and asked for the administrative official with whom I was bargainimg, the gatekeeper's assistant felt (as I did) that this person was probably someone from my class who was attempting to sabotage the research. Nevertheless, the institution could not ignore this phonecall—this call indicated that their identity was being threatened. In response to this call, the institution made a serious, forceful attempt to gain control of the situation.

Specifically, it was suggested that I discontinue my fieldwork until a meeting could be set up between the administrative gatekeeper and myself at which time a decision would be reached regarding if the project should be allowed to continue. I argued that it would be impossible for me to stop conducting fieldwork at this time because it would seriously affect the rapport I had established with my subjects. Reluctantly, the gatekeeper's assistant allowed me to continue the study in the interim.

In the following week, I met with the administrative official one more time. During our earlier negotiations, an attempt was made on the part of the gatekeeper to establish a somewhat collegial relation—ship with me; however, from this meeting onward, a gradual breakdown in collegiality and cooperation developed. The gatekeeper responded to my research in an extremely negative manner. In an attempt to convince the gatekeeper that the allegations made against my research were simply not true, I stated:

I explained to him that careful examination of this unfortunate incident indicated that someone in my class deliberately set out to undermine my fieldwork. I was aware that some of my fellow classmates were jealous of my research and earnest efforts and even made threats that they were going to sabotage the study, but I never thought that they would follow through on their threats. I emphasized to the gatekeeper that:

...this call was merely aimed at getting me into trouble. What I'm trying to say is that you can't see this incident as an effort to damage the hospital, but rather, it repre-

sents a blatant effort to damage myself and my professor; we are clearly the victims here! (Interview #16, December 4, 1979:4).

.....

The gatekeeper was not convinced by this explanation and offered an alternative interpretation of the incident:

This is all very interesting, I mean, this idea that people in your class made this phonecall, but I have a different perception of this incident. I think that your advisor and you are missing the point. You are attempting to rationalize the irrational. I think that this is one of the downfalls of sociology. My perception of all this is that if a person is doing this to you in class, he/she must be mentally ill. Perhaps the way in which you are discussing the topic of mental illness in your notes is upsetting this person. I try to listen to what the irrational is saying to me. I see this incident as an irrational person's way of crying for help. They do this by phoning the hospital. Obviously, there is something in your fieldnotes that is wrong; you are probably missing talking about what constitutes mental illness.... I'm not interested about who in your class did this. What is important is this 'message' and I'm going to have to follow it. When I get messages like this, they are usually telling me something important (Interview #16, December 4, 1979:4).

To ensure strictest confidentiality, I informed the gatekeeper that, from this point on, no one would receive further fieldnotes. I would only discuss my research with my advisor.

The gatekeeper in another effort to gain control over the research requested that he be given access to my fieldnotes. I flatly denied this request on the basis of an ethical obligation to protect my subjects. Realizing that I possessed this strong conviction to protect tect my informants, the gatekeeper reluctantly changed the subject. He

He handed me a counter-proposal written by two psychiatrists who had critiqued my original research proposal -- the same proposal to which one of the two psychiatrists had agreed two months earlier. This counterproposal presented a critique of my entire theoretical and methodological framework. Specifically, the psychiatrists expressed two central concerns: (1) they felt that the study would undoubtedly be evaluative in nature. It was expressed that "participant observation studies are affected by the biases of the investigator more than other kinds of studies and the 'facts' emerging from such a study could be merely affirmation of the biases." (November 27, 1979:1). (2) they felt that the study was invalid because it was not looking at the subject of mental illness. As a result of these concerns, it was requested that I only conduct fieldwork during recreational and self-care activities. I was no longer granted permission to conduct participant observation during formal therapy sessions -- a violation of one of the conditions of our original bargain. Having no choice in the matter but to abide by the gatekeeper's request, I agreed to this alteration of our original agreement.

Examination of this entire bargaining experience from an interactionist perspective reveals that the institution attempted to 'discourage' me from conducting this research from the outset; however, through persistence, I was reluctantly granted access. Throughout our negotiations and renegotiations, the institution's representatives struggled to gain control over the situation—especially in this latter attempt to gain control of the researcher's fieldnotes. While I managed to successfully resist, the institution did not concede. Although the institution was not able to con-

trol reality in terms of exerting control over the <u>nature</u> of the data being collected, its representatives still managed to control me 'spatially'—that is, they attempted to control the <u>areas</u> and <u>activities</u> which I was allowed to observe.

From a labelling perspective, this bargaining experience can be conceptualized as an on-going struggle of the researcher against being negatively labelled by the gatekeepers of the psychiatric institution. Specifically, the gatekeepers equated the inductive nature of the research with a lack of knowledge on the part of the researcher, thereby attempting to label me as an 'incompetent researcher.' Moreover, possessing the misconceptions that sociologists conduct 'expose-types' of research, and that the researcher was going to conduct such a study that present the hospital in a negative light, these gatekeepers attempted to attach to me the label, 'troublemaker.' Through my persistent efforts, I attempted to prove my competence as a qualitative researcher by justifying the methodological orientation of the study. I emphasized the non-evaluative nature of the study-I was not interested in 'exposing the conditions or treatment facilities of the institution.' While my efforts partially convinced the gatekeepers of the clarity and rigor of the research design, thereby somewhat dispelling their image of me as an 'incompetent researcher,' nevertheless, they still conceived of me as a 'troublemaker.' As a result, the gatekeepers attempted to control when and where I was allowed to conduct my participant observation.

During subsequent weeks, I conducted fieldwork as many times as possible. I sensed that the hospital was attempting to do something in

order to terminate this research. I was correct, for I received by mail a new research proposal (written by the hospital) detailing an entirely new set of conditions to which I was advised to agree. No attempt was made to explain why such a new proposal was developed. A note was attached to this proposal stating that both my advisor and myself were to sign this agreement and return it immediately. What it would be too lengthy to discuss this new proposal in its entirety, suffice it is to say that the new additional conditions were simply unacceptable—I was being further restricted in the hospital setting which would make it extremely difficult, if not impossible, to achieve the aims of my research. The document stated that from this point on I was to call the ward to arrange a weekly schedule with the ward supervisor when it would be 'convenient for the ward to have me.'

Moreover, this new proposal stated that, "all records (field-notes) will be viewed as the <u>property</u> of Springville Psychiatric Hospital and will be treated with respect and discretion"—a condition that I could not agree to under any circumstances.

Further, in another effort to hamper the research process, they drafted a statement downgrading my capabilities and the quality of the research:

The image of the hospital will be <u>maintained</u> and any discussion that fails to emphasize the <u>strengths</u> of the hospital or tends toward <u>sensationalism</u> will be avoided.

A final condition of the proposal stressed that the paper resulting from this fieldwork would not be published. Under the conditions

of our original bargain, it was agreed that while my immediate aim was not to publish this study, however at some later date should I decide to do so, I would allow the hospital pre-publication criticism rights.

In response to this new proposal, I formulated a letter criticizing every element that was unacceptable. I stressed the fact that the institution reneged on our original bargain and attempted a formal fait a accompli which represented a serious breach of professional ethics. I emphasized that such a unilateral power play was offensive to both my advisor and myself. I pointed out that I tried to cooperate with the hospital submitting numerous research proposals and that I was subsequently granted access and 'cooperation' from the institution. The letter was concluded by stating that this set of violations of the original agreement and the attachment of a new set of conditions reflected "unprofessional and unethical conduct unbecoming of representatives of an academically-affiliated institution."

Upon receipt of my letter, the administrative gatekeeper suggested that another meeting take place—this time among the gatekeeper, my advisor, myself and the medical director of the institution. I was emphatically told that:

This is not meant to sound like a threat...but we have to get this letter of agreement signed (the hospital's new proposal) or we'll be forced to terminate your placement! (Interview #32, March 4, 1980:2).

The institution was asserting its authority—its representatives were no longer interested in negotiating. They issued an ultimatum: either

I would agree to the conditions outlined in the new proposal or be forced to leave the setting. I received this ultimatum after seven months in the field during which time I managed to collect a wealth of information regarding the institutionalization of mental patients. Thus, it would not seriously affect the research if the fieldwork was terminated at this point. After discussing the situation with my advisor, it was mutually agreed upon that the best strategy would be for me to leave the field. I phoned the administrative gatekeeper and informed him that my fieldwork was now complete and therefore it was not necessary to have another meeting. In response, he argued:

Well, I think that there is a reason for this meeting...
There are so many things that need to be cleared up at this meeting...We have to clear things up regarding publication.
Nancy, you may think that you've finished, but we have to get a letter of agreement saying that you won't publish.
Right now, we have no control over you and I don't like that... (Interview #33, March 26, 1980:1).

This was a final attempt on the part of the institution to gain control over the situation. I stressed that according to our original bargain it was stated that my immediate aim was not to publish; however, if I should decide to do so, I would allow the hospital pre-publication criticism rights. I emphasized that I had cooperated with the hospital at all times. I had fulfilled every obligation of our original bargain. My fieldwork was now completed; therefore, I saw no reason for further negotiations. I thanked the gatekeeper and said goodbye.

In summary, the preceding discussion focussed on my negotiations

and renegotiations with 'powerful people' in a 'powerful institution.' l8

From an interactionist perspective, it represents the continuous struggle between the researcher and the gatekeepers, each party attempting to gain control over constructions of reality and definitions of the situation.

From a labelling perspective, it represents an on-going struggle of the researcher against being negatively-labelled by the gatekeepers as an 'incompetent researcher,' a 'troublemaker' who through her research would undoubtedly destroy the positive image of the hospital.

THE STATE OF T

This bargaining experience lends support to the view that the research bargain is more accurately conceptualized as a continual process that occurs throughout the entire research endeavour (see also, Haas and Shaffir, 1980:245). While this long series of negotiations and renegotiations was sometimes discouraging and anxiety provoking for the researcher, in retrospect, this experience was highly instructive about the problematics of negotiations with 'powerful people' in 'powerful institutions.'

In contrast to the difficulties experienced in making a bargain with gatekeepers of a psychiatric hospital and the extended period of time during which negotiations and renegotiations occurred, I was granted relatively easy access into the second research setting, the psychiatric department of a general hospital.

Specifically, during an initial telephone conversation between a psychiatrist gatekeeper and myself, I identified myself and explained that I was in the process of conducting research on the processes and consequences of institutionalization upon the self-images of mental patients. I pointed out that I had collected data on institutionalized patients, and at present

was interested in gathering data on discharged mental patients:

I'm a graduate student in Sociology at McMaster. Over the past year, I have been conducting a study on the processes and effects of institutionalization on the identities of mental patients. Up to this point, I have been focussing on hospitalized mental patients; in order to make my study complete, I need to gather additional data on ex-patients who are back in the community (Interview #35, October 27, 1980:2).

THE STREET

The psychiatrist responded to my research in a positive manner and suggested that we meet to discuss the project in detail.

During this meeting, I discussed the objectives and methodology of the research—I emphasized that the study would be unobtrusive and non-evaluative and I would ensure the confidentiality of the ex-patients. The psychiatrist approved of the study but indicated that the staff in the department of psychiatry would also have to approve of this research prior to its commencement. He felt that it would not be necessary to negotiate a bargain with the Ethics Committee of the hospital; rather, it would be sufficient to submit a research proposal to the psychiatry department for their approval.

However, approximately two weeks after submitting this proposal, I received a telephone call from the psychiatrist informing me that my research proposal had been discussed at a staff meeting during which time it was pointed out that I must gain approval of the Ethics Committee prior to beginning the study. Apologizing for the bureaucratic red-tape of the institution, the psychiatrist said:

Well, I'm afraid that you're in for some time-consuming activities. You'll have to appear in front of the Ethics Committee to answer their various questions, and you must submit to them a copy of your research proposal. The problem is that the committee only meets once a month, so you will be delayed in beginning your study. I realize that this is an inconvenience since you are on a time schedule (Interview #37, November 12, 1980:1).

TO TO TOTAL STATE

He stated that he would send me an application form for permission to conduct fieldwork which I was supposed to submit, along with a copy of the research proposal to the Director of Medical Services of the hospital. The psychiatrist also offered to send a letter to the Ethics Committee stating his full approval of the project.

Approximately three weeks after submitting the form and the proposal, I was granted an interview by the Ethics Committee. During this brief meeting, I outlined the areas of concern of the study, its methodology and anticipated significance. I expressed that I wanted to conduct participant observation in weekly group therapy sessions as well as interviewing individuals on a one-to-one basis. I also emphasized that patients would be selected on a voluntary basis and would be informed of the full aims of the study. The resulting data would be collated without patient names. After finishing my spiel, a member of the Ethics Committee raised one central concern: she felt that I should have patients sign a consent form prior to interviewing them on an individual basis—a recommendation to which I agreed. With no further questions asked or concerns raised, the chairperson of the Committee thanked me for coming and indicated that I would be notified of the Committee's decision regarding the approval of the project.

Three weeks after this meeting, I received in the mail a carbon copy of a letter addressed to the psychiatrist with whom I initially discussed my research, indicating that "the project was considered by the Research Committee at their December meeting....(and) recommended that this project be approved following (the psychiatrist's) approval for the same." The letter further indicated that, prior to my proceeding, it was necessary to send to the Director of Medical Services a copy of the consent form I was going to use. Upon receipt of this letter, the psychiatrist sent a letter to the Committee issuing his full approval of the research. I subsequently sent a copy of the consent form that I was going to utilize. Having sent both these items to the Ethics Committee, a bargain was finally achieved.

Learning the Ropes-Researcher Data-Collection Experiences with the Subjects

One of the major goals of the researcher is to become a well-socialized member of the group he wishes to understand. Only by taking the role of the other will the researcher come to comprehend the behaviour of the social group under investigation. In this section of the chapter, I will discuss my data-collection experiences with the patients, specifically focussing on some of the fears and problems encountered during my first days in the field 20 and how I managed to 'learn the ropes.' 21

As Sanders (1980:168) notes, learning the ropes of fieldwork results when the researcher: (1) possesses an understanding of the physical and social structures of the setting and the activities occurring within that setting; (2) feels comfortable interacting with the subjects;

(3) acquires the ability to use the group's language; and (4) possesses the ability to know when his subjects are lying, being facetious or attempting to mislead him.

Examination of the literature written on qualitative methodology reveals that frequently little reference is given to the data-collection stage of research. Because of this neglect, one is often led to believe that learning the ropes is either a process occurring prior to the commencement of the 'real fieldwork.' or is a process occurring simultaneously with the 'real fieldwork.' However, as Kleinman (1980:171) correctly emphasizes, "learning the ropes of fieldwork is not only coincident with doing the 'real work' of field research...but also a part of that process." There is analytical value in examining how a group responds toward the investigator; conversely, it is important to examine the researcher's feelings toward his subjects. During the initial stages of fieldwork, researchers may experience feelings of self-consciousness, discomfort or threat-important data for it allows the researcher to remain aware of his preconceptions, biases and subsequent role alterations as he begins to learn the ropes. In short then, learning the ropes is an integral part of the researcher's analysis of field data.

Fieldworkers learn the ropes by actively participating in the lives of their subjects, by observing, conversing with them, participating in various activities—a process termed as 'hanging around.' 22 During the twelve months of research, I participated in and observed as many activities of hospitalized mental patients and discharged patients as possible. I attended various group therapy sessions, participated in special social functions, such as in-patient dances and parties. I also joined hospital—

ized subjects during their leisure time at the patient canteen. I participated in various patient leisure activities such as listening to the stereo, watching television, playing pool, etc.

During the first days of 'hanging around' the researcher often feels uncomfortable and anxious. As Geer (1967:383) notes, these feelings result more frequently from preconceived ideas rather than actual problems:

We do it on each new study. We underestimate people's trust in our neutrality, their lack of interest...And we project theoretical problems into the field. Because the process of group formation is difficult to conceptualize, we suppose it will be difficult to observe. We expect ephemeral, unstructured situations...to appear incoherent.

I felt anxious and uneasy during my initial days of fieldwork.

My anxieties were derived from two central concerns: (1) the anticipated fear ²³ of being rejected; and (2) intense self-consciousness in the field.

In terms of the former, I imagined that the patients would not welcome my study—they would not want to 'open up' and discuss private experiences in their lives with a stranger. As Sanders (1980:16) correctly notes, we are socialized into thinking that it is illegitimate to violate others' norms of privacy. By so doing, we risk being rejected. Being reluctant to violate the patients' privacy for fear of rejection, I initially entered the role of a passive researcher. I did not ask ask personal or probing questions. Conversations were kept at a superficial level. However, as I gradually became more comfortable with my informants (and they with me), I overcame this imagined fear. I found

that the patients welcomed my questioning-they were willing to share

personal details of their lives with a researcher who listened and sympathized with them. As one patient told me:

It's really nice to talk to you—you listen to us—you take us seriously—no one else listens to us around here. (Interview #7, October 25, 1979:3).

In terms of the second problem of self-consciousness, researchers initially feel self-conscious when entering a new and unfamiliar social setting. 24 During my first day in the field, I felt as if everyone was watching and evaluating my behaviour. I felt embarrassed when I asked a 'stupid question.' However, I found that the patients did not make light of my ignorance; rather, they attempted to teach me what I did not know. Through interaction with the subjects and the development of rapport, I soon overcame this feeling of self-consciousness.

Explanations of My Study and Role as a Researcher to the Patients and their Responses

Prior to beginning a study, the researcher, guided by ethical considerations must decide whether to engage in overt or covert research. While some researchers support covert research, the majority of investigators oppose it.

Rejecting concealed research in which the aims of the study are unknown to the subjects and the investigator is taken to be someone else (usually one of the subjects), at no time did I pose as a patient; neither did I pose as a staff. I fully informed the patients and the staff of my

role as a researcher and the objectives of the study. During my initial meetings with the patients, I stated:

My name is Nancy. I'm a sociology student from McMaster University. As part of a course requirement, I'm here doing a study on what it is like to be a mental patient. I'll be coming to visit you for a few hours each week to talk about whatever you want—things you think are important for me to know (Interview #8, October 28, 1979).

I emphasized to the patients that whatever information was told to me would be kept in strictest confidence. I assured the patients, who feared reproachment and punishment from the staff, that all our conversations would remain confidential. Upon making these conditions of my research explicit, I soon developed rapport with the patients and gained their trust.

From the outset, the patients responded to my research in a positive manner. Much to my surprise, they were extremely open about private experiences in their lives. They enjoyed having a sympathetic person listen to them, a person who took their problems seriously. During my weekly visits to the hospitalized mental patients, the participants would eagerly await my arrival, standing at the front door. When I arrived, the patients would frequently argue among themselves regarding who would be first to talk about their experiences with me.

In addition to talking about their own lives, the respondents were interested in learning about my identity. I anticipated that they would ask me numerous questions about my research, but in contrast, they concentrated their questions on aspects of my identity, such as my background, academic

interests, personal likes and dislikes—questions which I readily answered.

By being open with the subjects and 'giving information' and not solely

'taking it' from them, both parties got to know each other and learned to

feel comfortable in each other's company.

In summary, by working hard, by respecting the subjects and taking their views seriously, the patients responded by teaching me the ropes—I achieved a subjective appreciation of the world of the mental patient—an appreciation that could not have been achieved by another methodology.

FOOTNOTES FOR CHAPTER FOUR

- For a discussion of the methodology of participant observation and qualitative methodology in general, see: Cicourel (1964);
 Bruyn (1966); Bogdan and Taylor (1975); Babchuck (1962:225-228);
 Adams and Preiss (eds.), (1960); Vidach (1955:354-360); Filstead (1970).
- For a detailed comparison of the traditional empiricist and phenomenonological perspectives, see: Bruyn (1966); Bierstedt (1949:584-592).
- This phase was coined by the anthropologist Franz Boas who argued that one must understand a culture from the subjective point of view of the natives themselves without imposing the researcher's Western framework of rationality.
- For a discussion of the limitations of this methology, see: Dean (1967); Becker (1958:652-660).
- In order to protect the identity of the hospital, a pseudonym has been employed.
- At this hospital, the patients were either classified as 'voluntary' or 'involuntary.' The former status refers to patients who admitted themselves into the hospital on a voluntary basis. These patients were allowed to discharge themselves when they no longer felt like being hospitalized. The latter status refers to patients, who through court orders or psychiatric diagnosis at a general hospital, were admitted into this psychiatric facility. That is, these patients have no choice in the matter—they are admitted on the basis of someone else's judgment that they are in need of psychiatric care. Involuntary patients are discharged at the discretion of the ward psychiatrist and other staff members.

- For a discussion on making bargains with gatekeepers see: Bog-dan and Taylor (1975:30-32); Becker (1970).
- 8 See for example, Haas and Shaffir (1980:244-255); and Geer (1970: 81-96).
- Bogdan and Taylor (1975:34) suggest that it is the best strategy to outline the study in <u>general</u> terms. "Observers needs not explain their substantive or theoretical interests or their specific techniques in great detail. In fact, it is probably unwise for researchers to volunteer elaborate details concerning the precision with which notes will be taken."
- This, and all subsequent quotations were taken from the researcher's fieldnotes of informal interaction with the staff and patients and were transcribed as nearly verbatim as possible. The names have been changed to protect the identities of the subjects.
- Catekeepers often ask how they (the institution) will benefit from the research project. This question was repeatedly asked of me during my negotiations and renegotiations with the gatekeepers. Researchers (both anthropologists and sociologists) are often criticized for going into the field, collecting data on aspects of human behaviour, leaving the field and then writing up their data—with—out giving the people under investigation anything in return. Due to this fact, coupled with the persistence of the gatekeepers, I felt an obligation to give the institution something in return; thus, I offered to give the institutional officials a copy of the resulting study as well as a presentation of my findings to the ward staff, in order that both parties would benefit from the research experience.
- This action is an illustration of an institution's attempt to 'pass the buck.' Because the research project was incompatible with the interests of the institution, its representatives actively attempted to 'channel' my interests elsewhere.

- For a discussion of this 'give and take' of exchange process, see: Johnson (1975).
- See Haas and Shaffir (1980:244-255) for a similar discussion.
- Every time I went to do fieldwork on the ward, I was forced to explain my identity to each of the staff members. The ward staff operated on a rotation basis whereby each month they were transferred to another ward; thus, each time I arrived, I was confronted with an entirely different set of 'ward gatekeepers' who were unaware of my identity and the aims of the study. As a result, I was forced to spend time explaining who I was and the objectives of the research to these staff members—actions which threatened to destroy the trust of the patients that I had worked so hard to establish. Specifically, my interaction with the staff was interpreted by the patients as being traitorous and it was only through perseverance that I was able to regain my informants' trust.
- For a discussion on the subject of 'going native' or 'over-rapport,' see: Spradley and McCurdy (1975:60); and Miller (1952:97-99).
- 17 See Bruyn (1966); and Schwartz and Schwartz (1955:343-354).
- For a discussion regarding similar problems of gaining access, see the following: Barber (1973:103-112); Habenstein (ed.) (1970); Haas and Shaffir (1978); West (1980); Hoffman (1980).
- As the reader can see, access was obtained into the second institution much more smoothly and expediently. The reasons for this relatively easy access were two-fold: (1) in the second institution, I was 'sponsored' by a member of the psychiatric profession rather than attempting to 'sell' the study on my own; (2) the psychiatrist at the second institution was more sympathetic to a sociological approach to the study of mental patients, for he himself was actively involved in psychiatric research that included social psychological and sociological factors in the analysis. In contrast,

the psychiatrists at the first institution advocated a strict traditional approach to the study of psychiatry and psychiatric patients—a stance that was antithetical to my sociological study. For this reason, coupled with the former, I had much more difficulty in obtaining access and maintaining relations in the first research setting in comparison to the second setting.

- See Blanche Geer (1967:372-398) for a detailed description of a researcher's first days in the field.
- See Shaffir et al. (eds.) (1980:111-116) for a discussion of this subject.
- This term has been employed by Becker et al. (1961); Gans (1962); Whyte (1955) and Glaser and Strauss (1967).
- Coupled with the fear of rejection, researchers are also often afraid that their subjects may cause them physical harm. In my case however, I never felt afraid of the patients I studied. I rejected popular stereotypes of mental patients as being individuals who 'don't know what they are doing,' and who are 'extremely dangerous.' My father was employed at a psychiatric institution for a number of years during which time I frequently interacted with patients—interactions which dispelled any popular conceptions I may have previously had of mental patients.
- 24 See Sanders (1980:160); and Bogdan and Taylor (1975:42-43).
- 25 See for example: Douglas (1976)...
- Such researchers as Erikson (1965); Gold (1958:221-222); and Davis (1961) oppose conducting covert research.
- 27 See Rosenhan (1973:250-258); and Goldman et al. (1970:427-434).

- 28 See Goffman (1961:ix).
- I felt an ethical obligation to answer patient questions posed to me in the course of my imterviewing. I felt that it was not fair of me to ask questions of them (many of which were personal in nature) and then refuse to answer questions they had of me.

CHAPTER FIVE

THE PRE-PATIENT PHASE IN THE CAREER OF THE MENTAL PATIENT

This chapter, based on my data analysis of eighteen subjects introduces the first stage in the career of the mental patient—the pre-patient phase. Specifically, it examines the social conditions which may lead to hospitalization in a psychiatric facility. Conceiving of mental illness as subjectively problematic, this discussion will focus on:

- (1) the social conditions upon which an individual is adjudged to be 'mentally ill' and is segregated from the community—the social audience who react to certain types of rule violations and redefine the individual as mentally ill;
- (2) the types of rule violations against which the audience react and take action;
- (3) the reaction of the individual adjudged to be mentally ill-his/her social experiences prior to hospitalization.

Pathways Leading to the Mental Hospital

Mental illness, according to the social-role model is <u>not</u> an inherent feature that characterizes the mentally ill; rather, it is a <u>social</u> <u>definition</u> bestowed by members of society upon individuals exhibiting certain types of behaviour.

Seen from this perspective, the career of the mental patient begins when societal members <u>react</u>—express discontent and/or take action against behaviour exhibited by the actor and subsequently define him/her as 'mentally ill.' Specifically, individuals are adjudged to be mentally ill when they exhibit a special type of culturally-inappropriate behaviour—behaviour to which Scheff (1966:33) refers as 'residual rule violations:'

The culture of the group provides a vocabulary of terms for categorizing many norm violations: crime, perversion, drunkenness and bad manners... Each of these terms is derived from the type of behaviour involved. After exhausting these categories, however, there is always a residue of the most diverse kinds of violations for which the culture applies no explicit label ... Although there is great cultural variation in what is defined as decent or real, each culture tends to reify its definition of decency or reality, and so provides no way of handling violations of its expectations in these areas. The typical norm governing decency or reality, therefore, literally 'goes without saying and its violation is unthinkable for most of its members. For the convenience of the society in construing those instances of unnamable rule-breaking which are called to its attention, these violations may be lumped together into a residual category: witchcraft, spirit possession, or, in our own society, mental illness.

Mental illness then, according to this perspective is conceived as a residual category of deviant behaviour-violations for which society provides

no explicit label.

In my study, the majority of the subjects began their careers as mental patients when members of society reacted to such residual rule violations as: (a) throwing dishes at a family member, (b) breaking a window in the house, (c) repeated urinating and defecating oneself, (d) refusal to maintain habits of cleanliness, (e) providing unacceptable responses to others in the context of social interaction, etc.

A large proportion of residual rule violations frequently go undetected—the behaviour remains unrecognized, or if it is acknowledged, it is rationalized or 'normalized' by the social audience. While various residual rule violations are ignored or rationalized away, in some instances, individuals exhibiting similar violations are reacted to by societal members and are labelled as mentally ill. Goffman (1961:134), taking as problematic the application of the label 'mentally ill' states:

For every offense that leads to an effective complaint, there are many psychiatrically similar ones that never do. No action is taken which leads to other extrusory outcomes; or ineffective action is taken leading to a mere pacifying or putting off of the person who complains.

In other words, defining an individual as mentally ill is not an automatic process. Whether a person is so labelled is contingent upon several factors, including the tolerance level of the community, the severity of the rule violation, and/or the social staus of the rule breaker relative to the audience reacting to his behaviour (Scheff, 1975:10).

A number of studies focussing on the tolerance limits of the family

have documented the capacity of family members to minimize, overlook, and rationalize evidence of mental illness. It is only in the event of a shift in the tolerance of the family members—when the individual's behaviour becomes unmanageable, when a crisis situation develops, that others define him/her as being in need of psychiatric help. The data from this study suggest that family members adopted similar patterns of accommodation—suspicions of mental illness were rationalized or normalized. Such accommodative mechanisms collapsed when family members were confronted with an unmanageable emergency. As the mother of a young woman, hospitalized for exhibiting 'bizarre' behaviour recalls:

My daughter wasn't quite 'right' for a long time. But it took me a long time to realize that. You see, at first I refused to face that anything was wrong with her. Her grades started getting worse when she entered high school. But I thought that she was just not trying hard enough. Then she started to get worse—I mean by being violent. Initially, I thought that she was just letting off steam—I didn't realize that she was 'sick.' But gradually I came to the conclusion that she was sick when one day she got so mad at me that she went 'beserk' and started throwing things. I was so afraid that I called the police. (Interview #47, January 28, 1981:1).

Not only do accommodative patterns break down in crisis situations, but also result when the individual repeatedly exhibits culturally—in—appropriate responses in the context of social interaction with others.

As Mechanic (1962:68) states:

Mental illness and other forms of deviancy become visible when persons in the participant's group recognize his inablity and reluctance to make proper responses in his network of interpersonal relations.

Similar to Laing and Esterson (1964), in my study I found that individuals were adjudged by significant others to be mentally ill as a consequence of their inability to accurately respond to interpersonal demands of others in the situation. As one family member recalls:

I gradually realized that something was wrong. When I would ask Joe a question, he would answer in a strange way. I mean, if I asked him a question about 'apples' he would answer and tell me about 'oranges.' Our conversations just didn't click! That's when I realized that he must be sick (Interview #56, April 2, 1981:1).

Similarly, a friend of an individual hospitalized in a mental facility states:

Before she was hospitalized, it was just like she was on another wavelength. I would be talking about a certain thing and she would answer me with a statement that was completely alien to the conversation. Sometimes she didn't answer at all. At first, I just shrugged it off, but when it persisted, I realized that she must be mentally ill and desperately in need of psychiatric help (Interview #58, April 13, 1981:4).

A third individual, recalling his wife's behaviour prior to hospitalization states:

My wife started acting 'funny' a few years ago. I mean, she went into this depression and wouldn't talk to anyone. No matter what anyone said, she just wouldn't answer. Then, after a while when she did answer, she would say things that didn't make any sense whatsoever. That's when I decided to get her psychiatric help (Interview #57, April 8, 1981:2).

In short then, the career of the mental patient begins when societal members notice that the behaviour exhibited by the person is strange or peculiar. Although the social audience may initially deny or minimize the peculiarity, if the individual repeatedly fails to respond to the interpersonal demands of the other in the situation, or if his behaviour becomes unmanageable, societal members move toward defining him as mentally ill.

A number of agents and agencies participate in the definitional process and subsequent hospitalization of the individual. As Goffman (1961: 135) notes, these "circuit of agents—and agencies—participate faithfully in his passage from civilian to patient status." Essentially, in my study, four agents reacted to residual rule violations exhibited by individuals, thereby activating commitment proceedings into the mental hospital:

- (1) Self as agent;
- (2) The legal system as agent;
- (3) Family and Friends as agents;
- (4) Psychiatric Officials in a general hospital as agents.

Self as Agent

One possible pathway leading to psychiatric hospitalization is through self-referral. A small number, upon self-examination find themselves to be acting in strange and incomprehensible manners. Such behaviour often frightens the person and leads him/her to believe that he/she is 'going nuts' or 'losing his mind.' This threatening view of self as being potentially mentally ill impels the individual to admit him/herself willingly to a psychiatric facility. One patient, discussing his experiences as a pre-patient states:

I kept hearing voices before I came to the hospital. I didn't know what was happening to me. It was so frightening. I thought that I was losing my mind—going bananas. That's when I decided to admit myself to the hospital and get help (Interview #19, January 2, 1980:5).

Similarly, another patient states:

It all started about four years ago—things happened to the family—my mother died; my two uncles died; my aunt died. I was very close to my aunt. My son and I used to live with my aunt in her house. I used to look after her. I saw that she was gradually going down hill and I knew that she was going to die. It was very hard on me. When she did die, I became very depressed...It was really depressing and my nerves got bad. To make things worse, my son decided that he would like to go out on his own. It was such a big shock and it depressed me further. I didn't want to eat or take care of myself. I knew that something was wrong with me. I wasn't 'normal.' That's when I decided to get help for my nervous breakdown and came to the hospital (Interview #53, March 13, 1981:3).

Concommitant with the self-realization that he/she is in need of psychiatric help, the pre-patient is also faced with the sometimes arduous and anxiety-provoking task of attempting to conceal from others his/her newly discovered discreditable image of self. The individual struggles to maintain his/her role as a 'normal' person in the context of social interaction with others while constantly fearful that he/she will be 'found out.' When the social situation becomes too stressful for the pre-patient to manage, he/she often feels relief by admitting himself/herself into the mental institution. As Goffman (1961:132-133) notes, "instead of being himself a questionable person trying to maintain a role as a full one, he can become an officially questioned person known to himself to be not so questionable as that." A person, describing his stressful experiences prior to hospitalization states:

THEMSTER TO

In my heart I knew that there was something mentally wrong with me, but I tried to hide it. I tried to hide it from myself but I eventually realized—I faced the fact that I needed help...I couldn't concentrate—I felt so mixed up—I thought that everyone was against me. And these feelings got progressively worse as time went on. As well as trying to hide my illness from myself, I also tried to hide it from my friends and other employees at work. Sometimes, it was quite difficult to pretend everything was ok. I kept wondering if they thought I was 'going mental'—whether they knew or not. I tried to hide it for a time, but that created more pressures on me until I couldn't take it anymore...It was a relief to go to the hospital. I didn't have to hide it anymore. Now I could get some help for my sickness (Interview #47, January 27, 1981:5).

Another person, recalling her experiences as a pre-patient states:

Things started building up on me. At first, I thought that I was just overtired—that I was working myself too hard. But

it got worse—it was harder and harder to make decisions. The work started piling up. I couldn't cope. I finally realized that something must be wrong with me. I was 'sick.' I wondered if my friends could tell that I was sick. I tried to hide it from them for awhile, but then it got worse and I had to get psychiatric help (Interview #56, April 10, 1981:2).

A third individual, discussing his stressful experiences prior to hospitalization states:

Life was hell before I was admitted to the hospital. I was so depressed. I couldn't eat. I felt like everyone was against me. I used to cry a lot—sometimes I couldn't stop. And this scared me to death. I didn't know what was happening to me. I was becoming mentally sick little by little. During this time it was difficult to carry on a normal life, when you know that something's wrong with you. It got to the point where I couldn't carry on anymore and then I went and got myself signed into the hospital (Interview #56, April 10, 1981:4).

For these individuals who were either frightened by the strange behaviour they were exhibiting, or were no longer able to manage the stressful experiences of attempting to conceal certain discreditable aspects of their selves from others, admission to the psychiatric hospital is seen in a positive light.

Despite the fact that some pre-patients admit themselves willingly to the mental hospital for treatment, the majority of individuals are
admitted by various agents and agencies against their will. For these
individuals admitted involuntarily, their social experiences as prepatients are generally seen in a negative manner (Goffman, 1961). Since
the majority of persons in my study were admitted in this latter manner, 10
this discussion will now focus on the various agents that participate in

committing the individual involuntarily and the individual's reaction to such action.

The Legal System as Agent

A number of individuals mark the beginning of their careers as mental patients when legal authorities interpret their behaviour as evidence of mental illness and consequently admit them to a psychiatric facility. Specifically, legal authorities participate in the individual's passage from civilian to patient status in two ways:

- (1) A person exhibiting 'bizarre' behaviour on the outside is apprehended by the police and admitted directly to a psychiatric hospital for treatment.
- (2) An individual committing a legal offense is apprehended by the police and imprisoned. At a subsequent hearing, the court decides to send the person to a mental hospital for psychiatric evaluation.

In terms of the former path, some individuals, although not breaking the law are apprehended by the police. That is, according to a provision in the legal mandate, police officers are given the authority to apprehend and hospitalize individuals acting in a strange manner—persons committing residual rule violations. From the standpoint of the police officer, psychiatric hospitalization of the individual is conceived

as a necessary step in order to ensure that he does not injure himself/
herself or others. However, from the point of view of the person himself/
herself, his/her apprehension and involuntary hospitalization are events
which he/she sees in a negative light. Specifically, most individuals
apprehended and hospitalized by the police cannot comprehend why such
action was taken against them. For these persons, hospitalization is unjust—they adamantly deny that anything is wrong with them. An individual,
expressing the moral outrage of his social situation as a pre-patient
states:

I just can't understand it. One minute I was out there minding my own business—not bothering anyone, then the next minute, I find myself being brought to the hospital by the cops. I never did anything. I'm not sick. There is no way that they should have done that to me! I demand to be let free! (Interview # 17, December 20, 1979: 15).

Similarly, another person states:

I was on a highway and the police just picked me up and brought me here. I never asked them why they picked me up. I was too scared. I can't understand why they brought me to this place. To this day, I still don't know why! (Interview #54, March 18, 1981:9-10).

A third individual, recalling his experiences with the police as a pre-patient states:

I wasn't doing nothing wrong when they (the police) picked me up and brought me to the hospital. I wasn't doing nothing I tell you. They had no reason. I wasn't breaking any laws.

I was just going about my own business, but they said that they thought I was confused. They were wrong. They had no right to imprison me like that (Interview #22, January 17, 1980:8).

In short, individuals I interviewed who were admitted to the psychiatric hospital by the police as a result of committing some residual rule violation felt that such action was totally unjust and demanded to be released immediately. These individuals felt that they were the victims of a gross misjudgment on the part of the police.

In contrast to individuals admitted to the psychiatric facility by the police as a consequence of exhibiting strange or incomprehensible behaviour, others are admitted by the legal authorities as a result of committing a legal offense. Persons breaking such laws as arson, burglary, murder, rape, etc., are apprehended by the police and taken into custody. During a subsequent court hearing, the details of the case are discussed; if the motive behind the crime appears to be irrational in the eyes of the judge and/or the defense attorney, the individual's sanity may be called into question. Hence, the court may decide to send the offender to a mental hospital for a psychiatric assessment in order to ascertain whether he/she is 'psychiatrically fit' to stand trial.

The legal offender, wearing handcuffs is transported to the hospital by the police at which time he is handed over to the hospital staff and becomes the property of the psychiatric institution.

For such individuals, their moral experiences as pre-patients are generally conceived of in a negative manner. That is, most persons admitted to the hospital in this manner express indignation regarding the

objectification of their selves by both legal authorities and hospital staff. One patient, recalling his social situation as a prepatient states:

They treated mellike an object, that's all! Right from the moment I was caught by the police, in the courtroom, right up until the time they handed me over to the ward staff of the hospital. No one talked to me like I was a person—they just talked about me as if I wasn't there. They treated me just like a piece of cargo that needed to be moved from one place to another (Interview #7, October 25, 1979:13).

Similarly, another patient recalls:

When I was brought up from the jail by the cops, everyone handled me like some piece of trash—something without feelings. They talked about me as if I wasn't capable of understanding what they were saying about me. It made me so mad (Interview # 9, October 31, 1979:15).

In contrast to persons admitted to the mental hospital by the police as a result of committing a residual rule violation—individuals who generally express moral outrage for such action taken against them, others committed via the court reacted to their incarcerations in a somewhat acquiescent manner. Although such pre-patients may not be in accordance with the adjudgment of their selves as potentially mentally ill, they are informed of the rationale behind the judgment, the course of the action that will be taken, and the length of time they will be required to remain in the hospital—knowledge which makes their social situations more tolerable. Conceiving of his situation as 'doing time', one patient states:

They (the court officials) don't know what they're talking about—trying to say that I'm crazy. I know that there's nothing wrong with me. Oh, I don't care what they think anyways. I'll just do my time. I only have to be here for thirty days for some kind of assessment, then I'll be out (Interview #9, October 31, 1979:17).

Another patient states:

There's nothing wrong with me. I'll be out of here pretty soon. I just have to be on my best behaviour and do my time and I'll soon be out. You just have to learn to bear it (Interview #9, October 31, 1979:18).

While some pre-patients processed by the court deny that they are mentally ill, others however, prior to being sent for psychiatric examination, have come to see themselves as mentally unbalanced. For these individuals, admission (even involuntary admission) to a mental facility is often seen as a welcome relief to their problems. As one person explains:

I went to jail for break and entry. But you know, I never knew that I committed the crime until after it happened, and the police picked me up and told me what I did. I knew I needed psychiatric help...I didn't object to coming here (Interview #7, October 25, 1979:7).

For persons, who prior to hospitalization were already questioning their sanity, their entire experiences as pre-patients are generally conceived as a positive step toward the alleviation of their problems.

Family and Friends as Agents

A third group of agents who frequently play a role in the individual's passage from person to patient are his/her family and friends a group of persons with whom the pre-patient most frequently interacts, and in whom he/she places complete trust.

During the pre-patient phase of his/her moral career, the individual gradually discovers that although he/she has placed reliance on family and friends to be supportive in times of trouble, these are often the very individuals who are the first to doubt the pre-patient's sanity and take action against him/her. When the individual repeatedly fails to correctly respond to the interpersonal demands of the other in the situation, the others react to such behaviour and adjudge the individual to be mentally ill. Subsequent to interpreting the pre-patient's behaviour as evidence of mental illness, friends or family of the person, frequently enlist the help of the family physician, mental health official, justice of the peace, or other official third parties in committing the individual to the psychiatric facility—action which the pre-patient views as trait—orous and conspiratorial in nature.

In some cases, the process of commitment begins when family members or friends attempt to persuade the individual into visiting a family physician or mental health counsellor for what they are repeatedly told is for 'consultation purposes only.' As one patient recalls:

My mother told me that nothing was going to happen to me. No one was going to take me anywhere. She said that the doctor just wanted to talk with me—we were

just going to talk—that's all! (Interview #20, January 5, 1980:8).

Similarly, another individual states:

My cousin came over and tried to talk me into seeing the doctor—just to 'talk things over.' She said that he would just talk about how I was doing and that would be the end of it (Interview #58, April 15, 1981:5).

If the individual refuses to oblige, he is often threatened by his significant others with desertion, legal action, or disfellowship. One person recalling his social experiences as a pre-patient states:

My wife told me if I wouldn't go to see the doctor then she was going to leave me. She said that she'd take the kids and get as far away from me as possible...(Interview #59, April 28, 1981:2).

Another patient recalls:

I was told that if I refused to keep the appointment with the family doctor, I would have to get out of my father's house and never show my face around there again (Interview #16, December 19, 1979:11).

A third individual states:

My husband threatened to take my two children and leave me. He said that either I go or he'd beat me up. So what choice did I have? (Interview #57, April 12, 1981:3).

Generally, a family member or friend, prior to attempting to persuade the pre-patient to 'consult' a professional, will have already set up the appointment. Moreover, he/she will have 'filled in' the official third party regarding the case history of the adjudged individual. Such action, according to Goffman (1961:137), "tends to establish the next-of-relation as the responsible person to whom pertinent findings can be divulged, while effectively establishing the other as the patient."

Prior to arriving for the appointment, friends or family members engage in polite small talk with the pre-patient -- no mention is made of the pre-patient's impending fate. The pre-patient's significant others emphasize repeatedly that they will not let anything happen to him. However, during the course of the interview, the pre-patient gradually discovers that a coalition has been formed against him in an attempt to commit him to a psychiatric institution. That is, upon arrival for the appointment, the pre-patient finds out that he has been accorded the role of 'patient' by the professional and his significant others. Moreover, from the nature of the conversation and the types of questions asked during the consultation, the pre-patient realizes that he has been 'informed on' or betrayed. The pre-patient finds out that his significant others-those individuals who prior to the appointment stressed that they were 'on his side' and would not let anything happen to him, turn against him and take the side of the professional who emphasizes the need for hospitalization. In short, the pre-patient sadly finds out that his friends or relations, the ones on whom he thought he could depend, have betrayed him. A patient, discussing his feelings of embitterment and betrayal toward his family states:

All the time, they kept telling me that they weren't going to let anything happen to me. That's what they told me. They never mentioned one word about putting me in this hospital. Then we got to Dr. ______'s office and I saw what was really going on. My whole family and the doctor were plotting against me behind my back. My mother told the doctor all kinds of things about me even before I showed up. They were trying to get me committed to this place. They're all traitors and I hate them! (Interview #21, January 11, 1980:5).

Similarly, a second patient recalls:

My wife assured me that she would not commit me to this place. Even my own doctor promised not to let that happen. But boy was I ever a fool! Everything was set up by my wife and the doctor behind my back. It went so quickly that I was committed before I even realized what hit me. One minute they were my loyal friends, and the next minute they turned against me and put me in this hospital. I just can't forgive them for that! (Interview #58, April 15, 1981:9).

Psychiatric Officials of a General Hospital as Agents

A fourth and final route leading to hospitalization in a psychiatric facility is via psychiatric officials of a general hospital.

Specifically, a number of individuals exhibiting symptoms of mental illness are reacted against by various social audiences and are admitted to the psychiatric ward of a general hospital for treatment.

If a person fails to respond to treatment, or is diagnosed as being in need of extended psychiatric care, psychiatric officials subsequently transfer him/her to a mental hospital. Some individuals respond to such action taken against their selves in a positive manner:

When I didn't get any better after the shock treatments, they transferred me to Meadowdale. I didn't really mind going there. I thought that I might get treated better than in the ordinary hospital. I thought that they could help me to get better (Interview #53, March 13, 1981:4).

Similarly, another individual states:

At first	in		(gene:	ral hos	pital)	they to	ried :	all
kinds of	drugs	on me	but it	didn*t	make m	e feel	much	better.
Dr.	the	ught t	hat it	would	be best	for m	e to	go to
the psyc	hiatric	hospi	tal	.I agre	ed beca	use I	wante	d to
feel bet								
5),					•	•	•	

Others, however respond to the action of transfer in a negative manner.

These individuals expressed fear and anxiety regarding their social fates:

I was scared, I mean, really scared when I found out that they were transferring me to this place. When I was on the psychiatric ward in the general hospital, I thought that they would just keep me there for a few days and then let me out, but I never thought I'd end up here. I didn't know how long they planned to keep me at this place and it scared me (Interview #45, January 21, 1981:5).

A second individual, discussing his feelings toward the action of transfer to a psychiatric institution states:

I was frightened to death when they told me where I was going to be sent. I didn't want to go there. I didn't know what to expect—I mean, what kind of patients were in there, what they were going to do to me. It really made me nervous (Interview #57, April 12, 1981:7).

While admission to a psychiatric ward of a general hospital against the pre-patient's will brings a loss of freedom, the action of transfer to the psychiatric facility proper represents yet a further loss:

It was bad enough in Oakridge (the general hospital). When they admitted me in there, they took away a lot of my things—they kept watching me all the time. I couldn't do anything freely on my own. But being sent here to Meadowdale was much worse. It was like becoming a lower being...You see, you lose most, if not all your freedom that you used to have when you come in. It's like going down a flight of stairs—you were once at the top, but when you get admitted, you quickly fall to the bottom (Interview #19, January 2, 1980:8).

Another person states:

It was worse once I got to Meadowdale. They took all my personals away...I couldn't even have my wallet and my money. You lose all your freedom. They lock you up like animals. On the outside you get to do pretty much as you want, but here you are locked in and controlled (Interview #19, January 2, 1980:6).

In general, most patients conceive of the pre-patient experience of transfer with animosity. While involuntary admission to the psychiatric ward of a general hospital is viewed by these individuals as a gross misjudgment, they conceive of the action of transfer as yet a further and more serious error on the part of others—a mistake that is unforgiveable. A patient, expressing contempt toward a psychiatric official's decision to transfer him to a mental hospital states:

I hate that doctor for sending me in here. How could he do this to me? He had no right. I feel like killing him.

I hated it when they admitted me to the psych. ward in the ordinary hospital—there was nothing wrong with me. They made a big mistake. But when that doctor said that I needed 'therapy' so he was sending me to Meadowdale, I nearly hit the roof. They were making a worse mistake—there was nothing wrong with me and they were sending me to the nut house. (Interview # 60, April 30, 1981:2).

THE TOTAL STATE

In summary, this chapter has dealt with the pre-patient phase in the career of the mental patient with specific emphasis on the social conditions upon which an individual is defined as mentally ill—the reaction of the social audience to specific types of rule violations being the crucial definitional element. Secondly, this chapter focussed on the various social agents which participated in the definitional process and subsequent hospitalization of the adjudged individual, and the reaction of the person to such action taken against him/her self.

The next chapter will deal with the second stage in the career of the mental patient—the in-patient phase. Specifically, it will deal with the various admission procedures to which a person is subjected upon institutionalization, and how such procedures and regulations function to strip the individual of his/her present identity. Secondly, this chapter will discuss the hospital's privilege system as the chief context within which the patient is forced to adopt a redefinition of self as 'mentally ill.'

FOOTNOTES FOR CHAPTER FIVE

- The reader should note that the pre-patient phase was not a central focus of this study; data collection focussed on the in-patient and ex-patient phases in the mental patient's career. This chapter then, based on my somewhat limited analysis of the pre-patient phase, presents a brief introduction to some of the social conditions which may lead to hospitalization, and the response of the individual to such action.
- 2 See Kitsuse (1965) for a discussion of the societal reaction component to the study of deviant behaviour in general.
- This stage of rationalization is referred to as 'denial' by Cumming and Cumming (1957:92-103). See also, Clausen and Yarrow (1955:25-33) and Hollinghead and Redlich (1958:170-176).
- For a discussion of some of the career contingencies that may affect hospitalization in a mental facility, see: Lemert (1946: 370-378); and Meyers and Schaffer (1954:307-310).
- See, for example, Sampson et al. (1962); Yarrow et al. (1955); Schwartz (1957); Hollingshead and Redlich (1958); Laing and Esterson (1964).
- Smith et al. (1963:228-233) in their research on the tolerance limits of the family found that accommodative patterns broke down when a 'last straw' type incident occurred.
- 7 See also Becker (1962:494-501); and Gough (1948:359-366) for a somewhat similar discussion regarding the definitional process of an individual as mentally ill.

- For the purposes of this thesis, an analytical separation has been made between the four routes leading to hospitalization. However, in reality, there exists an interplay between the various agents' reactions to the residual rule violation and the individual—an interplay which is beyond the scope of this thesis.
- 9 In this case, five subjects (or 25% of the sample).
- In this case, fifteen subjects (or 75% of the sample).
- See also, Bittner (1967:278-292) for a discussion of police discretion in the apprehension of the mentally ill.
- Psychiatric assessments ranged from 30-60 days during which time forensic staff evaluated patients by means of observation, interviews and psychological testing.
- This theme continues, and is intensified during the in-patient phase of the person's career as a mental patient.
- In my study, such individuals were in a minority, specifically only 5% of the sample.
- The individual discovers that his family members are actually serving as 'double agents'—on the one hand they pretend to be loyal and supportive, and yet, such persons turn against the individual and realign themselves with the professional.
- While some pre-patients are admitted directly to the psychiatric hospital for treatment, others are sent first to a psychiatric ward of a general hospital. The reason for such action is two-fold: (1) in some cases, individuals are living in a region

that does not contain a mental hospital; (2) with the trend toward deinstitutionalization of the mental health facilities, individuals are admitted to a psychiatric ward in a general hospital for short-term treatment and then are released back into the community. Only when such treatment fails are patients transferred for further treatment to a psychiatric institution.

CHAPTER SIX

THE IN-PATIENT PHASE IN THE CAREER OF THE MENTAL PATIENT

The Process of Admission: The Context Within Which Self is Mortified

As Goffman (1961:14) emphasizes in his study of total instil tutions, the individual generally enters the institution with a positive conception of self; however upon admission and thereafter he is
stripped of many of his accustomed possessions and symbols of identity
including clothing. The individual is subjected to a set of mortifying
experiences which function to strip him/her of his/her present identity.

The recruit comes to the establishment with a conception of himself made possible by certain stable social arrangements in his home world. Upon entrance, he is immediately stripped of the support provided by these arrangements. In the accurate language of some of our oldest institutions, he begins a series of abasements, degradations, humiliations and profanations of self. His self is systematically, or often unintentionally mortified. He begins some radical shifts in his moral career, a career composed of the progressive changes that occur in the beliefs that he has concerning himself and others.

In my study, I found that the patients were subjected to certain processes of mortification. Specifically, upon entrance the patient is

subjected to certain mortifying experiences or stripping processes that are elaborated in the form of 'admission procedures.' On entering the hospital, the individual is subjected to a systematic set of procedures: he is signed in, assigned to a specific nurse and psychiatrist, asked questions about his life history, certain possessions are taken from him and stored, he is given a physical examination, a haricut, instructed in the rules of the ward, assigned to a room, and is issued institutional clothing if he has none of his own. In other words, during admission procedures the individual is subjected to a variety of routine operations—in essence, he is coded into an object that can be easily processed by administrative mechanisms of the institution. While one might agree that these admission procedures are necessary for the efficient organization and operation of the institution, from the patients' point of view these procedures are seen in a different light. A new admission states:

When I came in here I felt so mixed up. I was scared. female police brought me up here; I had chains on my arms and my feet. They brought me up to the front door and handed me over to the hospital woman. I felt like a piece of garbage you know, not like a human being-they were treating me like a fucking object, that's all. So they handed me over to her and she took me to the ward. We entered the ward and I was signed in. Then they took me down to the sunporch and left me there alone. After about twenty minutes, they came in and got me, asked me a few questions and sent me back there alone again. This time they took me to a room where there was a doctor and a nurse. They were going to give me a physical examination. They told me to strip. I told them that I wasn't going to take my clothes off in front of a woman; she said that it was alright because she was a nurse. I told her to leave but she wouldn't. So I had to stand there nude, right in front of her-she just gawked at me. I felt like a piece of shit I tell you...the doctor examined me and I do mean all over...I felt so damned degraded...they do that a lot to us here (Interview #7, October 25, 1979:6).

Another patient, recounting his views on the admission procedures says:

When you come in, they take away your possessions, but they also do something that's much worse-they take away your self-respect. You come in here thinking that you are basically a good, decent human being, but they somehow manage to degrade you -- make you feel like an inferior, a nothing, a sub-human.) Let me tell you how it works: you come in here thinking you are a good person...But from the first minute you come through the front door, things start to happen-they take everything away from you--your clothes, money. They strip you and examine you. Then they put you in pajamas... you feel degraded)...You feel like you are being pushed down a flight of stairs, each step of the way makes you feel worse. Some patients accept they are nothings, no good; others try to fight it for awhile, but in the end the staff finally get you where they want you -- to admit that you are a nothing (Interview #28, February 6, 1980:11).

Upon admission then, the individual is subjected to a set of procedures which function to curtail or alter the self. The act of physical examination serves to strip the individual in two ways: not only does this procedure force the person to be physically stripped of his clothing, but also symbolically serves to strip him of his self-identity through degradation and humiliation. Moreover, during admission procedures and thereafter, the patient tends to be objectified by the staff. Whereas on the outside the individual is treated as a person and is given respect, upon hospitalization he is accorded an inferior status—he is reduced to the level of an object. In short, he is often treated as a non-person.

The individual undergoes mortification of self in other respects as well. Upon admisssion the patient is stripped of many of his material possessions such as his wallet, money, identification, jewelry and other

valuables—items that are locked away in storage. Since in Western society material possessions are a significant part of an individual's conception of self, to be stripped of such items represents a major attack on the self. While the institution may rationalize this procedure in terms of protection of the patient's property, the patients however, tend to conceive of this situation in a different manner:

THE REAL PROPERTY.

Don: You know what bugs me about this place?

Jane: What:

Don: They won't give me any of my 'personals'--you know, like my wallet, ID, and my razor and that kind of stuff...They have it but they just won't give it to me.

Jane: They did that to me, too. They took away all my jewelry and won't give it back to me. They want to see me beg for it and they still won't give it back.

Don: They make you feel like a piece of garbage—a nothing.
(When they take all our personals away, they take away
who you are...)(Interview #8, October 28, 1979:13).

One set of an individual's personal possessions is related to self-conceptions in a special manner. On the outside, an individual possesses some control over the personal front which he presents to others. That is, he employs a variety of symbols and tools to create a desired image of self. He employs what Goffman (1961:21) refers to as an 'identity kit.' However, upon hospitalization the individual is stripped of his usual appearance as well as the equipment used to create and maintain it. In essence, he suffers a personal defacement. A new admission, dis-

cussing how his personal appearance was altered shortly after admission states:

The barber came on the ward last week and dragged me over for a haircut. I didn't want to get it cut off but they (the staff) made me...The barber just plunked me in the chair—didn't even talk to me and just cut off my hair....Shit, I felt awful you know...they were changing so much of me...Where was the old me? (Interview #7, October 25, 1979:17).

Another patient, complaining about the loss of some items of her identity kit says:

I hate it in here. They won't give me my razor so I can shave my legs. They even took away my manicure kit. I look like a mess now. I never looked like this before you know...If I want to shave my legs, I have to ask for the 'safety razor' at the nursing station (Interview #8, October 28, 1979:12).

The loss of certain material possessions and one's identity kit prevent the individual from presenting his usual image of self and represents another instance of how self is altered upon hospitalization.

After admission self-images and identities are assaulted in yet another way—specifically by means of a forced deference pattern. Upon hospitalization the patient soon learns that he must act politely toward the staff or else he will receive negative sanctioning. This very action of required deference functions to attack one's self-conceptions. The self is also mortified by virtue of the fact that the patient is forced to request permission (and is sometimes forced to beg) for small items and activities, such as cigarettes, going for a bath, spending one's own

money, going to the canteen, washing one's clothes—items and activities which the person was able to exercise on his own on the outside. This obligation not only places the patient in an inferior or submissive role, but also leaves him wide open for interruptions from the staff. On the outside if an individual desires something, his requests are readily granted; in contrast, upon institutionalization, his requests may often be denied, questioned or even ignored:

Last night, I waited and waited at the nursing station. I wanted someone to unlock the door so I could take a bath. The staff all saw me standing there but they just ignored me. After about ten minutes I yelled out to them and told them what I wanted but they just put me off. They told me to come back later and ask again (Interview #20, January 5, 1980:6).

Another patient describes the condescending treatment accorded the patients by the staff:

They treat us like children in here... Now if you're a good boy and ask politely, you can have one cigarette each hour' (mimicking the voice of a staff). Shit, if you smoke, how much is one cigarette an hour? But like I said, they treat us like kids; they talk down to us. It makes us feel 'unhuman' if that is a word (Interview #7, October 5, 1979:9).

In sum, these demands for deference and the corresponding implications of inferiority present the individual with a self-image that is incompatible with his prior conception of self.

The self is also mortified in other respects. In Western society

there is no more important claim to status, prestige and identity than a person's job (Haas and Shaffir, 1978:33). As Everett Hughes (1958:314) states:

A man's work is one of the more important parts of his identity, of his self; indeed of his fate in the one life he has to live, for there is something almost as irrevocable about the choice of occupation as there is about the choice of mate.

A person's occupation is an intricate part of his self-identity. However, upon hospitalization the patient is separated from his occupation in the outside world; in its place, he is often given menial work to undertake for which he is paid a small sum of money. So for example, as part of the patient's 'therapy,' he may be obliged to help in the hospital kitchen clearing tables or scraping plates. For this labour he may be given a few cents or a cigarette. At the Industrial Therapy Workshop of the institution, patients are required to undertake light factory work such as filling packages and cartons with various goods. Others work at the hospital laundry where they wash and iron much of the institutional clothing. For this type of work, the patients are paid menial wages. Being forced to work at such tasks for small sums of money is degrading for the individual and represents yet another instance of how self is assaulted upon hospitalization:

Working at IT (Industrial Therapy) is awful. You start working for 20¢ an hour and you can work your way up to 70¢ an hour and that's doing heavy manual labour, like lifting crates. See that old lady over there? She makes

40¢ an hour stuffing kotex in a box. How degrading!...The money you make in here isn't enough to keep you in cigarettes. It's just token wages, that's all. I used to make over \$200.00 a week on the outside, but in here I'm lucky if I make \$10.00 a week...It's like a slap in the face, you know (Intervew # 7, October 25, 1979:14).

I TOTAL TOTAL

Two patients, comparing hospital wages to wages in another institution state:

Joe: You know, it's better in prison than in here. In prison the pay is better. Some inmates make 5 or 6 bucks an hour for doing work.

Tom: Ya, when I was in jail I got paid \$1.90 an hour for working in the kitchen. But here in the mental hospital all they pay is 20¢ an hour. Ya, convicts are treated better than mental patients—even though we have to do shitty jobs in jail too, at least we're paid a bit better than in this joint! (Interview #12. November 16, 1979:8).

It is further important to note that another type of mortification occurs upon admission. Essentially, a type of contaminative exposure coccurs upon entrance. While in the outside world an individual is able to segregate objects of self-feeling, such as his thoughts, actions and his physical being from certain contaminating items, in the hospital however, the boundaries of the self are violated (Goffman, 1961:23). Specifically, upon entering the hospital, a person's informational preserve relating to self is infringed upon. During entrance procedures information is collected regarding the patient's roles, statuses, and his past behaviours (including discreditable information) and is synthesized in the form of a case history of the patient. This file is made readily avail-

able to any staff member. The patient has virtually no control over who is allowed to learn certain discreditable facts about himself. In this sense, his territories of self are being violated. Because staff members have access to these case histories, sometimes in the course of group therapy sessions, the therapist forces the patient to reveal certain discreditable facts about himself in front of other group members. A patient recalling his experience in a group therapy session states:

I hate this place. They (the staff) know everything about us—especially all the bad things we've ever done. Every time I go to one of those sessions they keep bringing up my past. I know I've done a lot of wrong things in my life—I admit to that, but they keep wanting me to talk about things I did a long time ago, right in front of the other patients. It makes me feel ashamed when I have to tell bad things about myself and they hear it. I don't want everyone else in the group to know all the things I've done (Interview #25, January 26, 1980:11).

Another patient says:

When you come in here the staff learn all about you, especially all the bad things you don't want them to know. It's all written down in black and white in your case history record that's kept at the nursing station. Before I came here, I could hide some of these bad facts about me from other people, but in here that's impossible (Interview #12, November 16, 1979:6).

While the patient undergoes mortification of self by contaminative exposure of the kind discussed above, he also undergoes mortification of self through interpersonal contamination. As Goffman (1961:28)

emphasizes, "when the agency of contamination is another human being, the inmate is in addition contaminated by forced interpersonal contact and, in consequence, a forced social relationship." Many times during a patient's hospitalization his physical being and room may be searched. While the institution rationalizes these actions in terms of the protection of the patients, nevertheless, this very action of searching, along with the person who conducts the search, functions to violate the boundaries of the self. A patient expresses this view:

You know what bugs me about here? They search our rooms for 'junk' (drugs) all the time... There's no privacy at all... Like today me friend Jack came in to visit me and when he left they called me to the office while the others (the staff) went into my room and searched through all my stuff. I saw them do it. Man, we got no privacy at all! (Interview #7, October 25, 1979:7).

Self is also assaulted in one final respect—through the enforcement of strict rules and regulations. On the outside, the individual has some control over his world—he is a free, autonomous being. As such, he is free to make choices on his own as well as to act as he pleases within certain prescribed limits. The correctness of his actions is judged only at certain times. The individual is accorded certain civil rights as a Canadian citizen. In contrast however, upon institutionalization, the mental patient is subjected to a rigorous set of rules and regulations that serve to control every aspect of the patient's life. The civil rights he once enjoyed on the outside are taken away from him. He no longer has free choice nor is he able to act as he pleases. The

patient's actions are constantly being judged by the staff, and negative sanctions result if he violates an institutional norm. A patient aptly describes this situation:

Did you know that mental patients have no rights?...When you come in here, you lose all the rights you had on the outside. There is no free choice in here—choices are made for you. Like you have to eat at a certain time, wash at a certain time when they tell you, work when they tell you. In here they force you to think and act the way they want...They're always watching what you do and if you goof up, you're punished...(Interview #21, January 11, 1980:3).

Another patient states:

When a person is committed into the mental hospital he loses most of the rights he had on the outside as a citizen. For example, we aren't allowed to vote anymore, drive a car or any of those things—we lose all those rights we used to take for granted. With no rights left, we are left to abide by the rules of the hospital or else receive punishments. Taking away all these things is like taking away part of your identity—you lose who you once were. 6 (Interview #22, January 17, 1980:5).

In sum, the loss of certain rights combined with the vast body of rules and regulations to which the patient is subjected functions to threaten the individual's conception of self. No longer is he a self-deterministic entity; rather, the patient is reduced to a weak, helpless, dependent being.

In summary, I have attempted to illustrate how a mental institution, in its effort to resocialize and rehabilitate its charges, begins by in-

itially stripping the individual of his old identity, through various admission procedures. While the self is curtailed through these mortifying experiences, I would argue that it is largely within the context of the institution's privilege system that self is reconstituted.

It will be the purpose in the second part of this chapter to examine how self is reorganized in the framework of the privilege system. Specifically, I will undertake the following: (1) outline the major components of the privilege system; (2) its official purpose; and then (3) discuss its implications for self-images and identities.

The Privilege System: The Context Within Which Self is Reconstituted

Examination of the privilege system reveals that it is comprised of three basic components: formal norms, positive sanctions (rewards) and negative sanctions (punishments). In terms of the formal norms, it is evident that the mental institution possesses a rigorous and explicit set of rules and regulations detailing the main requirements of patient conduct. So for example, such rules prescribe at what time a patient gets up in the morning, what time he/she retires, when he/she is allowed to bathe, whether he/she is allowed to go to the canteen, attend social functions, etc.

A second major component of the privilege system may be termed positive sanctions or rewards. The mental hospital, in exchange for cooperation and compliance, offers the patient a small number of clearly-

defined privileges. As I pointed out earlier in this chapter, upon entrance the patient is stripped of many of his material possessions. Items and activities which the patient once took for granted on the outside, now become privileges upon hospitalization. Through the privilege system, certain possessions, items and activities are held up to the patient as possibilities which he can strive to regain. For example, when the patient becomes obedient in both thought and action, he might be given his identification and wallet back, allowed to go for a walk to the hospital canteen by himself, or even be allowed to venture into the city for a short time. In this sense, the individual feels that he is re-establishing some of the relationships with the outside world which were lost upon hospitalization.

A third and final component of the privilege system may be termed, negative sanctions or punishments. When a patient violates one of the institutional norms, he is punished. While corporal punishment is prohibited, punishments may take the form of being placed in the sideroom for a period of time, temporary loss of privileges, or even confinement to pajamas.

Two important points regarding the privilege system should be noted at this point. Firstly, discharge from the hospital is elaborated into the privilege system. As Denzin (1968:349-358) and Hollingshead and Redlich (1958) emphasize, certain patient behaviours and overall presentations of self serve to facilitate the acquisition of more privileges and ultimately, the patient's release, while other types of behaviour tend to lengthen his stay:

^{...}if the patient presents himself in such a way as to communicate his acceptance of the 'psychiatric line,' the therapist

will act toward him in a way which defines him as a good patient and one who will be easy to treat...It would be predicted that patients defined as holding initial favorable attitudes toward the 'psychiatric line' would...remain in the hospital a shorter length of time (Denzin, 1968:350).

Secondly, it is apparent that positive and negative sanctions of the privilege system become elaborated into a residential system. That is, certain places to sleep or interact become clearly-defined as places in which patients with <u>certain privileges</u> are allowed to frequent or abide. Basically, patients are moved back and forth from one spatial area to another within the ward as rewards or punishments for their behaviour and presentations of self. In this sense then, the system itself remains static while the patients are shifted back and forth within the structure. A patient explains how the privilege system operates:

When you first come in, you are often put in pajamas. You're not allowed to even wear clothes. You have no privileges at all....They put you on the back part of the ward; you can't go out to the other side of the ward—they have a locked door between the two sides of the ward. You're confined back here for a week or more (Interview #12, November 16, 1979:2).

Another patient recounts:

If you're good and behave like the staff want, you can work your way up and get more benefits....Like when you get semi-privileges you're allowed to go down to the other side of the ward into the music room and listen to the stereo. Also, you no longer have your room on the back part of the ward—the unprivileged side, but you get to move to a room on the front part of the ward with the more privileged people. Later you can work your way up to full-privileges which means that you

can go out on your own for walks or to the canteen (Interview #12, November 16, 1979:4).

At the official level, the privilege system can be seen to have two functions: (1) as a mechanism of social control; and (2) a therapeutic aim. That is, in terms of the former, the privilege system serves to make the patient obedient and cooperative in terms of the aims of the institution. The privilege system can be viewed as a mechanism of social control utilized to manage a large group of people. In terms of the latter function, the privilege system is also conceptualized by the institutional staff as having some therapeutic value. That is, this system functions to allow the patient to learn responsibility for his own actions as well as the actions of others. It serves to rehabilitate the patient so that he will be able to successfully return to the outside world. A staff member describing the purpose of the privilege system states:

The way I see the privilege system is this—it essentially has a therapeutic aim in that it provides the patient the opportunity to accept responsibility for himself and other patients...The privilege system also helps us control the patients...It provides the staff with information about a patient's behaviour that might otherwise be hidden, so we are able to judge him in this way...(Interview # 31, February 21, 1980:1).

Examination of the privilege system reveals that it is comprised of four levels. These are as follows:

Lowest Level

Level I

WARD PRIVILEGES—The patient is conto the back half of the ward, often being forced to remain in pajamas.

Level II GROUNDS-ACCOMPANIED—The patient has semi-privileges which means that he can move freely throughout the ward and is also permitted to go for a walk on the grounds if accompanied by a staff or a patient who possesses 'full-grounds.'

Level III FULL-GROUNDS-The patient has full privileges which means that he is allowed to go out of the hospital by himself, go to the canteen by himself, etc.

Highest Level IV OFF-GROUNDS--This level of the privilege system entitles the patient to leave the hospital grounds whenever he desires, providing that he returns within a specified time.

Essentially, when a patient enters the hospital, he is placed at Level I and is confined to the ward. After approximately one week's time, the patient can ask permission from the staff to attend a 'Privilege Meeting' which is comprised of two staff members and his fellow patients. At this meeting the patient asks permission to advance to the next level in the privilege system and be granted more privileges. The other patients then vote on whether the patient should be granted more privileges, and if a unanimous decision is reached, he is allowed to proceed to the next level. Although one might initially think that the privilege system operates in a democratic manner, closer examination reveals that this is not the case. Although the staff do not take part in the actual voting, they do however, possess the authority to veto any decision if they so

desire.

According to the ideology of the institution, the privilege system, the privilege system may be characterized as a resocializing and rehabilitating mechanism—a mechanism that helps the patient learn to 'accept responsibility' for his actions. As the patient gradually learns to be more responsible he moves from levels one through four and is eventually discharged back into society as a productive being. In contrast to the ideology of the institution however, when one attempts to make sociological sense of this system, it is seen in a different manner. Specifically, I would argue that the privilege system provides the chief context within which the patient is forced to adopt a redefinition of self as 'mentally ill' or 'deviant.'

From an interactionist perspective, one develops a self-identity on the basis of hos he perceives others are perceiving him. One's self-conceptions reflect the image he believes others have of him and is closely tied to the reactions imputed to other individuals. Self, is socially-constructed and socially-maintained. When an individual initially enters a mental institution he possesses a conception of self that was made possible by the stable arrangements of society. However, upon admission and thereafter the patient finds that it is virtually impossible to manage his old identity. The patient is separated from individuals and structures on the outside which functioned to validate his behaviour. This social separation, combined with the various admission procedures serve to destroy or alter the individual's conception of self. Once stripped to a liminal entity, the hospital offers the patient an alternative identity of self as 'mental patient'--a self-image that is in-

compatible with his prior conception of self. I would argue that although the patient may not initially agree with this redefinition of self as being mentally ill, nevertheless, through the structure of the privilege system along with the shared definition of himself as being mentally ill held by his significant others, the patient is ultimately forced to accept this redefinition. Blake and Moulton (1961:1-2) make the following statement regarding the processes of conformity, resistance to influence, and conversion to a new self-identity:

An individual requires a stable framework, including salient and firm reference points, in order to orient himself and to regulate his interactions with others. This framework consists of external and internal anchorages available to the individual whether he is aware of them or not. Within an acceptable framework he can resist giving or accepting information that is inconsistent with that framework or that requires him to relinquish it. In the absence of a stable framework he actively seeks to establish one through his own strivings by making use of significant and relevant information provided within the context of interaction. By controlling the amount and kind of information available for orientation, he can be led to embrace conforming attitudes which are entirely foreign to his earlier ways of thinking.

Further, Thomas Scheff (1966:57) in his discussion of role-playing writes:

Having an audience that acts toward the individual in a uniform way may lead the actor to play the expected role even if he is not particularly interested in doing so. The 'baby of the family' may come to find this role obnoxious, but the uniform pattern of cues and actions that confronts him in the family may lock in with his own vocabulary of responses so that it is inconvenient and difficult for him not to play the part expected of him. To

the degree that alternative roles are closed off, the prooffered role may come to be the only way the individual can cope with the situation.

In my study, I found that new patients initially respond to their hospitalization by denying the there is anything wrong with them. They often assert a story proving that they are not mentally ill, that someone or something else is to blame for their hospitalization, and that the hospital officials are therefore unjust in forcing this new label upon them. The following quotations from patients provide illustrations of this self-respecting tendency:

I've gone through a lot of pain in my life. It's all those people out there who put me in here you know... Everyone on the outside is against me. I'm not sick but they say I am. It's just not true. They forced me to come in here (Interview #18, December 27, 1979:5).

Another patient states:

You know, I'm not crazy like the rest of them in here. I just got fed up with the working conditions in my home town. The town's dying. There's no decent jobs anymore. I came to the hospital to rest up, that's all. It's sort of a retraining programme for me (Interview #9, October 31, 1979:7).

A third patient states:

I went AWOL when I was fighting in Vietnam and they're still looking for me. That's why I'm in this place right now. I'm

hiding from them. There's nothing wrong with me (Interview #20, January 5, 1980:8).

While most of the time the other patients openly accept the patient's apologia without question, sometimes however, the patients force each other to view themselves as mentally ill. The following conversations aptly illustrate this point:

Bob: There's nothing wrong with me. I don't know why I'm locked up in here.

Dick: There's something wrong with you.

Bob: What?

Dick: You're sick—you're a dangerous mental, Bob. That's why you're in here. If you don't believe that, you'll never get out (Interview #21, January 11, 1980:4).

Similarly, two patients engaged in conversation state:

Mary: You're crazy Joan, you're retarded. Face the fact!

Joan: I am not.

Mary: Yes you are and you'll be in here for the rest of your life. (Interview #20, January 5, 1980:10).

While fellow patients sometimes play a role in forcing the individual to accept this redefinition of self as mentally ill, I found that it is

the staff, who through the context of the privilege system's rewards and punishments play a significant role in forcing the individual to adopt this redefinition of self.

Scheff (1966:84) in his study discovered that patients who found evidence for mental illness in both their past and present actions were rewarded by the staff:

Labeled deviants may be rewarded for playing the stereotyped deviant role. Ordinarily patients who display 'insight' are rewarded by psychiatrists and other personnel. That is, patients who manage to find evidence of 'their illness' in their past and present behavior, confirming the medical and societal diagnoses, receive more benefits.

In my study I also found that patients, through the context of the institution's privilege system were often rewarded for playing the role of the insane person. As a patient learns to play the role of 'being crazy', he is rewarded by the staff members by being allowed to advance to a higher level in the privilege system. A patient states:

The only way to get out of this place is to 'act crazy' and do exactly what the staff want. If you don't give in to the staff, you'll stay in here forever. Like I told you before, the staff all think we are completely insane—they tell us we are. If you disagree with what they say and argue that you're not, then you don't get anywhere. They get mad at you and make it tough. But if you go along with them and 'act' mental, then you'll be able to go places and maybe one day will be let out. (Interview #28, February 6, 1980:11).

Similarly, another patient states:

The best thing I could do is to play along with whatever

the staff wants. If the staff think I'm crazy, I should act crazy and not make any trouble. That's the only way to get more privileges around here. That's the only way to ever get out of here.) If you don't, then you'll stay down at the bottom of the heap in here—you'll never move up in the privilege system or nothing (Interview #21, February 11, 1980:6).

The patient, in his desire to attain more privileges learns to play the role of 'mental patient' proffered by the staff.

Just as the patient is rewarded in the context of the privilege system for adopting this redefinition of self and its corresponding role, so too is he punished for refusing to do so:

If you deny you're mental and saying nothing is wrong with you, you get treated bad by the staff. They can make it tought. They say that you're misbehaving, so they keep you in pj's or put you in the sideroom (Interview #28, February 6, 1980:9).

Another patient states:

If you keep saying that you're not mentally insane, you'll get nowhere. The staff will think that you're really sick then. You keep saying nothing is wrong with you and the staff interpret that as defiance—they will say you are disobeying them and you'll get your privileges taken away. (Interview #22, January 17, 1980:9).

The patient finds himself in a difficult position. On the one hand, he does not want to accept this redefinition of self as *mentally ill.*

However, due to the shared definitions held by his significant others, coupled with the rewards and punishments accorded him in the context of the privilege system the patient comes to 'play the role' of mental patient.

I would argue that this situation has serious implications for the patient's identity. Although the symptoms displayed by the patient may begin as a conscious pretense, the constant reinforcement from the staff ultimately causes these behaviours to become involuntary and habitual. In effect, role internalization occurs. 11

Cain (1964:278-289), in his study on 'borderline children' provides evidence in support of this view. That is, he found that institutionalized children 'learn' how to play at being crazy. These children possessed images of madness which were constructed from popular conceptions. Similar to my study, these children were also rewarded by the hospital staff for adopting 'crazy' symptoms. (And although the child may have initially, consciously played the role of being 'crazy,' through the repeated acting out of the role, combined with the reinforcements from the environment, he ultimately internalized this behaviour, and thus, it became an unconscious part of his behavioural pattern.

In essence then, both Cain's study and this author's study indicate that the institution's perception of the individual produces a self-fulfilling prophecy whereby the patients come to be shaped in the image that the staff have of them. Just as a teacher's expectations in the classroom can produce brighter or duller students, and the expectations of a researcher can produce his own reality, my study reveals that the

expectations of the institutional staff have similar effects.

FOOTNOTES FOR CHAPTER SIX

- A total institution may be defined as: "an institution... with encompassing tendencies...Their encompassing or total character is symbolized by the barrier to social intercourse with the outside that is often built right into the physical plant, such as locked doors, high walls, barbed wire, cliffs, forests, or moors " (Goffman, 1961:4).
- This is the case in most total institutiona. See also, Sykes (1958:68); and Cohen (1953:19).
- 3 See also, Johnson and Dodds (1957) for a similar example.
- This phrase was initially used by Goffman (1961) in his study of total institutions.
- For a similar discussion of the prescriptions for behaviour in a mental hospital, see Smith and Thrasher (1968:316-324).
- Mezer and Rheingold (1962:827-831) discuss the civil/legal implications of being admitted to a mental hospital. Once institutionalized, the patient loses a large number of his civil rights which he/she previously took for granted. Specifically, once hospitalized, the individual is not allowed to make a will, a deed or contract, cannot receive property, loses the right to vote, operate a motor vehicle, cannot get married or divorced, etc. Mezer and Rheingold emphasize that when an individual loses his civil rights, this loss is not selective—it is not assumed that his presumed 'illness' has affected certain areas of competency and not others. Because he has been adjudged as mentally ill, this definition strips him of all his civil rights despite his actual ability to carry out his rights in certain areas.

- 7 See Sykes (1958:73) for a similar discussion concerning the institutionalization of prisoners.
- 8 This theme will be fully discussed in the latter part of this chapter.
- 9 This time is variable depending on the particular patient involved.
- 10 This term was initially used by Goffman (1961).
- If For the patient who remains in hospital for a short period of time, role internalization may not occur, but if he/she is hospitalized for a long time or has been repeatedly institutionalized, the person ultimately internalizes this behaviour.
- See Rosenthal and Jacobsen (1968:19-23).
- 13 See Orne (1962:776-783).
- 14 The in-patient phase in the career of the mental patient may be likened to the anthropological notion of 'rites of passage.' Essentially, this term refers to 'rites which accompany every change of place, state, social position and age! (Van Gennep, 1969:95). All rites of passage are comprised of three stages: (1) separation; (2) marginality or liminality; and (3) aggregation. During the first phase, symbolic behaviour functions to detach the individual from a fixed point in the social structure, from a set of cultural conditions or from both. That is, this symbolic behaviour, in the form of degradations and humiliations serve to strip the individual of his previous status. During the second phase, the liminal phase, the characteristics of the person are unclear. According to Turner (1969:95), "they are neither here nor there; they are betwixt and between the positions assigned and arrayed by law, custom, convention and cere-

monial."—individuals possess few if any of the attributes of their past identity. Moreover, they are often even stripped naked to symbolize that they have no status. Individuals are forced to humble themselves. They must obey their superiors implicitly. As Turner (1969:96) states: "it is as though the subjects are being reduced or ground down to a uniform condition to be fashioned anew..."

In the third phase, termed as aggregation or reintegration, the individual is reborn as a new being; he is now in a stable social arrangement once more. In short, the individual has been redefined as a clearly-defined structural type, and as such, is expected to behave in terms of this new definition. The reader can clearly see that the in-patient phase in the career of the mental patient has striking similarities with the notion of rites of passage—in symbolic terms, the patient is also stripped of his old status through sometimes degrading rites, he is reduced to a liminal being, then he is reborn with a new identity bestowed upon him by members of society and is obliged to act in accordance with this new status.

CHAPTER SEVEN

THE POST-PATIENT PHASE IN THE CAREER OF THE MENTAL PATIENT

As the previous chapter indicated, the individual enters the mental facility with a positive self-conception that was made possible by the stable arrangements of society. However, during admission procedures and thereafter, the patient discovers that it is virtually impossible to maintain his/her old identity. The in-patient is subjected to various admission procedures which function to destroy the person's conception of self. Once stripped to a liminal being, the institution offers an alternative identity of 'mental patient.' Although the inpatient may initially reject this label, through the constant reaffirmation of the institutional staff, combined with the structure of the privilege system, the patient is ultimately forced to adopt this redefinition of self—a generally permanent label and stigma that affects his/her further participation in society.

Once the mental patient is discharged from the hospital, there begins for him/her the task of rebuilding a more positive identity. The ex-patient attempts to forget his/her hospital experience; he/she desires to transform his/her deviant identity and begin life from where he/she left off prior to hospitalization. However, the post-patient quickly dis-

covers the difficulty of stopping a career in midpassage due to the numerous barriers and obstacles placed in front of him/her by society.

Generally, upon return to the community the ex-patient is confronted with unfavourable societal attitudes toward mental patients and mental illness in general. Due to the widespread negative social definition accorded mental patients, such individuals are persons with a stigma -a stigma which threatmens their social status as 'normal' members of society. Mental illness then, is perceived by post-patients as an attribute that is deeply discreditable to their identities. For other stigmatized individuals whose 'differentness' or 'failing' is immediately apparent to others, their task centres on the management of apprehension and tension in the context of social interaction with 'normals.' However, for most ex-mental patients, their differentness is not readily visible to others—thus, their task centres on the management of undisclosed discreditable information about their selves in order to control the stigma potential of mental illness, thereby enabling them to 'pass' as normal members of society.

It will be the purpose of this chapter to deal with the postpatient phase in the mental patient's career with specific reference to:

- (1) the perception of mental illness as stigma—how ex-patients define their condition as discreditable and objectionable; and
- (2) the major strategies employed by ex-patients to manage such undisclosed discreditable information about their selves in order to mitigate the potential negative impact of mental illness on their daily round.

The Perception of Mental Illness as a Stigmatized Condition

Stigma, although having an objective basis in the 'real world,' can also be conceptualized as a subjective feeling that arises from a person's perception of self as possessing some discreditable attribute or condition. Some studies on stigma assert that the stigmatized come person comes to see his/her condition as discreditable as a result of interaction with others who react to his/her condition in a negative manner. However, as others note, the stigmatized not only learns the social meaning of his failing through direct exposure to 'normals' who reject him/her, but also through the person's acquired knowledge of cultural conceptions concerning 'people of his kind.' That is, by virtue of his participation as a member of society he is aware of the negative societal attitudes attributed to his condition. Despite the fact whether the individual learns the social meaning of his failing through negative experiences with others in the context of interaction, or through his acquired knowledge of public attitudes toward his attributes or condition, the stigmatized ultimately decides to control such discreditable aspects of his identity in order to avoid further or potential negative reaction.

Examination of the data reveals that ex-mental patients come to perceive their condition as discreditable through both processes. Specifically, in terms of the former, some post-patients learn the social meaning of their failing through interaction with others who respond to the expatient's identity with a mixture of fear, rejection and disapproval:

People treated me differently once I got out of the hospital. They thought I was 'crazy.' They treated me so coldly. People I used to be good friends with now shunned me. They wanted nothing to do with me because I was in a mental hospital. (Interview #52, February 22, 1981:6).

Another discharged patient, discussing the fear of society toward exmental patients states:

No one really knows what it's like to be a mental patient—they're scared of us. When we get out, people act as though we were weired, dangerous, crazy beings...They run from us—afraid that we might do something to them (Interview #25, January 26, 1980:4).

Similarly, a third post-patient states:

When you're on the outside, it's like you have the 'mark of Cain.' When they find out you were a patient they treat you so meanly. They treat you like you were now, as a result of your hospitalization, somehow less of a person! (Interview #22, January 17, 1980:10).

For other ex-mental patients, they come to perceive their condition as discreditable through their awareness of societal conceptions (or misconceptions) regarding mental patients and mental illness. One ex-patient discussing the lack of societal understanding of mental patients and the corresponding stereotypical image they have of such individuals states:

The reason I decided to hide the fact that I was in a mental hospital was not because I had negative experiences with people who knew I was a patient. But basically, I did so because of the public misunderstanding of people like us. Let me explain. You see, the problem with this world is that no one understands what mental illness is all about. They have this stereotype of mental patients as madmen, or lunatics. That's why the public is so scared and stay away from us as if we had the plague. (Interview #28, February 6, 1980:4).

A second post-patient supports the idea of societal stereotypes affecting their re-entry into society when he says:

Most of society is mixed up about what mental illness really is. They have all these wild ideas about the way we walk, talk and act. They mostly believe that we are going to do them some harm—That we're 'out of our minds.' They don't understand that mental illness is a sickness, like all the rest. There's no difference (Interview #50, February 14, 1981:5).

Another individual, contrasting the societal conception of mental illness with that of the physically disabled says:

No one understands us. We are regarded worse than cripples you know. Cripples get more sympathy from the public—we (mental patients) get nothing. They just think of us as 'insane beings' and stay as far away from us as possible. No one sympathizes with us. On TV, they have telethons for cripples—the public is educated about the disease and feels sorry for them and tries to help. But do you ever see a telethon being held for mental patients? No way! (Interview #31, February 22, 1980:10).

In sum, the post-mental patient, either through direct exposure

to rejection, awareness of negative cultural conceptions, or both, comes to perceive the social meaning attached to his failing as threats to his/her status as a normal member of society. As a result, the ex-patient attempts to construct various strategies in order to control such discreditable information about self.

Strategies of Information Management to Control the Stigma Potential of Mental Illness

I Dissociation/Avoidance of Others with the Same Stigma

One of the major strategems of information management employed by ex-mental patients is to dissociate themselves and avoid contact with others of their own kind. That is, such individuals fear that interaction with other ex-mental patients might be observed by familiar 'normal' others, arousing these others' suspicions and leading them to discover the person's true identity. As a result, some post-patients attempt to avoid or limit such occurrences with other stigmatized individuals. One patient discussing this ploy states:

I was in this restaurant one day having a cup of coffee with an acquaintance of mine when all of a sudden I looked over to the booth next to me and saw a girl who I used to share my room with on the ward back in the hospital. I glanced over to her and she glanced back, but neither of us let on that we knew each other. Neither of us wanted the people we were with to find out the truth about us, so we acted as if we didn't know each other. (Interview # 52, February 22, 1981:7).

While in some instances ex-patients can manage to avoid interaction with others of their own kind without extreme difficulty, other cases however, prove to be problematic. Some post-patients, upon discharge are able to return to their families; other however, for various reasons, are unable to do so and are placed by caseworkers into boarding homes, approved homes, and homes for special care—facilities specifically employed to house ex-mental patients. As a result of such placement, discharged patients are forced to interact with others of their own kind on a day-to-day basis—a situation which serves as a constant reminder of the individual's past experiences as a mental patient, thereby impeding the rebuilding of a more positive identity. One post-patient, discussing her social situation states:

I hate living in this boarding home. The thing I hate most is the other people in there—the other patients. Each time I see them it reminds me of my sickness, that I was in the hospital as a patient. In fact, the whole damned boarding house reminds me of this. I just want to leave this place and get on my own. That way, I would feel better about myself (Interview #52, February 22, 1981:4).

Similarly, another patient states:

It's not very good living in a boarding home. I'd like to move somewhere else. I wouldn't even mind if I went to an 'ordinary boarding home.' You know, a boarding home where I could live with a family and not with a whole bunch of patients from the hospital. I just want to forget the past but that's hard to do when you're living with fourteen other patients (Interview #53, March 13, 1981:5).

Not only does forced association with other stigmatized individuals pose an an obstacle for the ex-patient's reorganization of self but also serves to heighten his fear that others will discover his true identity. One post-patient discussing this problem states:

THEFT

I don't dare have my friends over to the boarding home. If they ever found out that I was living with other mental patients, I don't thin they'd react very positively. I tell them that I'm just renting a room somewhere. I'm just afraid that one day they'll come over and find out about me (Interview #53, March 13, 1981:3).

In response to this undesirable and anxiety-provoking situation, some ex-mental patients leave such facilities and find places to live on their own where they are better able to control such discreditable information about their selves. One discharged patient, in his desire to dissociate himself with other ex-mental patients and thereby mitigate the stigma potential of his failing, moved from a boarding home to the Salvation Army where he seeks to pass as a transient—an identity that is also discrediting, but one which he perceives to be the lesser of two social evils:

I live at the Salvation Army now. I've been there for over a year. I used to live in one of those boarding homes but I didn't like it much. They guys used to be getting into fights all the time—that's what happens when a bunch of patients live together. And you'd have to share your room with two other patients. It's much better at the Sally Ann—here you got your own room. If there are fights you can shut your door. But what I like best about it is that I get to associate with more 'normal' people here. You get transients and guys out of jail in here, but it's better than associating with

those patients in the boarding homes....Here at the Sally Ann, the other guys think that I'm 'one of them' and I don't tell them any different (Interview # 55, April 6, 1981:3).

Just as some post-patients attempt to avoid association with others of their own kind for the reasons mentioned above, so too do they avoid certain social settings, such as ex-mental patient clubs and patient drop-in centres for the same reasons. As one post-patient states:

I don't like to go to the Drop-In Centre. I'm afraid that one time, I'll be walking out the door and I'll bump into someone who doesn't know that I was in the mental hospital—someone who wouldn't be understanding. But it's not only that I'm afraid that someone will see me coming or going from the Centre and will treat me meanly, but it's also because of the people that I don't go there very often. You see, this place is for patients who used to be in the hospital—that's the kind of people that go there. I'd rather belong to other groups instead of this mental patient one. That's why I joined a group called 'Happy Haven'—a group of ladies who meet in the north end of the city once a week—they aren't mental patients, they're just normal people (Interview #53, March 13, 1981:2).

In short then, in an effort to control the stigma potential of their failing and achieve a more positive identity, some ex-mental patients avoid certain types of social settings and instead, frequent social groups comprised of 'normal' others.

II Selective Concealment as a Strategy of Information Management

Another widely-employed strategy of information management adopted by post-mental patients is the technique of selective concealment of their discreditable aspects of self. Similar to Edgerton's (1967) study of the mentally retarded, my findings reveal that some post-patients, at certain times and in certain situations actively seek to conceal their failing; however, at other times and in other situations they freely disclose such information about their selves. Whether such individuals choose to conceal or disclose their discreditable features of self is contingent upon a number of variables including the ex-patient's perception of the stigma of mental illness, previous negative experiences with normal others, the type of social situation, and the post-patient's perception of how specific persons may react.

Examination of the data suggests that post-patients carefully segregate societal members in terms of those who can be trusted about their true identity and others who cannot. In general, family members and longterm friends were considered by ex-patients to be 'trustworthy' and 'understanding.' One post-patient, discussing the rationale behind his disclosure to close friends and family members states:

I don't tell everyone about my being in a mental hospital. The average person doesn't respond to such news very positively. But I did tell certain people—some of my close friends and my family. My parents were wonderful and so were my friends. They stood by me all the way and supported me (Interview #55, April 6, 1981:4).

Similarly, another ex-patient states:

The people I told about my sickness and my admission to the mental hospital were my aunts and uncles and some of my friends who also had nervous breakdowns—I confided in them because I knew that they would know what I went through and be understanding (Interview #53, March 13, 1981:5).

While certain individuals are considered by ex-mental patients to be 'safe others'—people who 'won't respond negatively' to their failing, others however, are classified as 'risky others.' The ex-patient, as a result of previous negative reactions from certain normal others regarding his condition, or due to his perception that certain others may respond with rejection and disapproval, concludes that the best strategy is to conceal such information about self from such 'risky others', thereby mitigating the stigma potential of mental illness. As one ex-patient states:

I had one bad experience with my neighbour and from then on I decided to hide the fact that I had been in the mental hospital from the other neighbours. When I got out of the hospital and went back to live in my apartment, I confided in one of the neighbours—I thought I could trust her but I was wrong. She didn't understand about mental illness is and as a result, she didn't want to come over and talk with me, or even associate with me anymore. That's when I decided to keep my hospitalization a secret from the other neighbours (Interview #58, April 25, 1981:2).

Another patient discussing the reason behind her selective concealment from certain friends states:

I have some friends but I didn't tell them that I went to the hospital. The reason was that I didn't think that they would be sympathetic to my problems. You see, I didn't think that they really understood what mental illness is all about—they just had some stereotypic image of mental patients. If I told them that I was once a mental patient, they would have probably shunned me. That's why I decided to keep it a secret (Interview #52, February 22, 1981:2).

Just as some post-patients employ the stragey of concealment when in the presence of 'risky others', they also do so in certain 'risky situations.' Specifically, the case of employment lis conceived by some ex-mental patients to be one such high risk situation. These individuals, either through negative discriminatory experiences with employers, or due to the anticipated negative reaction of employers, conclude that the best strategy is to conceal such discreditable information about their selves in situations of employment. One expatient, discussing the discriminatory attitudes of employers toward exmental patients states:

After you've been in the hospital and go to apply for a job, like in the factory, and they find out that you were in a mental hospital, they won't hire you. They won't take the risk (Interview #50, February 14, 1981:5).

Another post-patient speaking from experience recalls:

When you go out and look for a job, the employer asks what you've been doing the past few years. What do we tell them?—that we've been in a mental institution for the last two years? If we tell them the truth, they'll never hire us. No one is willing to take a chance with us. That's what

happened to me when I went to apply for this job. I told the truth and the employer told me, "Don't call us, we'll call you.'....So you see, any way you look at it, ex-mental patients can't get decent jobs by telling the truth. Just like negroes or some other group, we're also discriminated against (Interview #28, February 6, 1980:4).

Ex-patients then, as a result of disclosing their identity to employers, find themselves being discriminated against. The only kinds of employments made available to such persons are low-paying menial tasks--degrading jobs which function as impediments to achieving a more positive identity. As one individual remarks:

Before I went to the hospital, I was working at Massey-Fergusson for top money, but it's different when us patients get out. When patients get out they are forced to work for minimum wages, if that...I was offended at what they offered me when I finally graduated from that hospital. I was offered a job scraping the paint off old tool boxes, that's all. And the money wasn't very good. That's what's wrong with the system—once us patients get out of the hospital, we are forced to take degrading jobs. Don't get me wrong. I know that someone has to do the 'dirty' jobs too, and I'm willing to take my turn, but not all the time. It just isn't fair (Interview #55, April 6, 1981:2).

Thus, in order to combat such discrimination which limits their participation in normal social roles, specifically occupation roles, many ex-patients decide that the best strategy to employ is concealment of their true identity:

The only way I can get a job is if I lie about my past history—don't tell them that I was a mental patient. That's the only way to get 'normal' decent jobs (Interview #27, February 6, 1980:4).

III Voluntary Disclosure as a Management Technique

Just as some ex-mental patients attempt to mitigate the stigma potential of their failing by employing the strategy of selective concealment, others utilize the method of voluntary selective disclosure to achieve the same aim. Specifically, many post-patients often employ a type of disclosure referred to as 'preventive telling' 13 method of disclosure used to influence normal others' attitudes and/or behaviour toward self and ex-mental patients in general.

One type of preventive disclosure occurs in situations where the ex-patient anticipates that certain normal others will eventually find out about their discreditable identity and consequently reject him/her. Thus, in order to minimize the pain of subsequent rejection, such postpatient conclude that the best strategy to employ is preventive disclosure early in the relationship. As one ex-patient states:

I don't really like to tell people I was in the hospital, but if you don't and they find out about it later, they ask you why you kept such a thing a secret. As a result, they usually reject you—they don't want anything to do with you anymore. That's why it's better to tell these people at the beginning of a friendship and if they react negatively, well, it's less painful for you (Interview # 52, February 22, 1981:6).

Similarly, another individual states:

I've learned that it's best to tell some of your friends about it right away. That way if they freak out about it, at least you haven't wasted all that time in a false friend-ship (Interview #59, May 1, 1981:2).

A second form of preventive telling occurs in situations where the post-patient desires to enlighten normal others about mental patients and mental illness in general. That is, in reaction against popular stereotypic images of mental patients as 'dangerous, insane beings' who are something 'less than human,' the ex-patient attempts to inform others that mental illness is a medical problem—a condition that occurs 'be-yond their control' and can 'affect virtually anyone.' In short, through the strategy of preventive disclosure, the post-mental patient hopes to influence normal others' attitudes and behaviour toward individuals of their kind, thereby mitigating the stigma potential of mental illness on their selves and daily round:

I decided to tell people about me being a mental patient basically because people out in society have so many misconceptions about us. This is the result of how we are portrayed on television—the media are the ones really responsible. I try to tell people I meet that mental illness is a sickness like all the rest. I tell them that one day it could happen to them too. It is something that is beyond their control—they can't help it. By telling them this, they can be more sympathetic, and won't keep their distance from me (Interview #50, February 14, 1981:7).

Similarly, another ex-patient states:

I tell people that I was in the mental hospital because I want to help others who are facing the same predicament as me. Society is misinformed about what mental illness is. They have these strange ideas about mental patients and as a result, they've come to be afraid of us or despise us. Either way you look at it, they definitely don't

like us. By telling people about what mental illness is, by teaching the AI hope that some of them will really begin to understand what it's all about and treat us more humanely as a result (Interview #55, April 6, 1981:4).

Summary

When the mental patient leaves the hospital and re-enters the community, his/her problems are far from being over. Although the expatient desires to abandon the negative self-image of mental patient and begin the restoration of a more positive identity, he/she finds out that certain barriers hinder such aspirations. That is, the ex-mental patient discovers that due to the widespread negative public attitudes toward mental patients and mental illness in general, individuals of his/her kind possess a stigma, a stigma which impedes the rebuilding of a more positive identity, and also limits his/her participation in normal social roles.

In response to this undesirable social situation, many postpatients construct various strategems of information management such
as those discussed above—ploys designed to mitigate the negative impact of their failing on their selves and on their daily round.

FOOTNOTES FOR CHAPTER SEVEN

- For a general discussion of stigma, see Goffman (1963). See also Ray (1961) for a discussion of the difficulties ex-addicts face in attempting to transform their addict identity.
- For example, blindness or a facial deformity. For a detailed discussion of the management of discredited identities, see Davis (1961a:120-132); Barker et al. (1953); and Wright (1960).
- In contrast to certain labelling theorists, most notably Scheff (1966), who present a view of mental patients as powerless victims—passive individuals who readily accept this redefinition of self and its corresponding negative consequences limiting further participation in society, this thesis subscribes to the view that ex-mental patients are strategists who construct various ploys of information control—strategies which they hope will control the negative impact of having been a mental patient on their self-images and on their daily round. For a somewhat similar discussion of the strategies of information control used by the mentally retarded in order to 'pass' as normal societal members, see: Edgerton (1967).
- 4 See for example, Hunt (1966).
- 5 Becker (1963).
- For a discussion of such negative societal conceptions, see: Cumming and Cumming (1956), (1957); Lamy (1966); Nunnally (1957); and Woodward (1951).

- 7 Most ex-patients interviewed limited such occurrences in the sense that they only met with others of their own kind in 'safe places'—locations where they felt that normal others would not observe them.
- 8 See also Greenwald's (1958) study of prostitutes for a similar example.
- 9 Such reasons include having no family to which to return, or that the family no longer wants the individual.
- Approved homes, funded by regional government, are residential houses in which a family take in a few discharged patients to live with them. Homes for Special Care, funding by the provincial and federal governments, are facilities (sometimes residential homes or nursing homes) designed to care for post-mental patients as well as other individuals. Boarding homes, funded by regional government, are homes designed to care specifically for ex-mental patients.
- See also Goffman (1963) and Schneider and Conrad (1980) for parallel examples.
- See Miller and Dawson's (1966) study on the effects of stigma on the re-employment of ex-mental patients.
- This term was drawn from Schneider and Conrad's (1980) study of epileptics.
- Schneider and Conrad (1980) found that epileptics also employ this strategy of preventive telling in similar situations for similar ends.

CHAPTER EIGHT

SUMMARY AND CONCLUSIONS

This chapter is concerned with an overview of the career of the mental patient based on the analysis in the preceding chapters.

Included in this chapter are the conclusions, limitations of the study, implications, and suggestions for future research.

This thesis has presented an updated study of the social world of the mental patient as he/she subjectively experiences it within a Canadian psychiatric setting. Advocating a symbolic interactionist perspective, specifically a social-role or labelling approach to the study of mental illness, emphasis was placed on the social processes-by which an individual is segregated, labelled, and subsequently treated as mentally ill. Moreover, this study has also focussed on the individual's reactions to such labelling, the extent to which he/she adopted this redefinition of self as mentally ill, and the consequences of this new identity on the person's future participation in society.

As the data suggested (and is supported by other research), the career of the mental patient began when societal members reacted to evidence of culturally-inappropriate behaviour or residual rule violations exhibited by the actor and subsequently defined him/her as mentally ill.

Although the social audience may have initially denied or rationalized the peculiar behaviour, if the actor repeatedly failed to correctly respond to the interpersonal demands of the other in the situation, or if the actor's behaviour became unmanageable, societal members moved toward defining him/her as mentally ill.

As the study indicated, a number of agents and agencies participated in the definitional process and subsequent hospitalization of the individual. One such possible agent was the individual himself/herself.

Of the subjects interviewed, a small number, upon self-examination discovered themselves to be acting in a strange and incomprehensible manner—behaviour which was frightening and led such persons to believe that they were losing their minds. Such a view of self as being potentially mentally ill impelled these individuals to seek psychiatric help.

While some individuals referred themselves to the hospital upon the self-realization that they were in need of psychiatric help, others, for a time, attempted to conceal their newly discovered discreditable image of self. Such persons struggled to maintain their roles as 'normal' individuals in the context of social interaction with others while constantly afraid that they will be 'found out.' It was only when the social situation became too stressful that these individuals decided to admit themselves into the psychiatric institution.

A second agent who frequently played a role in the individual's passage from person to patient was the legal system. As the data indicated, a number of individuals marked the beginning of their careers as mental patients when legal authorities interpreted their actions as evi-

dence of mental illness and consequently admitted them to a mental hospital. Specifically, legal authorities, upon observing individuals exhibiting strange behaviour—committing residual rule violations, apprehended and admitted such persons directly to the psychiatric hospital for treatment. While the legal authority conceived of hospitalization as a necessary step in order to ensure that the person would not cause injury to himself/herself or to others, the individual conceived of his/her apprehension and involuntary hospitalization in a negative manner. For persons who were admitted to the psychiatric hospital by the police as a result of committing some residual rule violation, such action was felt to be totally unjust. As the data suggested, such individuals generally felt that nothing was mentally wrong with them; neither had they committed any legal offense. Therefore, according to their logic, such individuals were the victims of a gross misjudgment on the part of the relice.

While individuals, admitted to the psychiatric facility by the police as a result of exhibiting strange or incomprehensible behaviour, expressed the moral outrage of their social situations, others, admitted by the court as a consequence of committing some legal offense, conceived of their situations in a somewhat different manner. The data indicated that although such pre-patients were not in accordance with the adjudgment of their selves as potentially mentally ill, they were informed by court officials of the rationale behind the judgment, the course of action that would be taken, and the length of time required to stay in the hospital—information which made their social situations more bearable.

A third group of agents who often played a role in the hospitalization of the individual were his/her family and friends. As the study indicated, during the pre-patient phase of the individual's career, he/she gradually discovered that the significant others on whom he/she relied to be supportive in times of trouble, were the very persons to first question the person's sanity and initiate commitment proceedings. Subsequent to interpreting the pre-patient's behaviour as evidence of mental illness, friends or family members enlisted the aid of certain official third parties such as the family physician, mental health officials, justice of the peace, etc., in committing the individual to the psychiatric facility. Such action was generally viewed by the pre-patient as traitorous and conspiratorial in nature.

A fourth group of agents who sometimes played a role in the individual's passage from person to patient were psychiatric officials of a general hospital. Some individuals interviewed, upon exhibiting symptoms of mental illness were reacted to by various societal members and were admitted to the psychiatric ward of a general hospital for treatment. When such individuals failed to respond to treatment, or were diagnosed as being in need of extended psychiatric care, they were subsequently transferred by psychiatric officials to the mental hospital for further treatment. While some individuals interviewed responded to such action in a positive manner, the majority responded to the action of transfer in a negative manner. In terms of the latter group, such individuals viewed this experience with fear and anxiety. For such persons, the action of transfer represented a further loss of freedom.

A significant turn occurred during the in-patient phase of the person's career. As the data suggested, prior to arrival at the psychiatric institution, the individual possessed a positive self-conception made possible by the stable arrangements of society. However, upon admission and thereafter, the individual found that it was virtually impossible to maintain his old identity. Specifically, on entering the hospital, the in-patient is separated from the individuals and structures which validated his/her behaviour. Such a social separation functioned to destroy the person's conception of self. Moreover, the data suggested that self was also destroyed in yet another manner—specifically, through the various hospital admission procedures to which the person is subjected.

Once stripped to a liminal being, the institution offered the patient an alternative self-identity of 'mental patient'—a self-conception that was incompatible with his/her prior conception of self. As the data indicated, although the in-patient initially denied that he/she was mentally ill and thus rejected this redefinition of self, through the context of the privilege system's rewards and punishments, the in-patient was ultimately forced to adopt this new self-identity—an identity which affected his/her further participation in society.

Upon release from the psychiatric facility, the person entered the post-patient phase of his career—a phase generally characterized by stigmatization. As the study indicated, once discharged, the person sought to rebuild a more positive self-identity; however, numerous barriers posed as threats to achieving such a goal. Due to the pre-

vailing negative social definition accorded mental illness and mental patients, such individuals are persons with a stigma—a stigma which threatens their social status as 'normal' societal members. In order to mitigate the potential negative impact of mental illness on their self-identities, such post-patients employed various techniques of information management, ranging from selective concealment to voluntary disclosure, thereby enabling them to live their lives as 'normal' members of society.

As with most research, the present study does suffer from certain limitations. In order to gather data on the social life of the mental patient as he/she subjectively experienced it, strict statistical measurements or controls were not employed. It is the contention of the researcher that one cannot achieve a full understanding of certain phenomena by merely statistically examining the relationship between social facts or the causes of social phenomena. Understanding can only be achieved by focussing on the subjective states of the individuals themselves.²

A second methodological limitation centres on the retrospective nature of the research. With respect to the pre-patient phase of the mental patient's career, information was collected based on retrospective accounts of individuals' experiences as pre-patients rather than following persons firsthand as they moved through this first stage in their moral careers.

A third and final limitation of this research focuses on its onesided nature. Due to time constraints, the researcher was unable to conduct a holistic study of the social world of the mental patient. Some sociologists contend that by focussing solely on one group, to the exclusion of another, will undoubtedly lead to a biased and distorted presentation of the findings. In response to this charge, in accordance with Becker (1966:240), I would argue that one cannot avoid taking sides when conducting fieldwork. But taking sides does not necessarily mean that our findings will be distorted and therefore be rendered useless. By employing our theoretical and technical resources impartially, by paying attention to our feelings as we conduct the study, and by limiting our conclusions, researchers can produce valid and reliable findings.

THE STATE OF STATE

Despite these limitations, the data reported here lead to some interesting conclusions and suggestions for future research. This study, to a certain extent, replicates some of the findings of previous studies on the career of the mental patient. Specifically, similar to Goffman's (1961) study, my data indicate that individuals, during the pre-patient phase of their moral careers, sometimes undergo experiences which they conceive as alienating and treacherous in nature. Moreover, this study supports previous research on the in-patient phase of the mental patient's career. Although patients are subjected to more 'humane' treatment than in the past, and are now forced to remain in hospital for shorter lengths of time, they are still subjected to various institutional procedures which function to strip the person of his former identity and force the person to adopt this new identity of mental patient.

While this research replicates some of the findings of previous investigations, it also adds to our knowledge concerning the consequences

of institutionalization on the identities of mental patients. Whereas previous studies have generally portrayed ex-mental patients as power-less individuals who readily accept this redefinition of self as mentally ill and its corresponding negative consequences limiting his/her further participation in society, the findings of this present study suggest that ex-mental patients are more accurately conceptualized as strategists who actively construct various tactics of information control in order to mitigate the stigma potential of mental illness on their daily round.

The implications of this research are two-fold. First, from an analysis of the data one is led to conclude that mental institutions, in their present form, should be abolished. As Townsend (1978) and others have stated, mental hospitals like prisons and other institutions are counterproductive in the sense that they perpetuate exactly what they profess to erase. Lambo (1964), a Nigerian psychiatrist supporting the deinstitutionalization of mental health services from institutional to community-based care, has suggested that, in place of hospitalization, individuals with acute emotional problems would visit 'therapeutic communities'—communities that are not separated from society. Family members would accompany the person to the therapeutic community and would be actively involved in the treatment of the individual. For others, who have transitory behavioural problems, or were improperly socialized, Lambo (1964) suggests the construction of half-way houses, community clinics and day-care facilities.

A second implication of this research centres on the powerful and deleterious effect of staff expectations on the identities of hos-

played by the patient began as a conscious pretense in order to gain more privileges, through the constant reinforcement of the staff, the patient ultimately internalized such behaviours. In short, the institutional staff's perception of the patient produced a self-fulfilling prophecy whereby he/she came to be shaped in the image that the staff had of him/her.

While the present study was descriptive and exploratory in nature, it has provided insight into our understanding of the processes and consequences of institutionalization upon the self-images and identities of mental patients. In terms of future research, further attention needs to be given to specific aspects in the three phases of the career of the mental patient.

One such area of concern that needs further investigation centres on the role played by the pre-patient in the decision to hospitalize himself/herself. Although researchers in the area of medical sociology have discovered the differential willingness of certain social classes to define themselves as ill, few such studies have been conducted with reference to mental illness. While the present study has touched upon this issue, further systematic examination is needed.

On a methodological level, the majority of studies (including the present one), with respect to the pre-patient phase in the patient's career, have been descriptive and/or retrospective in nature. Thus, the data collected may have been subject to distortion and/or forgetting by the patient. Future investigations should be longitudinal in nature—such studies should be designed and executed in a manner which would

follow the individual from the period prior to committing a residual rule violation until the time he/she is admitted into the psychiatric facility.

With respect to the post-patient phase of the mental patient's career, future research should focus on the effect different types of hospitalization have on the self-images and behaviour of post-mental patients. With the one recent exception of Howard et al. (1980), few attempts have been made to examine the effects of such differences.

Another area of potential research centres on the reaction of family members to the post-patient. With the exception of Freeman and Simmon's (1963) study, the majority of research in this area has merely focussed on such variables as differences in family type and social class factors to the exclusion of the <u>interaction</u> between family members and the post-patient himself/herself.

In order to gain a holistic understanding of the moral career of the mental patient, it would be useful for future research to empirically compare how different cultures, theories or conceptions of mental illness affect the symptomatology and course of the illness. While societal conceptions of mental illness have been compared between non-Western and Western societies, little attention has been placed on comparing cultural conceptions of insanity between Western societies.

Despite the need for more research, the present study has provided a contribution to our understanding of the processes and consequences of institutionalization on the self-images and identities of
mental patients and toward the formulation of a comprehensive sociological

theory of mental illness in general.

FOOTNOTES FOR CHAPTER EIGHT

- See Scheff (1966); Laing and Esterson (1964); and Mechanic (1962).
- This is not to say that qualitative research represents an end in itself. The present study merely represents an initial analysis from which hypotheses can be constructed and subsequently tested at a later time.
- Although it would have been ideal to follow pre-patients firsthand as they moved through the first stage of their careers, such
 an aim was difficult to achieve. Due to the numerous problems encountered in attempting to make a bargain with the psychiatric institution, combined with certain time constraints to complete the
 research, I was forced to rely on retrospective accounts of individuals experiences as pre-patients.
- Although the researcher desired to study the situation from the point of view of the institutional staff as well, given the length of time to conduct this research, such an aim was impossible to achieve. Thus, the researcher decided to focus on one side—the side of the mental patient and present it faithfully.
- 5 Goffman (1961); Goldman et al. (1970); and Scheff (1966).
- 6 See for example, Scheff (1966).

- 7 See Mechanic and Volkart (1961) for a study which has centred on this problem in medical sociology.
- 8 See Fogelson (1965); Newman (1964); and Wallace (1972).
- 9 Townsend's (1978) recent study comparing cultural conceptions of mental illness in America, and Germany is a notable exception.

BIBLIOGRAPHY

- Adams, Henry B. "'Mental Illness' or Interpersonal Behavior?" American Psychologist, 19, 1964, 191-197.
- Adams, R.N., and J.J. Preiss (eds.), <u>Human Organization Research</u>, Homewood, Illinois: Dorsey Press, 1960.
- Babchuck, N. "The Role of the Researcher as Participant Observer and Participant as Observer in the Field Situation." Human Organization, 21, v. 3, 1962, 225-228.
- Balance, W.D.G. (et al.) "Mental Illness: Myth, Metaphor or Model." Professional Psychology, I, 1970, 133-137.
- Balint, M. The Doctor, His Patient, and the Illness. New York: International Press, 1957.
- Barber, B. "Research on Human Subjects: Problems of Access to a Power-ful Profession." Social Problems, 21 (summer), 1973, 103-112.
- Barker, R.G. (et al.) "Adjustment to Physical Handicap and Illness: A Survey of the Social Psychology of Physique and Disability." Social Science Research Council Bulletin, 55, 1953.

- Becker, Ernest. "Socialization, Command of Performance, and Mental Illness." American Journal of Sociology, 67, 1962, 494-501.
- Becker, Howard S. "Problems of Inference and Proof in Participant Observation." American Sociological Review, 23, 1958, 652-660.
- New York: The Free Press, 1963.

- Becker, Howard S. (et al.) <u>Boys in White.</u> Chicago: University of Chicago Press, 1961.
- Begelman, D.A. "Misnaming, Metaphors, The Medical Model and Some Muddles." Psychiatry, 34, 1971:38-58.
- Berger, P., and T. Luckmann. The Social Construction of Reality:

 A Treatise in the Sociology of Knowledge. New York: Doubleday, 1966.
- Bernard, Jessie. Social Problems at Mid-Century. New York: Holt, Rinehart and Winston, 1957.

- Bierstedt, R. "A Critique of Empiricism in Sociology." American Sociological Review, 24, 1949, 584-592.
- Bittner, E. "Police Discretion in Apprehending the Mentally Ill."

 <u>Social Problems</u>, 14 (Winter), 1967, 278-292.
- Blake, R.R., and J.S. Moulton. "Conformity, Resistance and Conversion." in Conformity and Deviation, I.A. Berg and B.M. Bass, (eds.), New York: Harper, 1961.
- Blumer, Herbert. "Society as Symbolic Interaction." in <u>Human Behavior</u> and <u>Social Processes</u>, A. Rose, (ed.), Boston: Houghton Mifflin Co., 1962.
- Herbert Mead." American Journal of Sociology, 71, (March), 1966, 535-544.
- New Jersey: Prentice-Hall, 1969.
- Bogdan, R., and S. Taylor. <u>Introduction to Qualitative Research</u>
 <u>Methods</u>. New York: John Wiley and Sons, Ltd., 1975.
- Braginsky, B. (et al.) <u>Methods of Madness: The Mental Hospital as a Last Resort</u>. New York: Holt, Rinehart and Winston, 1969.

- Bruyn, Severyn. The Human Perspective in Sociology: The Methodology of Participant Observation. New York: Prentice-Hall, 1966.
- Bryan, James Q. "Occupation Ideologies and Individual Attitudes of Call Girls." Social Problems, 13 (4), 1965, 441-450.
- Cain, Albert. "On the Meaning of 'Playing Orazy' in Borderline Children." Psychiatry, 27, 1964, 278-289.
- Carpenter, W.T. (et al.) "The Treatment of Acute Schizophrenia without Drugs: An Investigation of Some Current Assumptions." American Journal of Psychiatry, 134, 1977, 14-20.
- Caudill, Wm. (et al.) "Social Structures and Interaction Processes on a Psychiatric Ward." American Journal of Orthopsychiatry, 22 (April) 1952, 314-334.
- Chauncey, R. "Comment on 'The Labelling Theory of Mental Illness."

 American Sociological Review, 40, 1975, 248-252.
- Ciba Foundation Symposium. <u>Transcultural Psychiatry</u>. Boston: Little, Brown, 1965.
- Cicourel, A. Method and Measurement in Sociology. New York: Free Press, 1964.

- Cicourel, A., and J. Kitsuse. The Educational Decision-makers. Indianapolis: Bobbs-Merrill, 1963.
- Clausen, J.A., and M. Yarrow. "Paths to the Mental Hospital." <u>Journal</u> of Social Issues, 11 (4), 1955, 25-32.
- Cohen, Elie. <u>Human Behavior in a Concentration Camp</u>. New York: Grosset and Dunlap, 1953.
- Cooper, David. <u>Psychiatry and Anti-Psychiatry</u>. New York: Tavistock, 1967.
- Crespo. Manuel. "The Career of the School Skipper." in <u>Decency and Deviance</u>: Studies in <u>Deviant Behaviour</u>, J. Haas and Wm. Shaffir, (eds.), Toronto: McClellend and Stewart Ltd., 1973.
- Cumming, J., and E. Cumming. "Affective Symbolism, Social Norms and Mental Illness." Psychiatry, 19, 1956, 77-85.
- Health Education. Mass.: Harvard University Press, 1957.
- ______. "On the Stigma of Mental Illness." Community Mental Health Journal, 1, 1965, 135-143.
- Davis, F. "Comment on Initial Interaction of Newcomers in Alcoholics Anonymous." Social Problems, 8, 1961, 364-365.

- "Deviance Disovowal: The management of strained interaction by the visibly handicapped."

 Social Problems. 8, 1961a, 120-132.

 Passage Through Crisis: Polic Victims and Their Families. Indiana, Bobbs-Merrill Co., 1963.

 Davis, N. "Labeling Theory in Deviant Research: A Critique and Reconsideration."

 Sociological Quarterly, 13, 1972, 447-474.

 Sociological Constructions of Deviance: Perspectives and Issues in the Field. Iowa: Wm. C. Brown Company, 1975.

 Dean, John P. "Limitations and Advantages of Unstructured Methods."

 In An Introduction to Social Research, John T. Dobey, (ed.),

 New York: Meredith, 1967.

 Denzin, Norman K. "The Self-Fulfilling Prophecy and Patient-Therapist
- Proposed Synthesis." American Sociological Review, 34, 1979, 922-934.

Hill, 1968, 349-357.

Interaction." In The Mental Patient: Studies in the Sociology of Deviance, S. Spitzer and N. Denzin, (eds.), New York: McGraw-

Denzin, N. and S. Spitzer. "Paths to the Mental Hospital and Staff
Predictions of Patient Role Behavior." Journal of Health and
Human Behavior, 7, 265-271.

- Douglas, J.D. <u>Investigative Social Research: Individual and Team Research.</u> Beverly Hills, California: Sage Publications, 1976.
- Durkheim, Emile. The Division of Labor in Society. New York: The Free Press, 1956.
- Eaton, J., and R. Weil. <u>Culture and Mental Disorders</u>. Glencoe, Illinois: The Free Press, 1955.
- Edgerton, R.B. "Conceptions of Psychosis in Four East African Societies." American Anthropologist, 68, 1966, 408-425.
- Mentally Retarded. Berkeley: University of California Press, 1967.
- Engel, G.L. "The Need For a New Medical Model: A Challenge for Bio-medicine." Science, 196, 1977, 129-136.
- Erikson, K.T. "Patient Role and Social Uncertainty--A Dilemma of the Mentally Ill." Psychiatry, 20, 1957, 263-274.
- 9, 1962, 307-314.
- Social Problems, 14, 1965, 366-373.
- ----- Wayward Puritans. New York: John Wiley and Sons, 1966.

- Esquirol, J.E. "A Treatise on Insanity." In <u>Documentary History of</u>

 Psychiatry: A Source Book on Historical Principles, C.E. Goshen,

 (ed.), New York: Philosophical Library, 1967, 316.
- Favazza, A., and M. Oman. Anthropological and Cross-Cultural Themes in Mental Health. Columbia: University of Missouri Press, 1977.
- Filstead, W. Qualitative Methodology: Firsthand Involvement with the Social World. Chicago: Markham, 1970.
- Fine, B. "Labelling Theory: An Investigation into the Sciological Critique of Deviance." Economy and Society, 6, no. 2, 1977, 166-190.
- Fogelson, R.D. "Psychological Theories of Windigo 'Psychosis' and a Preliminary Application of a Model's Approach." In Context and Meaning in Cultural Anthropology, M. Spiro, (ed.), New York: Free Press, 1965.
- Forster, E.B. "Theory and Practice of Psychiatry in Ghana." American Journal of Psychotherapy, 16, 1962, 7-51.
- Foucault, Michel. Mental Illness and Psychology. New York: Harper and Row, 1976.
- Foulks, E.F. (et al.) (eds.) <u>Current Perspectives in Cultural Psy</u>chiatry. New York: Spectrum Publications, 1977.

- Freeman, H.E., and O.G. Simmons. The Mental Patient Comes Home. New York: Wiley, 1963.
- Freidson, E. "The Production of Deviant Populations." In Society and Rehabilitation, Marvin Sussman, (ed.), Washington: American Sociological Association, 1966a, 82-93.
- tation, Marvin Sussman, (ed.), Washington: American Sociological Association, 1966b.
- ----- Profession of Medicine. New York: Dodd, Mead, 1970.
- Gans, H. "The Participant Observer as a Human Being: Observations on the Personal Aspects of Field Work." In <u>Institutions and the Person</u>, H.S. Becker et al. (eds.), Chicago: Aldine, 1962, 300-317.
- Geer, B. "First Days in the Field." In Sociologists at Work, P.E. Hammond, (ed.), New York: Doubleday, 1997, 372-398.
- Chicago: Aldine, 1970, 81-96.
- Gennep, Arnold Van. The Rites of Passage. Chicago: University of Chicago Press, 1960.
- V. Turner, Chicago: Aldine, 1969.
- Gibbs, Jack P. "Conceptions of Deviant Behavior: The Old and the New." Pacific Sociological Review, 9, 1966, 9-14.

- Glaser, B., and A. Strauss. The Discovery of Grounded Theory. Chicago: Aldine, 1967.
- Aldine, 1971. Status Passage: A Formal Theory. Chicago:

- Gluckman, Max (ed.). Essays on the Ritual of Social Relations. Manchester: University of Manchester Press, 1962.
- Goffman, E. "The Moral Career of the Mental Patient." Psychiatry, vol. 22, 1959, 123-142.
- ----- Asylums. Chicago: Aldine, 1961.
- ----- Stigma. Englewood Cliffs, New Jersey: Prentice-Hall, 1963.
- Gold, R.L. "Roles in Sociological Field Observations." Social Forces, 36, 1958, 217-223.
- Goldman, A.R. (et al.). "On Posing as Mental Patients: Reminiscences and Recommendations." <u>Professional Psychology</u>, vol. 1, 1970, 427-434.
- Gordon, G. Role Theory and Illness. New Haven: College and University Press, 1966.
- Gordon, J.B. "The Meta-Journey of R.D. Laing." In R.D. Laing and Anti-Psychiatry, R. Boyers (ed.). New York: Harper and Row, 1971.

- Gough, Harrison G. "A Sociological Theory of Psychopathy." American Journal of Sociology, 53, 1948, 359-366.
- Gouldner, A.W. <u>The Coming Crisis in Western Sociology</u>. New York: Basic Books, 1970.
- Gove, Walter. "Societal Reaction as an Explanation of Mental Illness: An Evaluation." American Sociological Review, 35, 1970, 873-884.
- New York: Wiley, 1975a.
- Scheff." American Sociological Review, 40, 1975b, 242-248.
- Greenwald, H. The Call Girl. New York: Ballintine, 1958.
- Haas, J., and Wm. Shaffir. "Do New Ways of Professional Socialization Make a Difference?" Paper presented at the Ninth World Congress of Sociology, Uppsala, Sweden, 1978.
- borough: Prentice-Hall, 1978b.
- Maintaining Research and Researcher Bargàins." In Fieldwork Experience: Qualitative Approaches to Social Research, Wm. Shaffir, R. Stebbins, and A. Turowetz, (eds.). New York: St. Martin's, 1980.

- Habenstein, R. (ed.). Pathways to Data: Field Methods for Studying Ongoing Social Organizations. Chicago: Aldine, 1970.
- Hall, Oswald. "The Stages of a Medical Career." American Journal of Sociology, 53, 1948, 327-336.
- Hammer, Muriel. "Influence of Small Social Networks as Factors in Mental Hospital Admission." Human Organization, 22, 1963, 243-251.
- Hathaway, Craig. "The Participant-Observer as Patient: Issues in Methods and Theories of Making Sense of Self and Situations; Engaged in Anticipation of Formulating Alternative Group Therapeutic Procedures." Paper presented at the Canadian Sociology and Anthropology Association meetings, Montreal, Quebec, 1980.
- Haywood, J.C. (ed.). Socio-Cultural Aspects of Mental Retardation:

 Proceedings. New York: Appleton-Century-Crofts, 1970.
- Herman, Nancy J. "A Critical Evaluation of the Labelling Theory of Deviant Behaviour." Unpublished Paper, McMaster University, 1980.
- Hoffmann, J.E. "Problems of Access in the Study of Social Elites and Boards of Directors." In <u>Fieldwork Experience</u>, Wm. Shaffir et al. (eds.). New York: St. Martin's, 1980.
- Hollinghead, A.B., and F. Redlich. <u>Social Class and Mental Illness</u>. New York: Wiley, 1958.
- Hornosty, R.W., and Cyril Levitt. An Introduction to the Social Psychology of George Herbert Mead. Modular Instructional Package #2, McMaster University, 1980.

- Howard, L. (et al.). "Hospital Structure, Staff Perceptions of Patients and Patient Self-Perceptions: Further Examination of the Dynamics of Labeling in Three Psychiatric Settings." Paper presented at the Sociology of Psychiatric Care Session, Canadian Sociology and Anthropology Association Meetings, Montreal, Quebec, 1980.
- Huber, Joan. "Symbolic Interactionism as a Pragmatic Perspective:
 The Bias of Emergent Theory." American Sociological Review,
 38, 1973, 278-284.
- Hughes, Everett C. Men and their Work. Illinois: The Free Press, 1958.
- Hunt, P. (ed.). Stigma: The Experience of Disability. London: Chapman, 1966.
- Jewell, Donald. "A Case of the 'Psychotic' Navaho Indian Male."

 <u>Human Organization</u>, 11, no. 1, 1952, 32-36.
- Johnson, D. McI., and N. Dodds, (eds.). The Plea For the Silent.
 London: Christopher Johnson, 1957.
- Johnson, J. Doing Field Research. New York: Free Press, 1975.
- Kanter, Rosabeth M. "Symbolic Interactionism and Politics in Systematic Perspective." Sociological Inquiry, 42, 1972, 77-92.
- Kennedy, J. "Cultural Psychiatry." In <u>Handbook of Social and Cultural Anthropology</u>. J. Honigmann, (ed.). New York: Rand McNally, 1974, 1119-1198.

- Kiev, Ari. Transcultural Psychiatry. New York: Free Press, 1972.
- Kitsuse, J.I. "Societal Reactions to Deviant Behavior: Problems of Theory and Method." Social Problems, 9, 1962, 247-256.
- ----- "Editor's Preface." American Behavioral Scientist, 14, 1970, 163-165.
- Kleinman, Sherryl. "Learning the Ropes as Fieldwork Analysis." In <u>Fieldwork Experience</u>, Shaffir et al., (eds.). New York: St. Martin's Press, 1980, 171-183.
- Kluckholn, Clyde. <u>Culture and Behavior</u>. New York: Free Press, 1962.
- Kluckholn, Florence. "The Participant-Observer Technique in Small Communities." American Journal of Sociology, 46, 1940, 331.
- Kolb, Wm. "A Critical Evaluation of Mead's 'I' and 'Me' Concepts." Social Forces, 22, 1944, 291-296.
- Kringlen, Einar. "Schizophrenia in Twins." Schizophrenia Bulletin, (December). National Clearing House for Mental Health Information, 1969.
- Kuhn, Manford H. "Major Trends in Symbolic Interactionist Theory in the Past Twenty-five Years." Sociological Quarterly, 5, (Winter), 1964, 61-84.

- Kutner, Luis. "The Illusion of Due Process in Commitment Proceedings."

 Northwestern University Law Review, 57, (September), 1962, 383-393.
- Laing, R.D. The Divided Self. New York: Pelican, 1965.
- ----- Self and Others. New York: Pelican, 1969.
- Laing, R.D., and A. Esterson. Sanity, Madness and the Family. London: Tavistock, 1964.
- Lambo, T.A. "Patterns of Psychiatric Care in Developing African Countries." In Magic, Faith, and Healing, Ari Kiev, (ed.). Illinois: Free Press, 1964.
- Lamy, R.E. "Social Consequences of Mental Illness." <u>Journal of Consulting Psychology</u>, 30, no. 5, 1966, 450-455.
- Leifer, R. "The Medical Model as Ideology." <u>International Journal of Psychiatry</u>, 9, 1970-71, 13-31.
- Lemert, E. "Legal Commitment and Social Control." Sociology and Social Research, 30, 1946, 370-378.
- ---- Social Pathology. New York: McGraw-Hill, 1951.

- ------. "Paranoia and the Dynamics of Exclusion." Sociometry, 25, 1962, 2-25.
- wood Cliffs: Prentice-Hall, 1967.
- Lewis, Aubrey. The State of Psychiatry. New York: Science House, 1967.
- Lichtman, Richard. "Symbolic Interactionism and Social Reality: Some Marxist Queries." <u>Berkeley Journal of Sociology</u>, 15, 1970, 75-94.
- Manis, Jerome, and B. Meltzer (eds.). Symbolic Interaction. Boston: Allyn and Bacon, 1967.
- Manis, J. (et al.). Symbolic Interactionism: Genesis, Varieties and Criticism. London: Routledge and Kegan Paul, 1975.
- Margetts, E.L. "Clinical Universalism." Ciba Foundation Symposium. In <u>Transcultural Psychiatry</u>. Boston: Little, Brown, 1965.
- Matza, David. Becoming Deviant. New Jersey: Prentice-Hall, 1969.
- May, Rollo. "Contributions of Existential Psychotherapy." In Existence, R. May (ed.). New York: Simon and Schuster, 1967.

- Mead, G.H. "What Social Objects Must Psychology Presuppose?" Journal of Philosophy, VII, 1910, 178. ----- The Social Self. Journal of Philosophy, X, 1913, 374. pany, 1932. ----- Mind, Self and Society. Chicago: University of Chicago Press, 1934. ----. Movements of Thought in the Nineteenth Century. Chicago: University of Chicago Press, 1936. The Philosophy of the Act. Chicago: University of Chicago Press, 1938. ----. On Social Psychology. Chicago: University of Chicago Press, 1964, edited by Anselm Strauss.
- Mechanic, David. "Some Factors in Identifying and Defining Mental Illness." Mental Hygiene, 46, 1962, 66-74.
- Mechanic, D., and E. Volkart. "Stress, Illness and the Sick Role."

 American Sociological Review, 26, 1961, 51-58.

- Meltzer, Bernard H. The Social Psychology of George Herbert Mead.
 Michigan: Centre For Sociological Research, Western Michigan
 University, 1959.
- tionism: A Reader in Social Psychology, J. Manis and B. Meltzer, (eds.). Boston: Allyn and Bacon, 1972, 4-22.
- Mercer, Jane. "Social System Perspective and Clinical Perspective: Frames of Reference for Understanding Career Patterns of Persons Labelled as Mentally Retarded." Social Problems, 13, 1966, 21-34.
- Meyers, J., and L. Schaffer. "Social Stratification and Psychiatric Practice: A Study of an Outpatient Clinic." American Sociological Review, XIX, 1954, 307-310.
- Mezer, R.R., and P.D. Rheingold. "Mental Capacity and Incompetency: A psycho-legal problem." <u>American Journal of Psychiatry</u>, 118, 1962, 827-831.
- Miller, D., and W. Dawson. "Effects of Stigma on Re-employment of exmental patients." Mental Hygiene, 49, 1965, 281-287.
- Miller, D., and M. Schwartz. "County Lunacy Commission Hearings: Some Observations of Commitments to a State Mental Hospital." Social Problems, 14, 1966, 26-35.

- Miller, S.M. "The Participant Observer and Over-rapport." American Sociological Review, 17, 1952, 97-99.
- Mills, C.W. "Situated Actions and Vocabularies of Motive." American Sociological Review, 5, 1940, 904-913.
- Murphy, Jane M. 'Psychiatric Labeling in Cross-Cultural Perspective." Science, 191, 1976, 1019-1028.
- Newman, P.L. "Wild Man' Behavior in a New Guinea Highlands Community." American Anthropologist, 66, 1964, 1-19.
- Nunnally, J.C. "The Communications of Mental Health Information:
 A Comparison of the Opinions of Experts and the Public with
 Mass Media Presentations." Behavioral Science, 3, 1957, 220-230.
- Offer, D., and M. Sashbin. Normality: Theoretical and Clinical Concepts of Mental Health. New York: Basic Books, 1966.
- Opler, M.K. (ed.). <u>Culture and Mental Health</u>. New York: MacMillan, 1955.
- Orne, M.T. "On the Social Psychology of the Psychological Experiment: With Particular Reference to the Demand Characteristics and their Implications." American Psychologist, 17, 1962, 776-783.
- Parsons, Talcott. The Social System. Glencoe, Illinois: Free Press, 1951a.

- Perspective." American Journal of Orthopsychiatry, 21, 1951b, 452-460.
- The Patient and the Mental Hospital, Greenblatt, Levinson and Williams, (eds.). Glencoe, Illinois: Free Press, 1957, 108-129.
- American Values and Social Structure." In Patients, Physicians and Illness, E.G. Jaco, (ed.). New York: Free Press, 1958.
- Phillips, D.L. "Rejection: A Possible Consequence of Seeking Help for Mental Disorders." American Sociological Review, 28, 1963, 963-973.
- Ray, Marsh B. "Abstinence Cycles and Heroin Addicts." Social Problems, 9, 1961, 132-140.
- Ropers, Richard. "Toward a Phenomenological Sociology." <u>Catalyst</u>, (Winter), 1973, 15-28.
- Rose, Arnold. "A Systematic Summary of Symbolic Interaction Theory." In <u>Human Behavior and Social Processes</u>, A. Rose, (ed.). Boston: Houghton Mifflin, 1962.
- Rosen, George. Madness in Society. New York: Harper, 1968.

- Rosenhan, D.L. "On Being Sane in Insane Places." Science, 179, 1973, 250-258.
- Rosenthal, David. "Problems of Sampling and Diagnosis in the Major Twin Studies of Schizophrenia." Journal of Psychiatric Research, 1, 1961, 116-134.
- Rosenthal, R., and L. Jacobsen. "Teacher Expectations for the Disadvantaged." Scientific American, 218, 1968, 19-23.
- Roth, Julius A. <u>Timetables:</u> Structuring the Passage of Time in Hospital Treatment and Other Careers. New York: Bobbs-Merrill, 1963.
- Rubington, E., and M. Weinberg. (eds.). <u>Deviance: The Interactionist Perspective</u>. 2nd edition. New York: <u>MacMillan</u>, 1973.
- Perspectives. New York: Oxford University Press, 1977.
- Sampson, Harold (et al.). "Family Processes and Becoming a Mental Patient."

 American Journal of Sociology, 68, 1962, 88-96.
- Sanders, Clinton. "Rope Burns: Impediments to the Achievement of Basic Comfort in the Field Research Experience." In Wm. Shaffir et al., (eds.), <u>Fieldwork Experience</u>, New York: St. Martin's Press, 1980.

- Sarason, I., and V. Ganzer. "Concerning the Medical Model." American Psychologist, 23, 1968, 507-510.
- Sarbin, T.R. "The Scientific Status of the Mental Illness Metaphor."
 In Changing Perspectives of Mental Illness, S. Plog and R. Edgerton, (eds.). New York: Rinehart and Winston, 1969.
- Scheff, Thomas. "The Societal Reaction to Deviance: Ascriptive Elements in the Psychiatric Screening of Mental Patients in a Midwestern State." Social Problems, 11, 1964, 401-413.
- Being Mentally Ill. Chicago: Aldine Publishing Co., -
- and Row, 1967.

 Mental Illness and Social Processes. New York: Harper
- Sociological Review, 40, 1974, 252-257.

 Merican
- Labelling Madness. New Jersey: Prentice-Hall, 1975. -
- Schneider, J.W., and P. Conrad. "In the Closet with Illness: Epilepsy, Stigma Potential and Information Control." <u>Social Problems</u>, 29, no. 1, 1980, 32-44.

- Schutz, Alfred. The Phenomenology of the Social World. Illinois: Northwestern University Press, 1967.
- Schwab, J., and M. Schwab. Sociocultural Roots of Mental Illness:
 An Epidemiological Survey. New York: Plenum, 1978.
- Schwartz, Charlotte. "Perspectives on Deviance-Wives' Definitions of their Husbands' Mental Illness." Psychiatry, XX, 1957, 271-291.
- Schwartz, M.S., and C.G. Schwartz. "Problems in Participant Observation."

 American Journal of Sociology, 60, 1955, 343-354.
- Scott, R.A. The Making of Blind Men. New York: Basic Books, 1970.
- Shaffir, Wm., R. Stebbins, and A. Turowetz (eds.). Fieldwork Experience.
 New York: St. Martin's Press, 1980.
- Shaskolsky, Leon. "The Development of Sociological Theory in America:
 A Sociology of Knowledge Interpretation." In Sociology of Sociology,
 L.T. Reynolds and J. Reynolds, (eds.). New York: David McKay Co.,
 1970, 7-30.
- Shibutani, Tamotsu. "Reference Groups and Social Control." In <u>Human</u>

 <u>Behavior and Social Processes: An Interactionist Approach</u>, A.

 <u>Rose (ed.).</u> Boston: Houghton Mifflin.

- Siegler, Miriam, and Humphrey Osmond. <u>Models of Madness, Models of Medicine</u>. New York: MacMillan, 1974.
- Smith, D.L. "Symbolic Interactionisms Definitions of the Situation from Becker to Lofland." Catalyst, (Winter), 1973, 62-75.
- Smith, H., and J. Thrasher. "Roles, Cliques and Sanctions: Dimensions of Patient Society." In <u>The Mental Patient: Studies in the Sociology of Deviance</u>, S. Spitzer and N. Denzin, (eds.). New York: Mc-Graw-Hill, 1968, 316-323.
- Smith, K., M. Pumphrey and J. Hall. "The 'Last Straw': The Decisive Incident Resulting in the Request for Hospitalization of One Hundred Schizophrenic Patients." American Journal of Psychiatry, 120, 1963, 228-233.
- Spitzer, S., and N. Denzin (eds.). The Mental Patient: Studies in the Sociology of Deviance. New York: McGraw-Hill, 1968.
- Spradley, J.P., and D.W. McCurdy. <u>Anthropology: The Cultural Perspective</u>. New York: Wiley, 1975.
- Strauss, J.S. (et al.). "Do Psychiatric Patients Fit their Diagnoses? Patterns of Symptomatology as Described with the Biplot." Paper presented at the 1977 meeting of the American Psychiatric Association, Canada.
- Sykes, Gresham. The Society of Captives. New Jersey: Princeton University Press, 1958.

- Townsend, J. <u>Cultural Conceptions and Mental Illness</u>. Chicago: University of Chicago Press, 1978.
- Turner, Victor. The Ritual Process. Chicago: Aldine, 1969.
- Vidach, A. "Participation Observation and the Collection and Interpretation of Data." American Journal of Sociology, 60, 1955, 354-360.
- Waitzkin, H., and B. Waterman. The Exploitation of Illness in Capitalist Society. Indianapolis: Bobbs-Merrill Co., 1974.
- Wallace, A.F.C. "Mental Illness, Biology and Culture." In <u>Psychological</u> Anthropology, F.L.K. Hsu, (ed.). San Francisco: Schenkman, 1972.
- Weinberg, M.S. "Becoming A Nudist." Psychiatry, 29, no. 1, 1966.
- West, W. Gordon. "Access to Adolescent Deviants and Deviance." In <u>Fieldwork Experience</u>, W. Shaffir, et al., (eds.). New York: St. Martin's Press, 1980:31-43.
- Whatley, C.D. "Social Attitudes Toward Discharged Mental Patients." Social Problems, 6, 1959, 313-320.
- Whybrow, P.C. "The Use and Abuse of the 'Medical Model' as a Conceptual Frame in Psychiatry." Psychiatry in Medicine, 3, 1972, 333-342.

- Whyte, Wm. F. <u>Street Corner Society</u>. Chicago: University of Chicago Press, 1955, (orig. 1943).
- Wood, E.C., J. Rakusin, and E. Morse. "Interpersonal Aspects of Psychiatric Hospitalization." <u>Archives of General Psychiatry</u>, 3, 1960, 632-641.

- Woodward, J.L. "Changing Ideas on Mental Illness and Its Treatment."

 American Sociological Review, 16, 1951, 443-454.
- Wright, B.A. Physical Disability, A Psychological Approach. New York: Harper, 1960.
- Yarrow, M., C. Schwartz, H. Murphy, and L. Deasy. "The Psychological Meaning of Mental Illness in the Family." <u>Journal of Social Issues</u>, 11, no. 4, 1955, 12-24.
- Zung, W. "A Cross-Cultural Survey of Symptoms of Depression." American Journal of Psychiatry, 126, 1969, 116-121.